



AGREEMENT BETWEEN OAKLAND UNIFIED SCHOOL DISTRICT AND OAKLAND EDUCATION ASSOCIATION

Representing Teachers and Other Certificated Classifications

For the Period July 1, 2021 through October 31, 2022

Oakland Unified School District (OUSD) 1000 Broadway, Suite 400 Oakland, CA 94607

Oakland Education Association (OEA) 272 E. 12th Street Oakland, CA 94606 (510) 763-4020 Fax (510) 763-6354

IN	WITNESS	WHEREOF, the	parties	have	executed	this	Agreement	on	the	5th	day of
	August	2022.									

OAKLAND EDUCATION ASSOCIATION

OAKLAND	UNIFIED	SCHOOL	DISTRICT
---------	---------	---------------	----------

_	Kulen	
By:	Keith Brown (Aug 5, 2022 15:59 PDT)	
	Keith Brown, Presid	ent

Jenine Lindsey, Executive Director, Labor

Bargaining Teams

Oakland Unified School District
Jenine Lindsey, Executive Director, Labor Relations
Sondra Aguilera, Chief Academic Officer
Tara Gard, Chief of Talent
Jennifer Blake, Executive Director Special
Ed/SELPA

TABLE OF CONTENTS

ARTICLE 1 - AGREEMENT		8
1.1 General Provisions1.2 General Bargaining Provisions1.3 Term of Agreement1.4 Reopeners	8 8 8 8	
ARTICLE 2 - RECOGNITION		9
2.1 Oakland Education Association	9	
ARTICLE 3 - DEFINITIONS		10
3.1 General	10	
ARTICLE 4 - NON-DISCRIMINATION		12
4.1 General	12	
ARTICLE 5 - EMPLOYER RIGHTS		13
5.1 General	13	
ARTICLE 6 - ASSOCIATION RIGHTS		14
ARTICLE 7 - EMPLOYEE RIGHTS		16
7.1 General 7.2 Faculty Council	16 16	
ARTICLE 8 - AFFIRMATIVE ACTION		20
8.1 General 8.2 Affirmative Action Committee	20 20	
ARTICLE 9 - ACADEMIC FREEDOM		22
9.1 General 9.2 Lesson Plans	22 22	
ARTICLE 10 - HOURS OF WORK		23
 10.1 Work Year 10.2 Workday 10.3 Covering Classes for Other Unit Members 10.4 Faculty Meetings 10.5 Preparation Periods for Elementary Teachers 10.6 General Provisions 10.7 Teacher Commitment Program 10.8 Extended Day Kindergarten 10.9 Extended Collaboration Time 10.10 Premium Extra Duty Pay January 1, 2022 - October 31, 2022 10.11 Shared Commitment to Foundational Learning Series 	23 24 26 27 27 29 30 30 31 31	
10.12 Shared Commitment to Our Students	32	

ARTICLE 11 - L	<u>-EAVES</u>		34
11.1 Gener		34	
	of Absence Without Pay	34	
	of Absence With Pay	36	
11.4 Family		38	
	ded Sick Leave	38	
	rophic Leave	39	
	ational Leave	40	
	Absences Chargeable to Sick Leave	41	
•	ancy Disability	42	
	ry Leave	44	
	ral Leave	45	
11.12 Jury		45	
	gency Leave	45	
11.14 Sabb		46	
	e for Educational Meetings and Conferences	46	
11.16 Perso	onal Leave	46	
ARTICLE 12 - A	ASSIGNMENT/TRANSFER/VACANCY/CONSOLIDATION PO	<u> DLICY</u>	48
12.1 Definit	ions	48	
12.2 Vacan		48	
12.3 Applic	ation Process	58	
12.4 Assigr	nment Factors	58	
12.5 Transf		58	
	yee-Initiated Transfer (Voluntary)	59	
12.7 Emplo	yer-Initiated Transfer (Involuntary)	59	
12.8 Conso	lidations	60	
12.9 Transf	er/Consolidation Due To School Closure/Replacement	62	
12.10 Class	sroom Teacher Assignment	63	
	nment For Adult Education Program	64	
12.12 Assig	Inment of "Teachers on Special Assignment" (TSAs)	66	
	Childhood Education Center Transfers	66	
12.14 Revie	ew Process for Electronic Forms of Communication	66	
ARTICLE 13 - I	PERFORMANCE EVALUATION		68
13.1 The E	valuation System	68	
	nentation of Evaluation System	69	
•	ation Sequence	71	
	leted Forms	72	
	reement	72	
•	ies During Evaluation Period	73	
	tionary Interim Report	74	
	I Evaluation Report	74	
	w of Unsatisfactory Rating	75	
	ationary and Tenured Personnel	75	
	plaints Against Employees	76	
	nstructional Assistance Program (SIAP)	77	
	GRIEVANCE POLICY		78
14.1 Gener	al	78	

14.4 14.5 14.6 14.7 14.8 14.9	Informal Resolution of the Problem Step A - Pre-Formal Stage Level I Level II Level III Mediation Binding Arbitration General Provisions	78 78 79 79 79 80 80 80 80	
<u>ARTICL</u>	E 15 - CLASS SIZE AND CASELOADS		85
15.2 15.3 15.4 15.5 15.6 15.7 15.8 15.9	. ,	85 85 86 86 89 89 89 89	
	E 16 - COMMITTEES		91
16.2	Joint Study Committees Consultation Committees (CC) Special Education Committee	91 91 91	
ARTICL	E 17 - SAFETY AND SECURITY CONDITIONS		93
17.3 17.4 17.5 17.6 17.7 17.8 17.9 17.10 17.11 17.12 17.13 17.14 17.15	Fire and Disaster Disruptive Actions by Students Assault, Attack or Threat Against a Unit Member by Students Civil Disorder Unauthorized Person(s) After School Procedures Regarding Unsafe or Harmful Conditions School Site Discipline Committee Safety Information Classrooms Maintenance Employer Safety Committee Vehicle Teacher Initiated Suspension Form Use of Recording Devices	93 93 93 94 94 94 95 96 96 96 96 96 96 97	
<u>ARTICL</u>	E 18 - ORGANIZATIONAL SECURITY		98
	General Membership Dues Agency Fee Agency Fee Exception Reporting and Payments	98 98 98 98 99	

ARTICL	E 19 - SAVINGS PROVISION		100
19.1	General	100	
ARTICL	E 20 - CONCERTED ACTIVITIES		101
20.1	General	101	
ARTICL	E 21 - SPECIAL SERVICES AND SPECIALIZED ASSIGNMENTS		102
21.1	General Provisions	102	
	Counselors	102	
	Coaches	103	
	Bilingual Education	103	
	Home and Hospital Instruction	104	
	Instrumental Music Program	104	
21.7		104	
21.8	Librarians	104	
21.9	Nurses	104	
	Psychologists	105	
21.11		105	
21.12			
	Special Day Class Teachers (SDC)	106	
	Resource Specialists (RS)	107	
21.15	Full Inclusion Education	108	
21.16	Physical Education (P.E.) Teachers	109	
21.17	Journalism Teachers	109	
21.18	Substitute Teachers	109	
21.19	Year-Round Schools	113	
21.20	Classroom Teacher Instructional Improvement Program	113	
21.21	Adult Education Hourly	113	
ARTICL	E 22 - EARLY CHILDHOOD EDUCATION CENTERS		114
22.1	General	114	
	Definitions	114	
22.3	Association Rights	114	
22.4	Safety Conditions (ECE)	115	
22.5	Vacancies	117	
22.6	Application Procedure	117	
22.7	Assignment Factors Considered for Position	118	
22.8	Transfer – General Provision	118	
22.9	Employer - Initiated Transfer	118	
22.10	Consolidations	119	
22.11	Transfer/Consolidation Due to Center Being Closed/Replaced	120	
22.12	Unit Member Assignment	120	
22.13		121	
ARTICL	E 23 - SUMMER SCHOOL EMPLOYMENT		125
23.1	General Provision	125	
	Calendar	125	
	Staff Selection	125	
	Evaluation	125	
	Salary	126	
23.3	Salary	120	

23.6	Alternate Teachers	126	
23.7	Sick Leave	126	
23.8	Site or Program-Specific Summer Programs	126	
ARTICL	E 24 - COMPENSATION		127
24.1	Compensation by Fiscal Year	127	
24.2	General Provisions for Placement on Teachers Salary Schedule	128	
24.3	Column Requirement	130	
24.4	Anniversary Increments	132	
24.5	Special Rates for Certain Unit Members	132	
	Salary Provisions for Certain Unit Members	133	
	Work Year for Unit Members Paid on the Teachers' Salary Schedule	134	
24.8	Health and Welfare Insurance	135	
24.9	Health and Insurance Revisions 2003-2004	138	
24.10		138	
24.11		138	
24.12		138	
24.13		138	
24.14	Vision Care	139	
24.15	Annuity Program	139	
24.16	Health Benefits Improvement Committee (See Side Letter Agreemen	nt in Appen	dix 14)
24.17	General Information on Employer-Paid Insurance Plans	139	
24.18	Transportation Reimbursement	140	
24.19	•	141	
24.20	Programs for Personal Growth	141	
24.21	•	142	
	E 25 - PEER ASSISTANCE AND REVIEW		143
25.1	General	143	
	Definitions	143	
25.3	Joint Committee	144	
	Participating Teachers	147	
25.5	Consulting Teachers	150	
25.6	Peer Assistance and Review Process for Permanent Teachers	153	
25.7	Participating Teacher Due Process Rights	154	
25.8	Miscellaneous	154	
	E 26 - CONTRACT WAIVERS		156
		450	
26.1		156	
26.2	Waiver for Specialized Programs developed at site-level	156	
ARTICL	E 27 - SPECIALIZED PROGRAMS		157
27.1	Specialized Program Development	157	
	Specialized Programs	157	
	Application Process	157	
	• •		450
AKTICL	E 28 - RESTRUCTURED SCHOOLS		159

ARTICLE 1 - AGREEMENT

1.1 General Provisions

- 1.2 This Agreement fully and completely incorporates the understanding of the parties hereto, constitutes the entire Agreement between the parties, and supersedes all previous agreements, understandings and prior practices related to matters included within this Agreement.
- 1.3 The Employer and Association will support this Agreement for its duration, provided that nothing herein shall prohibit the parties from changing the terms of this Agreement by mutual consent or pursuant to articles contained herein.
- 1.4 The Employer shall make no changes in matters affecting salaries, benefits, or working conditions covered by this Agreement without meeting and negotiating with the Association. Nothing contained in the Agreement shall be interpreted or applied to eliminate, reduce, or diminish any bargaining unit member's benefits.
- 1.5 The provisions of this Agreement shall not be interpreted or applied in a manner that is arbitrary, capricious, or discriminatory. Rules that are designed to implement this Agreement shall be consistent with the terms of the Agreement.

1.2 General Bargaining Provisions

1.2.1 A contract between the Employer and an individual bargaining unit member shall be expressly subject to the terms and conditions of this Agreement.

1.3 Term of Agreement

- 1.3.1 The term of this Agreement shall be as follows:
 - 1.3.1.1 From July 1, 2017 through June 30 2018.
 - 1.3.1.2 From July 1, 2018 through June 30, 2021.
 - 1.3.1.3 The Association shall present bargaining proposals for a successor Agreement on or about December 1, 2020, and the parties shall begin bargaining not later than January 15, 2021, or at times that are mutually agreed to by the parties.
 - 1.3.1.4 From July 1, 2021 through October 31, 2022.

1.4 Reopeners

- 1.4.1 Notwithstanding any other Article in this Agreement, the parties shall reopen negotiations during the term of this Agreement as follows:
 - 1.4.1.1 Article 21 (Inclusive Practices), and Article 10 for the 2019-20 School Year.
 - 1.4.1.2 Up to one (1) additional article for each party for the 2019-20 School Year.
 - 1.4.1.3 Up to two (2) additional articles for each party for the 2020-21 School Year.

ARTICLE 2 - RECOGNITION

2.1 Oakland Education Association

- 2.1.1 The Employer hereby recognizes the Oakland Education Association (hereinafter referred to as the Association) as the sole and exclusive bargaining representative for all of the following employee groups:
 - 2.1.1.1 All part-time, full-time, temporary, probationary and permanent pre-kindergarten/Early Childhood Education through twelfth (12th) grade teachers.
 - 2.1.1.2 Contracted Adult Education Teachers
 - 2.1.1.3 All Substitute Teachers
 - 2.1.1.4 Teachers on Special Assignment (TSA)
 - 2.1.1.5 Counselors
 - 2.1.1.6 Librarians
 - 2.1.1.7 Nurses
 - 2.1.1.8 Psychologists
 - 2.1.1.9 Interns and Pre-Interns
 - 2.1.1.10 Teachers-in-Charge
 - 2.1.1.11 Speech and Language Pathologists
 - 2.1.1.12 Adult Education Hourly Teachers

The current terms and conditions of the agreement between AFT Local 771 and OUSD, attached hereto and incorporated within shall remain the status quo for Adult Education hourly certified employees until such time as the parties reach agreement over the terms and conditions to be incorporated into the OEA/OUSD negotiated agreement.

- 2.1.1.13 All other certificated classifications not specifically excluded below
- **2.1.2** This contract shall not apply to:
 - 2.1.2.1 Confidential, Management and Supervisory employees as defined in Chapter 10.7, Sections 3540 through 3549.3 of Division 4 of Title 1 of the Government Code.
- 2.1.3 The parties recognize that subcontracting and/or diversion of bargaining unit work shall occur only in unusual circumstances. The Employer agrees such duties or work shall be initially offered to current bargaining unit members and to notify the Association through the Board's legislative agenda of any subcontracting or diversion. In circumstances that may impact the bargaining unit, the Employer will negotiate the decision and implementation upon request of the Association. It shall be the responsibility of the Faculty Council and the site administrator to resolve issues regarding work-space and equipment of unit members.

ARTICLE 3 - DEFINITIONS

3.1 General

The following definitions shall be used for reference throughout this contract.

- 3.1.1 **Superintendent** is defined as the Superintendent or State-appointed Administrator of the District or their designee.
- 3.1.2 **Employer** is defined as the Oakland Unified School District (OUSD) sometimes referred to as Board or District.
- 3.1.3 **Association** shall mean the Oakland Education Association (OEA), an affiliate of both the California Teachers Association (CTA) and the National Education Association (NEA).
- 3.1.4 **Principal/Site Leader** is defined as the management employee who has the responsibility for one or more schools or offices.

Regional Executive Officer (REXO)/Network Executive Officer/Central Administrator is defined as the management employee who has responsibility for one or more site administrators, primarily at school sites.

Assistant/Associate Superintendent/Central Administrator is defined as the management employee who has responsibility for one or more site administrators, primarily at non-school sites.

The foregoing definitions are solely for the purpose of clarifying references in this Agreement to administrative positions and do not preclude or impede the District's exercise of discretion to e.g. alter, reorganize, re-title, etc. administrative positions. In such event, the District will notify OEA of the "successor" administrative positions which will thereafter be deemed to apply accordingly to this Agreement.

- 3.1.5 **Unit Member** is defined as any certificated employee who is included in this bargaining unit and therefore covered by the terms and provisions of the Agreement.
- 3.1.6 **Seniority** shall accrue from the date of the first day of paid probationary service.
- 3.1.7 **Day** is defined as an Employer-scheduled working day, except as otherwise specified herein.
- 3.1.8 **Daily Rate** is defined as a unit member's annual salary plus stipends, divided by the number of contract days of service in a work year.
- 3.1.9 **Hourly Rates** are defined as those rates that are proportional to the daily rate divided by the number of hours in the workday.
- 3.1.10 **Emergency** is defined as an unforeseeable or unanticipated circumstance requiring immediate action.
- 3.1.11 **School Site** is defined as the Employer location where a unit member is assigned to perform duty.
- 3.1.12 **School closure or closed school** is one where no school opens at the same site or where the academic program and CDE code for the academic program operated at the site is changed.

- 3.1.13 **HR Director** is the designee of the Superintendent responsible for individual employee matters. Also may be termed Executive Officer or Assistant/Associate Superintendent of Human Resources.
- 3.1.14 Other definitions are included in the specific article to which they apply except where those terms and provisions indicate otherwise.
- 3.1.15 Wherever the terms "send," "deliver," "forward," "provide," "notice," "notify," "will make available" or words to similar effect are used in this Agreement regarding the exchange of or request for information, they are deemed to include electronic transmission, except for the following items, which shall also include "hard copy":
 - a. Notices of assignment, reassignment, transfer, or consolidation
 - b. Contractual disciplinary notices
 - c. Forms required in the evaluation process
 - d. Responses to grievances
 - e. PAR assistance plan

The parties agree to meet and confer from time to time over modifications to the foregoing list.

ARTICLE 4 - NON-DISCRIMINATION

4.1 General

- 4.1.1 The Employer shall not discriminate against any unit member on the basis of race, color, creed, age, sex, ethnic background, national origin, political affiliation, domicile, sexual orientation, marital status, handicap, membership or participation in the activities of the Association.
- 4.1.2 The Association will admit unit members to membership without discrimination on the basis of race, color, creed, age, sex, ethnic background, national origin, political affiliation, sexual orientation, marital status, domicile, or handicap.
- 4.1.3 Unit members shall be entitled to full rights of citizenship.
- 4.1.4 In application or oral interview procedures, the Employer shall not inquire about a candidate's membership in or preference for employee organizations.
- 4.1.5 In its employment practices, the Employer will implement and maintain policies and procedures in accordance with the Americans with Disabilities Act (ADA).

ARTICLE 5 - EMPLOYER RIGHTS

5.1 General

- 5.1.1 The Employer, on its own behalf and on behalf of the electors of the Employer, hereby retains and reserves unto itself without limitation, all powers, rights, authority, duties and responsibilities conferred upon and vested in it by the law, Constitution of the State of California and the Constitution of the United States.
- 5.1.2 The exercise of the powers, rights, authority, duties and responsibilities by the Employer, the adoption of policies, rules and regulations and practice in furtherance thereof, and the use of judgment and discretion in connection therewith shall be limited by the terms of this Agreement.
- 5.1.3 The Employer shall review, analyze and evaluate on a continuing basis the needs of the school district so that all committees, programs and projects will contribute to the best possible educational program.
- 5.1.4 The Employer and the Association shall make an effort to assure that the administration of all rules and regulations are reasonable.

ARTICLE 6 - ASSOCIATION RIGHTS

- 6.1.1 The Association shall have use of District facilities when not otherwise required for District business during regular business hours.
- 6.1.2 Association Communication with Members
 - 6.1.2.1 The Association shall have the right to use of District delivery service and mailboxes for distribution of organizational material. The Association shall count, package, and label materials for convenient handling. Labels shall clearly indicate the recipient by name, and the name of the organization responsible for the publication and distribution. Rights herein are subject to postal regulations of U.S. Government. If the labels are not clear, the material shall be returned to the sending organization.
 - 6.1.2.2 The Association shall have the right to use the District electronic mail service and unit member electronic mailboxes for communications to unit members without interference, censorship, or examination of such communications by the District, unless required by law. Such use shall be for the sole purpose of seeking to obtain unit member personal email addresses if such are not provided pursuant to section 6.1.6 below, or become outdated.
- 6.1.3 The Association shall have the use of the designated bulletin board for posting organizational literature at each school/center site and administrative building in areas frequented by employees, such locations to be selected by mutual agreement of the administration and the Association.
- 6.1.4 The Association shall be provided with a packet of materials for the regular Board meeting, which shall include agenda, contracts and agreements, new program proposals and supportive material. This packet shall be sent whenever such packets are mailed or delivered by the Employer.
- 6.1.5 The Employer shall provide the Association with a minimum of ten (10) days written notice of all new employee orientations events held at the beginning of each school year.
 - 6.1.5.1 Up to thirty (30) minutes of time shall be made available for the Association to present during these beginning of the year orientations.
 - 6.1.5.2 In addition, a location with easy access to orientation participants shall be provided for the Association to provide materials including but not limited to membership forms.
 - 6.1.5.3 For orientation materials that are provided electronically, the Association shall be provided access to the presentation and an opportunity to insert its own materials in any virtual meetings (Google hangouts, webinars, etc.), or electronic presentations (i.e. PowerPoint or other presentation software).
 - 6.1.5.4 For new employees hired after the beginning of the year orientation, the Association shall have access to the District online calendar of orientations held throughout the year. A reasonable amount of time shall be made available for the Association to present during these

- orientations. The provisions of sections 6.1.5.2 and 6.1.5.3 shall also apply to these orientations.
- 6.1.5.5 If the new employee orientation is held during student instructional days, the Employer shall provide release time, for a minimum of two (2) Association representatives to attend the orientation. The Association shall reimburse the District the substitute cost for such release time.
- 6.1.6 The Employer shall provide to the Association on a monthly basis, a sortable list of available information including the names, date of hire, assignments (including site), employee ID#, and current contact information: home addresses, telephone number and personal email address of all bargaining unit members if such address is available.
- 6.1.7 The Employer will provide to the Association on or before 45 days into the new school year, copies on a monthly basis of the forms compiled by the Employer that contain names and credentials for substitute teachers. This information will include home addresses and telephone numbers unless authorized by the individual employee to be withheld.
- 6.1.8 In making additional duty assignments, the Site Administrator shall take into consideration the fact that a unit member has been elected or appointed as an Association Representative and will make every attempt to reduce extra duty responsibilities.
- 6.1.9 Board rules and regulations and administrative bulletins shall be made available to unit members at the work site upon request.
- 6.1.10 Authorized representatives of the Association shall have the right of access to unit members and areas in which unit members work for the purpose of transacting official business, provided that this shall not interfere with or interrupt normal school operations or assigned employee duties.
- 6.1.11 Authorized representatives of the Association shall notify the administration office at the site immediately upon arrival.
- 6.1.12 After school hours on the first unit member work day Monday of each month shall be reserved for Association organization.
- 6.1.13 The Employer shall make every reasonable effort to promptly provide all requested public information to the Association to fulfill its role as the exclusive bargaining representative.
- 6.1.14 The Association shall have the right to distribute Association materials to all new unit members. The Employer will notify the Association of the date and time of the General Orientation Meeting and upon the Employer's invitation will allow the Association to distribute the Association materials during the meeting.
- 6.1.15 The designated Association Site Representative shall have time for Association reports and business after the close of the school/site faculty meeting.
- 6.1.16 The Employer shall provide release time up to five days, with the cost of substitutes to be borne by the Association, for up to three teachers with assigned duties who are elected to serve as official Association Representatives to the National Education Association Convention.

ARTICLE 7 - EMPLOYEE RIGHTS

7.1 General

- 7.1.1 A unit member shall be entitled to have present an Association representative when being formally reprimanded (a formal reprimand may be reduced to writing). The Employer shall employ a progressive discipline approach where appropriate. When request for such representation is made, the formal reprimand shall not be discussed until the unit member has the opportunity to have an Association representative present.
- 7.1.2 A unit member shall also be entitled to have present at meetings an Association Representative when the unit member has specific reason based on prior communications to believe that the meeting is intended for verbal reprimand.
 - 7.1.2.1 When in the opinion of the site administrator disciplinary action may occur, the site administrator will inform the unit member of their right to have a representative present. Where the member during the course of the conference has a reasonable belief that discipline may arise from the discussion, he/she may request to have an Association Representative present and the request will be accommodated if the administrator agrees that discipline may arise from the discussions.
- 7.1.3 A unit member shall not be subjected to discipline, reprimand, reduction in status or demotion without reasonable and just cause.

7.2

- 7.2.1 Any such discipline, reprimand, reduction in status or compensation shall be subject to the professional grievance procedures hereinafter set forth. All information relating to the above actions will be made available to the unit member upon request. Such information will be made available to the Association representative if the unit member requests the release of such information in writing.
- 7.2.2 Each unit member is guaranteed their right to professional representation through the Association.
- 7.2.3 Extra duty compensation assignments such as Yearbook supervision are considered term (year-to-year) assignments. The parties acknowledge that coaching assignments represent unique considerations and will be specifically discussed in Article 21 Special Services and Specialized Assignments.

7.3 Faculty Council

7.3.1 Faculty Council Purpose

A Faculty Council (FC or Council) shall be established in each school of the District to facilitate communication between the faculty and the school site administration. The FC shall be the site faculty decision-making team. Purposes of the FC are:

- 7.3.1.1 To provide a means for the faculty to make suggestions and recommendations for improvements related to the operation of the school, to both the Principal/Site Leader and the school site leadership groups and to discuss any items of common concern.
- 7.3.1.2 To provide an opportunity for the school site leadership groups and Principal/Site Leader to present concerns in order to receive reactions and recommendations from the faculty.

7.3.1.3 To discuss site-based reform proposals proposed by school site leadership groups, administration or bargaining unit member(s) regarding the delivery of instruction. To bring a proposal which would significantly impact the content and delivery of instruction before the entire faculty for thorough discussion and approval.

7.3.2 Faculty Council Function

It shall be the function of the FC to meet at least monthly in order to identify and discuss any and all problems and issues of common concern related to the program at the school. The FC will also make recommendations on site-based reforms regarding the delivery of instruction. The specific function of the Council shall be to advise and recommend. Additional functions and responsibilities included but are not limited to:

- 7.3.2.1 Elect a FC chairperson.
- 7.3.2.2 Hold regularly scheduled meetings.
- 7.3.2.3 Recommend placement of the staff development days or the equivalent number of hours and the professional development activities for the days when dates are not determined by the Association and/or the Employer (Article 10 Hours of Work).
- 7.3.2.4 Conduct secret ballot elections when appropriate.
- 7.3.2.5 Appoint or elect a FC representative to the Site-Based Decision Making Team (SBDMT) when appropriate.
- 7.3.2.6 Forward appropriate waiver requests to the Superintendent.
- 7.3.2.7 Forward appropriate waiver requests to the Association.
- 7.3.2.8 Accept the FC agenda items from unit members, which concern instructional issues.
- 7.3.2.9 Work closely with the Principals/Site Leaders to explore options for reducing or eliminating the need for roving teachers.
- 7.3.2.10 Recommend procedures to ensure staff presence for school site continuity during lunch periods.
- 7.3.2.11 Recommend suggested means to reduce paperwork required of unit members.
- 7.3.2.12 Review any changes in teacher assignments in the final master program.
- 7.3.2.13 Participate in room assignment decisions.
- 7.3.2.14 Make recommendations, along with the site administrator, to the Employer who shall work to ensure that unit members will not be required to work under unsafe or hazardous conditions or to perform tasks, which endanger their health, safety, or well being.
- 7.3.2.15 Make recommendations about campus security to the Employer in order to provide a secure campus learning environment at each school site.
- 7.3.2.16 Participate when practicable with the Superintendent in decisions to close a school when reasonable assurance of safety to students and unit members do not exist. This may occur in conditions of civil disorder or student unrest.

- 7.3.2.17 Meet with the site administrator for the purpose of pre-planning and review of the school's after-school athletic or other events for which there may be a question of student or staff safety. Should the site administrator and FC not be in agreement on a safety-related topic, the FC may appeal the administrator's decision to the District Events Safety Council.
- 7.3.2.18 Develop and implement necessary security provisions for night functions that are approved school activities.
- 7.3.2.19 Meet with the site administrator in Early Childhood Education Centers for the purpose of pre-planning and review of the center's after-hours events for questions of student or staff safety.
- 7.3.2.20 Work with the site administrator to resolve issues regarding workspace and equipment of unit members.

7.3.3 Faculty Council Election

- 7.3.3.1 During the month of September of each school year, an Association Representative shall arrange for a secret ballot election of members to serve on the FC.
- 7.3.3.2 The Council shall consist of a minimum of three (3) unit members and maximum of fifteen (15) unit members with not more than one member from a grade level in elementary schools and not more than one member from a department in secondary schools. Any OEA bargaining unit member employed full-time at the school site shall be eligible for election to the Council.
- 7.3.3.3 Any part-time OEA bargaining unit member assigned more than one-half time to a particular school shall also be eligible for election to the FC.
- 7.3.3.4 All Council members shall be nominated and elected by secret ballot at large.
- 7.3.3.5 Each nominee, regardless of the number of unit members from each grade level (elementary) and each department (secondary), will have their name listed on the ballot under the grade level or department or school site determination. The ballot will instruct the faculty to vote for one from each grade level or department.
- 7.3.3.6 Members shall be elected at-large at adult schools and Early Childhood Education Centers.
- 7.3.4 The members of the Faculty Council shall elect a chairperson and other officials. Among other duties, the chairperson shall set the agenda and present it to the Principal/Site Leader prior to the meeting. The Council shall hold regularly scheduled monthly meetings. Additional meetings may be scheduled as mutually agreed to by the Council chairperson and the Principal/Site Leader. During the regularly scheduled monthly meetings, the Principal/Site Leader or the Council chairperson may invite other school staff members to participate in the Council discussions. The Principal/Site Leader may place matters on the agenda.
- 7.3.5 Minutes of the concerns discussed and decisions made by the FC shall be taken at each meeting. When the FC meeting with the Principal/Site Leader is concluded, the unofficial minutes shall be reviewed for accuracy by the Principal/Site Leader

and the faculty council chairperson. A copy of the official minutes shall be distributed to:

- 7.3.5.1 Each Faculty Member
- 7.3.5.2 Principal/Site Leader
- 7.3.5.3 Area or Associate Superintendent
- 7.3.5.4 Director of Labor Relations
- 7.3.5.5 Association.
 - 7.3.5.5.1 Responsibilities for reproducing and distributing the minutes rest with the Principal/Site Leader and the FC chairperson.
- 7.3.6 Disputes involving the interpretation and/or application of this provision shall be resolved in the following manner:
 - 7.3.6.1 The Chairperson of the FC shall advise the site manager, Labor Relations and the Association in writing, of the nature of the dispute and of the resolution sought by the members of the council.
 - 7.3.6.2 If the site manager is unable to resolve the dispute within ten (10) working days, the matter shall be referred to the Association/Employer Resolution Committee (RC), consisting of:
 - Six (6) members
 - Three (3) members appointed by each of the respective parties.
 - 7.3.6.3 The members of the RC shall be:
 - The FC Chair or an Association Site Representative;
 - The Site Manager or designee;
 - An Employer Representative;
 - The Association President or designee;
 - Member of the Superintendent's Executive Staff;
 - Another Association Member or Staff assigned by the Association.
 - 7.3.6.4 The RC shall engage in an open and frank discussion concerning the dispute. The RC shall submit a recommendation to resolve the dispute. This resolution shall be determined by consensus decision-making and sent to the faculty for implementation within ten (10) working days.
 - 7.3.6.5 If the dispute is a violation of this contract, the matter shall be handled in accordance with Article 14 Grievance Policy.

ARTICLE 8 - AFFIRMATIVE ACTION

8.1 General

- 8.1.1 The Employer shall seek racial, ethnic, gender identity or expression, and individuals with disabilities balance in each Oakland public school consistent with the racial, ethnic, gender identity or expression, and individuals with disabilities composition of the community, consistent with state and federal law.
- 8.1.2 All statistical data gathered or printed by the Employer relative to affirmative action shall be made immediately available to the Association.
- 8.1.3 The Employer shall provide to the Association an annual report for each job classification, the number of applicants designated by ethnicity and gender.

The Employer shall provide to the Association a site-by-site break down of the number of bargaining unit members by ethnic code and gender.

8.1.4 The Association and Employer shall implement a joint "Equal Opportunity Committee" to meet quarterly to review District certificated employment demographic data and make recommendations to improve hiring practices.

8.2 Affirmative Action Committee

- 8.2.1 The Employer's Affirmative Action Committee, composed of bargaining unit members, administrators and the community, shall continue to advise, recommend and monitor the Employer's Affirmative Action Program. The Association shall appoint the bargaining unit members on the Committee. The Committee will monitor the Employer's Affirmative Action Program as defined in section 44101 of the Education Code in its current version as set forth below and as it may be modified in the future:
 - (a) (1) "Affirmative action employment program" means planned activities designed to seek, hire, and promote persons who are underrepresented in the work force compared to their numbers in the population, including individuals with disabilities, women, and persons of minority racial and ethnic backgrounds. It is a conscious, deliberate step taken by a hiring authority to assure equal employment opportunity for all staff, both certificated and classified. These programs require the employer to make additional efforts to recruit, employ, and promote members of groups formerly excluded at the various levels of responsibility who are qualified or may become qualified through appropriate training or experience within a reasonable length of time. These programs should be designed to remedy the exclusion, whatever its cause.
 - (2) Affirmative action requires imaginative, energetic, and sustained action by each employer to devise recruiting, training, and career advancement opportunities that will result in an equitable representation of women and minorities in relation to all employees of the employer.
 - (b) "Goals and timetables" means projected new levels of employment of women and minority racial and ethnic groups to be attained on an annual schedule, given the expected turnover in the workforce and the availability of persons who are qualified or may become qualified through appropriate training or experience within a reasonable length of time. Goals are not quotas or rigid proportions. They should relate both to the qualitative and quantitative needs of the employer.

- 8.2.2 The District agrees to pursue the following strategies insofar as they are consistent with Education Code section 44101 cited above:
 - 8.2.2.1 Development of recruitment pipelines to support affirmative action goals and timetables by enlisting classified personnel, substitutes and community members with relevant life and career experiences in District-paid credential programs.
 - 8.2.2.2 Recruitment from the community to replenish classified and substitute personnel as those employees enter the teaching pipeline.

ARTICLE 9 - ACADEMIC FREEDOM

9.1 General

- 9.1.1 Academic freedom shall be guaranteed to teachers in the study, investigation, presentation and interpretation of facts and ideas insofar as such facts and ideas reflect state and local prescribed courses of study.
- 9.1.2 Such academic freedom shall be subject to standards of professional responsibility with due regard for the maturity level of the students, laws of the State of California, and the Board rules and regulations.
- 9.1.3 Teachers shall have the responsibility for determining grades for students in the Oakland Unified School District in accordance with the standards for grading as established by District policy. Such grades shall not be changed except as permitted in Section 49066 of the Education Code.

9.2 Lesson Plans

- 9.2.1 The District will establish requirements for lesson plan contents, and provide them to the Association prior to proceeding with Sections 9.2.2 through 9.2.5 below.
- 9.2.2 The District will draft lesson plan format norms and seek input from the Association prior to finalizing them. The parties may mutually agree to utilize the Consultation Committee (see Section 16.2) to implement this subsection.
- 9.2.3 The District will send out the finalized norms for lesson plan format appropriate to elementary and secondary classes to each school site.
- 9.2.4 The site administrators will work with bargaining unit members to comply with the content requirements and utilize formatting norms. In doing so, the site administrators will take into account the parties' agreement to reduce paperwork (see Section 10.6.4).
- 9.2.5 Site-level issues and concerns about lesson plans will be addressed through the Faculty Council (see Section 7.2).

ARTICLE 10 - HOURS OF WORK

10.1 Work Year

The work year for teachers and nurses shall consist of 180 instructional days and six (6) non-student days, for a total of 186 duty days in the work year.

Non-student days are allocated as follows:

- Three (3) staff development days.
- Two and one half (2.5) planning days: One and one half (1.5) at the beginning and one (1) at the end of the work year.
- One half (.5) day for staff collaboration at the beginning of the work year. See Article 10.9 for process.
- 10.1.1 The minimum number of annual instructional minutes for each grade level shall be as follows:

Grade Level

•	Kindergarten	36,000
•	Grades 1-3	50,400
•	4-8	54,000
•	9-12	64.800

- 10.1.2 Subject to the approval of and the guidelines established by the Superintendent, or their designee, school site administrators shall work with school site faculty to develop a schedule for additional minimum days, which will assure meeting legally mandated minimum instructional minute requirements.
 - 10.1.2.1 Use of any leave on any staff development day shall require documentation pursuant to Article 11-Leaves.
- 10.1.3 In Adult Education, the number of instructional days will be determined by the needs of the program and the number of hours contracted between the period of July 1 through June 30 of a given year. The work year of Adult Education unit members shall be the equivalent to a maximum of 183 instructional days and three (3) staff development days, with the understanding that a full-time contract is a maximum of 1,098 instruction hours and 18 staff development hours for a total of 1116 duty hours. Adult Education unit members shall be given written notice of class, subject, site assignment and a schedule of the teaching days, hours to be worked each day, non-teaching days and planning and prep days by June 8th for the subsequent Adult Education Program year.
 - 10.1.3.1 The Site Administrator shall schedule both the teaching and non-teaching days of the certificated staff at their site prior to the start of the program year.
 - 10.1.3.2 At the request of the Employer, full-time contract Adult Education Teachers and TSAs have the option to work beyond their contract at the Adult Extended Contract Rate. Such requests by the Employer shall be made on an equitable basis.
 - 10.1.3.3 Unit members shall be permitted to request specific non-teaching days. Said written request must be submitted to the Site Administrator no later than April 1 of the year prior to the program year.

- 10.1.3.4 The Site Administrator shall consider unit member requests for specific non-teaching days to the extent possible in light of the overall program needs at their site.
- 10.1.3.5 Changes in the final schedule of non-teaching days shall be based upon program needs.
- 10.1.3.6 When two or more unit members request the same non-teaching days, seniority plus program needs shall be taken into consideration by the Site Administrator in making their decision.
- 10.1.4 Counselors shall work five (5) days prior to the teacher work year and five (5) days after the teacher work year (total of 196 workdays).
- 10.1.5 In Adult Education, counselors shall be assigned according to the number of hours for which they are contracted during a given year, July 1 through June 30.
- 10.1.6 Psychologists shall have a work year ten days longer than the teacher work year.
- 10.1.7 If no calendar10.1.7)" \s "calendar" \c 8 agreement for the subsequent school years is reached by March 1, the Association and Employer shall engage in expedited arbitration, and further agree to waive their respective rights to transcripts and post hearing briefing.

10.2 Workday

- 10.2.1 The duty day for members assigned to elementary schools shall be a continuous block of six hours and 45 minutes, including a 30-minute duty-free lunch. For secondary schools, a continuous block of seven hours, including 30-minute duty-free lunch. The working hours shall not commence before 8:00 a.m. nor conclude later than 3:45 p.m.
- 10.2.2 At the secondary level there may be an "A" or "B" period. The "A" or "B" period shall be the same number of minutes as other periods on the site and occur before the beginning of the regular working hours.
 - 10.2.2.1 Teachers may volunteer to work the "A" or "B" period. The "A" or "B" period shall be determined by program needs.
- 10.2.3 When the location of an employee's assignment changes (transfer or relocation) the site administrator shall provide the employee with one work day for moving within the same site during the normal work hours. Additional time may be granted by the current site administrator if the move within the same site has not been completed in one day. Employees moving to a different site shall be provided two (2) work days, one provided by their current site administrator and one provided by the new site administrator. Additional time may be granted by the new site administrator if the move to a different site has not been completed in two (2) days.
 - 10.2.3.1 When the location of an employee's assignment changes due to consolidation or administrative transfer, upon the employee's request, the District will provide reasonable assistance, i.e., moving boxes and transporting the employee's materials to the new location.
 - 10.2.3.2 Unit members who are not provided adequate time for moving during normal work hours, and who come in on their own time to move, shall be compensated at their hourly salary schedule rate, up to a maximum of one work day.

- 10.2.4 Normal working hours for psychologists shall be eight hours, including lunch (8:30 a.m.- 4:30 p.m.).
- 10.2.5 Normal working hours for Teachers-on-Special Assignment (TSAs) shall be a continuous block of seven hours including a thirty (30) minute duty-free lunch. Teachers-on-Special Assignment, when assigned, shall participate in professional activities and perform professional duties beyond their normal working hours, for a maximum of ten (10) hours per month. Beyond the ten (10) hour maximum, compensation shall be paid at the extra duty hourly rate. (See Appendix 210.2.5)" \s "Appendix 2" \c 8)
- 10.2.6 It is the policy of the Employer to establish reasonable and appropriate regulations governing compensation of teachers for non-classroom duties assigned by the Principal/Site Leader. The Principal/Site Leader, bearing in mind the special competencies of individual staff members, shall make every effort to give assignments during and beyond the duty day in an equitable manner. Each faculty member shall assume a fair share of the duty work load, which shall include the regular duty assignments performed during regular work hours, as well as supervisory or other responsibilities related to instructional and other extra curricular activities for students which are assigned by the Principal/Site Leader.
- 10.2.7 New employees shall be given consideration, wherever possible, for fewer or less demanding assignments of extra responsibilities.
- 10.2.8 Unit members other than TSAs shall participate in professional activities and perform professional duties beyond their regular workday as assigned by the appropriate administrator to a maximum of five hours per calendar month for the work year.
 - 10.2.8.1 Beyond the five hour maximum, compensation shall be paid at the extra duty hourly rate. Such assignments shall be equitably distributed.
 - 10.2.8.2 Such unit members shall have the opportunity to volunteer for such assignments to the site administrator, who shall make the final decision. It is understood that employees shall attend two public meetings during the school year, such as open house, back-to-school night, etc., and in addition, secondary teachers shall attend promotion and graduation exercises.
 - 10.2.8.3 Faculty and professional learning community meetings are not covered in this provision (see Section 10.4).
 - 10.2.8.4 The Employer shall set aside two hours of the five hours per month provided for in this provision for use throughout the school year as appropriate for parent-student-teacher contact.
 - 10.2.8.5 The Employer, site administration and school site staff shall encourage parents to take advantage of State legislation which permits parents to take unpaid time off to participate in their children's school program, and urge parents to contact teachers and counselors to set up conferences regarding their children's progress in school.
- 10.2.9 Unit members other than TSAs may volunteer for assigned duties beyond the five-hour limit, for which they shall be compensated. State and Federal mandated programs, by definition, and where requirements for staff involvement are beyond

- this Agreement, shall be controlling and they will be compensated beyond the five-hour limit.
- 10.2.10 Teachers shall be in their classes in advance of the first bell for a period of time necessary to begin their instructional program without delay. On days of inclement weather or emergency, teachers of first-period classes must be present in their classrooms for ten minutes prior to the bell, which calls students to the first class.
- 10.2.11 Whenever possible, substitutes shall be provided for all certificated staff assigned to school sites. Unit members should request substitutes for those certificated staff categories where substitutes are provided by the Employer.
 - 10.2.11.1 Substitute assignments are to include but not limited to the reasons listed below:
 - Employee illness or injury;
 - Participation with certain District Committees;
 - Court appearance; Jury duty;
 - Death of member of immediate family;
 - Emergency illness in immediate family;
 - Religious holidays;
 - Approved educational meetings and conferences;
 - Participation in the Association negotiating team:
 - Other meeting for purposes of negotiation;
 - Reasons approved by the Superintendent.
 - Teachers shall have the right to request specific substitutes; these requests shall be honored when feasible. Unit members shall provide necessary lesson plans for substitute employees. In the case of a last minute emergency, unit members shall have available emergency lesson plans for use by substitute employees. Emergency lesson plans shall be available in the unit member's classroom/work area or designated location in the school.
- 10.2.12 A guaranteed duty-free lunch period of no less than 30 continuous minutes shall be provided for all unit members. As long as there are sufficient numbers of unit members remaining on campus at any time to handle emergency situations, teachers shall be free to leave campus during their lunch periods.
 - 10.2.12.1 After receiving recommendations from the FC, the site administrator will establish procedures to ensure school site continuity.

10.3 Covering Classes for Other Unit Members

- 10.3.1 Elementary Substitute Duty: When teachers agree or are assigned to provide substitute coverage to parts of classes when a regular substitute has not been provided by the District's Human Resources Services and Support, the pay which would have been paid to a daily substitute shall be paid proportionately to the teachers involved. Students shall be distributed in the fewest groups possible and in the most appropriate grade levels possible.
 - 10.3.1.1 Whenever possible, volunteers will be solicited from among teachers. If there are an insufficient number of volunteers, the site administrator will assign a teacher. The assignment will be made on an equitable and rotational basis.

- 10.3.2 Secondary Substitute Duty. When Secondary school teachers are assigned for substitute teaching when a regular substitute has not been assigned, the pay, which would have been paid to a daily substitute, shall be paid proportionately to the teacher.
 - 10.3.2.1 Whenever possible, volunteers will be solicited from among teachers. If there are an insufficient number of volunteers, the site administrator will assign a teacher. The assignment will be made on an equitable and rotational basis.

10.4 Faculty Meetings

Teachers shall have the responsibility to attend faculty and professional learning communities or department meetings, which are called by the Principal/Site Leader or professional learning communities or department chairperson. The Principal/Site Leader may call a maximum of ten regular faculty meetings per year with one meeting of the ten held as part of the preschool orientation and planning program held at each school site before the opening of a new school year each fall. Teachers who attend faculty, professional learning communities or department meetings beyond their regular work day for more than three accumulated hours within a one month calendar period shall be compensated for the additional time at the extra duty hourly rate. Said compensation shall not apply to emergency faculty meetings.

- 10.4.1 Additional emergency faculty meetings may be called by the Principal/Site Leader as a result of unforeseeable or unanticipated circumstances, which require immediate action.
- 10.4.2 Teachers may place items concerning instructional issues on the agenda for FC meetings and may request that items be placed on the agenda of faculty meetings.
- 10.4.3 Professional learning communities and department chairpersons may call meetings, such meetings not to exceed ten (10) meetings per year. One meeting to take place as part of preschool orientation at the school site before the opening of school.
- 10.4.4 Faculty, professional learning communities/collaboration meetings normally begin within ten minutes after the dismissal of students, or soon thereafter as the faculty can be assembled, and do not exceed one hour and fifteen minutes in length.

10.5 Preparation Periods for Elementary Teachers

- 10.5.1 For the duration of this contract, the Employer shall provide each elementary teacher with two (2) 50-minute preparation periods per week during which time the elementary teacher shall not have classroom teaching responsibilities of which one (1) preparation period will be scheduled for common preparation and one (1) preparation will be staffed to provide intervention and enhancement opportunities as provided herein. The District shall continue to implement interventions and enhancement under the Educational Enhancement/Intervention Program (EEIP) and ensure it's alignment with the Multi-Year Academic Acceleration Plan (MAAP).
- 10.5.2 The first 30 minutes of each workday shall be a preparation period. Teachers will have rotating morning yard duty during the last ten (10) minutes of each such preparation period. The teacher who has morning yard duty is off duty ten (10) minutes before the end of that workday.

- 10.5.3 All elementary teachers not receiving two preparation periods weekly from the third week of school shall receive compensation for each preparation period at the teacher's per diem rate.
 - At Elementary sites, the first two (2) Wednesday minimum days of the year, and the last Wednesday minimum day of the year shall be used exclusively for elementary unit member directed preparation and activities.
- 10.5.4 The OEA/OUSD Educational Enhancement/Intervention Program (EEIP)
 - The OEA and OUSD mutually recognize the critical importance of a working collaboration between teachers, administrators, school site staff and the community to ensure effective interventions and sound educational enrichment programming for elementary students that goes above and beyond the core program provided by the District. The parties also recognize that all District resources should be maximized, including categorical resources and Measure G, to provide intervention and enrichment programs.
 - 10.5.4.2 The objectives of the EEIP are to:
 - 10.5.4.2.1 Provide educational enhancement opportunities for all elementary students in the District. These enrichment/intervention opportunities shall include but not be limited to art, physical education, music, technology and science.
 - 10.5.4.2.2 Establish and maintain an authentic School Site Council (SSC) process at all schools that ensures teacher participation and representation in addressing and identifying the academic needs of students in accordance with the OUSD Board adopted School Governance Policy and Administrative Regulations.
 - 10.5.4.2.3 Establish and maintain a comprehensive approach to implementing effective intervention and enhancement strategies particularly at elementary schools to meet the identified needs of District students and to provide a well-rounded educational experience for elementary students as a building block to future success at the secondary level in the District and beyond.
 - 10.5.4.2.4 Ensure alignment of the EEIP with the school site plan and the site budgeting process at each elementary school.
 - 10.5.4.2.5 Ensure effective alignment with elementary school day schedules
 - 10.5.4.2.6 Create flexibility in program implementation to maximize student achievement and the use of District teachers through supplemental instruction in the Arts, Sciences and Technology.
 - 10.5.4.3 Procedure
 - 10.5.4.3.1 The EEIP shall be used to help provide preparation periods as outlined above.
 - 10.5.4.3.2 The site Faculty Council shall make any recommendation regarding the EEIP program annually (by January 31 of each year) to the SSC.
 - 10.5.4.3.3 Any EEIP recommendation shall be reviewed and evaluated as a part of the regular school site planning process.

10.5.4.3.4 The OUSD shall post and recruit teachers in order to be fully staffed by August 1 of each school year.

10.6 General Provisions

Lesson preparation, study, student evaluation, conferences, etc., are to be conducted during those periods of time so scheduled. Teachers shall have the flexibility as to location at the site to perform such functions, provided that such performances do not interfere with their regularly assigned duties and that of others at the site. IEP Conferences will be scheduled during the regular working day, whenever possible. Up to 2 hours of substitute coverage will be provided for general education and special education teachers for each such meeting during the duty day. In the event that such meeting must be scheduled outside of the duty day in order to accommodate parents or guardians, unit members shall be compensated at their per diem rate.

- 10.6.1 Since double sessions and year-round schools affect the normal duty days, the Employer and Association agree to meet and consult at least one time prior to a Board decision to modify other school sites into either double sessions or year-round schools.
- 10.6.2 If it is determined by either party during consultation that additional year-round schools or double sessions may require negotiations in the areas of wages, hours or working conditions, the parties shall meet and negotiate over the issues prior to the implementation of double sessions or year-round schools.
- 10.6.3 The Employer will provide a daily conference period for all teachers and counselors assigned to secondary schools. The length of the conference period will be the length of the class period.
- 10.6.4 The Employer shall reduce the paperwork required of teachers. The FC should address suggested means to achieve said reductions.
 - 10.6.4.1 Additional hours required for district mandated testing and the paperwork associated, including any associated "bubbling in" or transfer of data to Scantron forms shall be carried out either during the school day or as part of the five hours of professional responsibilities in Article 10.2.8.
- 10.6.5 Adequate travel time, per move, will be provided to all teachers who move from one work site to another. Such time shall be considered as duty time and shall be exclusive of lunch and preparation time.
- 10.6.6 Special Education Provisions.
 - 10.6.6.1 Each Special Education teacher shall be assigned to a classroom/adequate work area with appropriate privacy and furniture for students, Paraprofesionals
 - 10.6.6.2 This classroom shall include adequate ventilation and secure storage for instructional materials and records.
 - 10.6.6.3 Whenever a Special Day Class (SDC) is placed on a District campus, which includes general education classes, preference shall be given to locating the SDC class in close proximity to age-appropriate general education classes.
 - 10.6.6.3.1 This provision shall not be construed in such a way as to result in increased facility costs or to require that classes be placed in rooms

which jeopardize the ability to implement and support the IEPs and programs for disabled students.

- 10.6.6.4 Special Education teachers shall be treated as part of the faculty and shall have the same rights and responsibilities of regular classroom teachers.
- 10.6.6.5 Special Education teachers shall have, on an equal basis with teachers at the site, access to all school site equipment and instructional material designated for general use by regular classroom teachers.

10.7 Teacher Commitment Program

The "Teacher Commitment Program" shall include special efforts to reach parents/caregivers of most vulnerable students considered to be at risk of "falling through the cracks" in the school system.

- 10.7.1 "Vulnerable" students are to be identified by:
 - their attendance;
 - academic performance and/or classroom conduct;
 - academic or citizenship grade below "C;" and/or
 - the Local Control Funding Formula ("LCFF")
- 10.7.2 The Employer shall conduct this program to enable each faculty to:
 - Identify "vulnerable" students by the end of the fourth week of each semester
 - Contact the parents/caregivers of all such students by the end of the sixth week of each semester by letter or phone.
- 10.7.3 The Employer shall enable each teacher to fulfill the following expectations:
 - 10.7.3.1 Each elementary teacher shall contact all parents/caregivers of each "vulnerable" student by letter or telephone and arrange a conference with the parent/caregiver and child.
 - 10.7.3.2 Each secondary "advisory" teacher shall contact parents/caregivers of "vulnerable" advisory student by telephone or letter to:
 - Alert the parent of problems in the "vulnerable" advisory and other classes:
 - Encourage the parents/caregivers to call the appropriate teachers to set up conferences; and,
 - Provide extension numbers and conference periods of the teacher(s) to facilitate contact.
 - 10.7.3.3 If a secondary school does not have an "advisory" in the regular day program, the school shall facilitate a system whereby parent(s)/guardians are contacted.
- 10.7.4 The schools shall aggressively publicize Back-to-School Night at the beginning of the School Year in order to inform parents/caregivers in a more timely fashion of the curriculum, student expectations, school, and teacher contact information and report card periods.

10.8 Extended Day Kindergarten

10.8.1 Under the auspices of the OEA/OUSD Joint Committee on Extended Day Kindergarten, OEA and the District representatives shall evaluate the Extended Day Kindergarten program annually no later than March 1 to determine the success of

- the program, as well as adherence to this collective bargaining agreement between OEA and OUSD. Kindergarten teachers shall be afforded the same terms and conditions as other classroom teachers.
- 10.8.2 A school site decision to establish an Extended Day Kindergarten shall be made through a collaborative process that includes teachers, the Faculty Council and the site administrator.

10.9 Extended Collaboration Time

- 10.9.1 All student minimum days at each school shall be extended 30 minutes beyond the normal duty day to support staff collaboration.
- 10.9.2 Such minimum days shall be limited to no more than one day per week and shall not apply to the first two (2) minimum days referenced in Article 10.5.3.1.
- 10.9.3 Collaboration activities shall be determined at the site level in collaboration with the Faculty Council and may consist of a variety of activities, including but not limited to, planning efforts to improve academic performance of students, supporting teachers with implementing academic course standards, articulation with Programs for Exceptional Children, department and grade level lesson planning, project planning and professional development.
- 10.9.4 Provisions shall be made to include unit members who are not normally included in minimum day plans.
- 10.9.5 It is the express intent of the parties that this time be integrated as part of a larger block of teacher-driven collaboration time occurring on these minimum days but shall not be used to extend the duty day on those days beyond thirty (30) minutes.

10.10 Premium Extra Duty Pay Effective January 1, 2022 through October 31, 2022

Effective January 1, 2022 through October 31, 2022, in addition to the compensation increases outlined in Article 24 (shown above), FTE unit members shall be paid for one hour per month at the Premium Extra Duty Rate (see Appendix 2 below). Unit members shall at their sole discretion determine the schedule and use of this time to:

- a. Extend collaboration time per Article 10.10.9;
- For additional prep used for cycles of inquiry connected to the Joint Commitment to Student Success (see below);
- c. Peer to peer wellness sessions for social emotional support; and/or
- d. Completion of professional learning from OUSD's Foundation Learning Series which may include live, synchronous or asynchronous sessions.

10.11 Shared Commitment to Foundational Learning Series

The parties agree that Foundational Learning is critical to the growth and development of staff in OUSD. Therefore, the Parties agree to the following guidelines for the implementation of Foundational Professional Learning for Summer and Fall of 2022.

a. All FTE unit members and new hires are encouraged to attend at least two
 (2) OUSD Foundational Professional Learning sessions during one of the weeks designated for Foundational Professional Learning during the

summer.

- b. Teachers and other 10 month FTE unit members and new hires shall be paid the Premium Extra Duty rate for participation (see Appendix 2 below \$45.00).
- c. As a part of the compensation package referenced in Article 24 and Article 10 Section 1 above of this Agreement, all unit members shall complete two sessions of Foundational Learning pertaining to their grade level or area of expertise either during the summer sessions or at another time during the school year by October 31, 2022.
- d. In OUSD, we have defined foundational professional learning as the following:
 - 1. Anti-Racist Learning; Standards & Equity Institute;
 - 2. Title IX (Sexual Harrassment); Suicide Prevention
 - Standards-based instruction focused on English Language Development, English Language Arts, Math, Science, and Music and Arts;
 - 4. Guided Language and Acquisition Design (GLAD);
 - 5. ALLAS:
 - Multi-Tiered Systems of Support;
 - 7. Multi-Sensory Instruction;
 - 8. Universal Design for Learning (UDL);
 - 9. Positive Behavior Interventions and Supports (PBIS);
 - 10. Restorative Justice (RJ);
 - 11. Oakland Educator Teacher Framework (OETF); and
 - 12. Leadership Development.

10.12 Shared Commitment to Our Students

OEA and OUSD are jointly committed to student success. As such the parties affirm a commitment to ensuring the following:

- Our youngest students develop the literacy skills they need to become lifelong readers, writers, critical thinkers and empowered community members;
- b. OUSD's middle school students are prepared to thrive in high school; and
- c. High school students in OUSD complete A-G requirements and are prepared for college.

ARTICLE 11 - LEAVES

11.1 General Provisions

The Employer shall provide the leaves set forth in this article and any other leaves mandated by State law to eligible bargaining unit members. The provisions of this Article shall apply only to eligible bargaining unit members.

- 11.1.1 A unit member granted a leave of absence for more than a semester, or more than six (6) months in a twelve (12) month program, shall be placed on the unassigned list
- 11.1.2 If the position from which the leave was granted has been filled by a substitute or temporary unit member or if the position is vacant, the returning unit member shall have the right to return to that position or a comparable position, provided it is done immediately upon return from leave and within one year.
- 11.1.3 Unit members who are unassigned and/or who are returning from leave shall be given the first opportunity to accept vacancies for which they are qualified.
- 11.1.4 For the purpose of this article, "immediate family" is defined as: Mother, father, grandmother, grandfather and grandchild of employee or spouse of employee, spouse, son, son-in-law, daughter, daughter-in-law, stepchild, mother-in-law, father-in-law, brother or sister of employee, guardian or domestic partner, dependents of the employee, other person permanently living in the immediate familial household of the employee.

11.2 Leave of Absence Without Pay

11.2.1 General Provisions

- 11.2.1.1 A leave of absence without pay is defined as an approved absence from service for a prescribed period of time (without salary and certain benefits). Unit members shall be allowed to pay premiums in order to continue fringe benefits while on leave.
- 11.2.1.2 A leave of absence without pay may be recommended by the Superintendent to the Board of Education for study, health, maternity or pregnancy, paternity, adoption or family responsibility, travel, unusual opportunity for professional employment, and/or valid personal reason.
- 11.2.1.3 Except for maternity/paternity/adoption, pregnancy and military leaves, leaves shall be granted only to tenured unit members.
- 11.2.1.4 Leaves for emergency reasons may be requested after deadlines; at least two weeks notice should be given the Superintendent in any case.
- 11.2.1.5 Leaves of absence without pay are granted for a maximum of one year.
- 11.2.1.6 Additional years may be granted in exceptional cases upon request and recommended by the Superintendent. A leave beyond two consecutive years may be granted by the Superintendent when a special benefit is accrued to the Employer.
- 11.2.1.7 Applications for leave shall be filed by March 1, for the following fall semester and November 1, for the following spring semester.

11.2.2 Professional Growth Leave

A unit member granted a leave for professional growth shall return to duty within 45 days of cessation of such leave. A unit member granted leave of absence under this provision shall present documentation of work equivalent to at least 12 semester units in one year.

11.2.3 Leave for Travel

A leave for travel may be granted for a school year or for one entire semester (See Article 24 - Compensation regarding credit).

11.2.4 Leave for Teaching in Another Country

Leaves of absence for teaching in another country may be granted. In granting leaves for teaching in another country, consideration will be given to length of service, teaching record, order of application and type of assignment. Generally, unit members must have had at least five years of service with the Employer.

11.2.5 Leave for Public Office

A unit member is permitted either part-time or full-time leave of absence without pay for the purpose of campaigning for public office to a maximum of one year. A unit member elected to full-time public office shall be granted leave of absence without pay for the period of time served in the office. Leave for less than full-time public office may be granted by the Employer.

11.2.6 Release Time Provisions for Association Officers

- 11.2.6.1 The Employer shall grant release time for up to two (2) Association officers. The Association President and any other designated officer shall receive compensation and benefits as though they were an employee of the District. The Association shall designate in writing the appropriate placement for each officer on the District's existing certificated teacher salary schedule (See also Section 24.7.5).
- 11.2.6.2 The Association shall reimburse the District for full costs of the Association President and any other designated officer on a quarterly basis.
- 11.2.6.3 Upon the end of such release time, the unit member(s) shall have the right to be reassigned to their former site and, if possible, former position, and will maintain seniority and shall advance on the salary schedule as if the officer had continued to teach.
- 11.2.6.4 The Association shall maintain workers' compensation coverage for each officer designated under these release time provisions.

11.2.7 Legislative Leave

- 11.2.7.1 A permanent unit member in a position requiring certification qualifications, who is elected to the Legislature, shall be granted a leave of absence from duties as an employee of the District by the governing board of the District.
- During the leave of absence, the unit member may be assigned by the Employer to perform less than full-time services requiring certification qualifications, for mutually agreed upon terms, conditions, and compensation.

11.2.7.3 The classification of the unit member shall not be affected. The unit member shall be entitled to return to the classification held at the time of election, at a salary to which they would have been entitled had the unit member not been accorded this absence, provided the return occurs within six months after the term of office expires.

11.2.8 General Leave

- 11.2.8.1 Twenty-five (25) full-time general leaves may be granted each year to permanent unit members who have rendered at least seven full-time consecutive years of satisfactory service immediately preceding general leave.
- 11.2.8.2 At least five (5) of these leaves shall be granted in the Early Childhood Education Centers. Adult Education and Early Childhood Education unit members with less than full-time contracts are eligible.
 - The leave shall be limited to one school year and shall be non-renewable.
 - The leave shall be without pay.
 - The request/application for leave must be filed with the understanding that the right to return to the original position is not guaranteed.
 - Upon written authorization, this leave shall be granted with the understanding that the right to return to the original position is not guaranteed.
 - This leave shall not be granted during an evaluation year or for the year in which a consecutive annual evaluation is scheduled.
 - The final determination or selection of applicants shall rest with the Employer.
 - This leave may not be used for seeking, attaining, or holding other public school employment or association work.

11.3 Leave of Absence With Pay

11.3.1 Eligibility

11.3.1.1 To be eligible to apply for leave of absence with pay, the unit member shall be in a paid status and scheduled for work on the day or days of absence.

11.3.2 Sick Leave

- Sick leave accrual shall be based on one (1) day per month of service. Unit members absent because of illness or accident not incurred on duty shall receive full salary during such illness for a period of 10, 11 or 12 days per year. Depending upon the regular 10, 11 or 12-month assignment and in the use of sick leave, the full pay shall become available at the beginning of each school year of service.
- 11.3.2.2 Unused sick days shall be cumulative without limit during employment by the District, except that days for sick leave shall neither accrue nor accumulate during a year when the unit member is on unpaid leave of absence. Upon late employment or early termination of employment, a reduction of sick leave shall be made on the basis of one day per month.
- 11.3.2.3 Upon re-employment within 39 months of resignation, the accumulated sick leave balance shall be reinstated.

- 11.3.2.4 Unit members with accumulated sick leave earned in other California public schools shall have said leave credited toward their sick leave accumulation when employed in the District.
- 11.3.3 Days of sick leave granted a unit member assigned less than full-time shall be in the proportion that the assignment bears to the assignment of a full-time unit member.
 - 11.3.3.1 All unit members working at least twenty (20) days in the fiscal year and not otherwise covered by sections 11.3.2 and 11.3.3 above shall accrue sick leave as follows:
 - 11.3.3.1.1 Unit members shall have available for use twenty-four (24) hours or three (3) days of paid sick leave annually on July 1st of each fiscal year or subsequent hire date.
 - 11.3.3.1.2 Paid sick leave may be used in two-hour increments for hourly unit members and paid at the employee's hourly rate.
 - 11.3.3.1.3 Paid sick leave shall be paid at the applicable substitute rate for substitute unit members.
- 11.3.4 An absence due to illness or injury, whether or not covered by sick leave, which exceeds five days shall be supported by: a written statement of a licensed physician giving the reason for absence, the first and last day of illness or injury and the date the unit member is able to return to normal duties.
- 11.3.5 Unit members absent on sick leave shall notify the supervisor or the Substitute Office at least one day in advance of expected return in order that any substitute service may be terminated.
- 11.3.6 Following an absence of thirty (30) days or more due to illness, the unit member shall submit a statement from the primary physician and/or appropriate specialist indicating the unit member is able to return to normal duties.
 - 11.3.6.1 If the Employer has good reason to believe the unit member may not be able to carry out their duties, it shall so inform the employee of the basis for this belief in writing within ten (10) days of receiving the employee's doctor's statement indicating the employee is able to return to normal duties. The Employer may require an additional verification from a primary physician and/or appropriate specialist or require clearance by an Employer- designated physician, before allowing the unit member to return to active work status. Any required examination by an Employer-designated physician shall be at the Employer's expense. It is the responsibility of the unit member to attend all scheduled appointments. Failure to do so shall result in loss of paid leave status.
 - 11.3.6.2 If the District requires clearance by a district-designated physician, it shall obtain such clearance within fifteen (15) days of notifying the employee it intends to seek the opinion of a District-designated physician. Failure to meet this timeline shall result in the employee either being returned to work on the 16th day, or be placed on paid administrative leave pending completion of the District-designated physician's medical evaluation.
 - 11.3.6.3 If the report of this unit member's physician conflicts with the report of the Employer's physician, a third physician who shall be selected from a list of qualified medical examiners obtained from the State Industrial Medical

Council. The selected physician shall perform a medical evaluation and make the deciding report.

- During the time it takes to resolve the conflicting physicians' reports, and to receive the final report from the third party qualified medical examiner referred to above, the unit member shall remain on fully paid status, and with no further sick leave deduction.
- 11.3.7 Absences paid under the occupational leave policy, including those due to certain childhood communicable diseases deemed to be work-connected and which cause doctor imposed quarantine, are not charged to the unit member's sick leave benefits. However, the underlying medical condition must be verified by a doctor's certificate and the occupational leave must be in compliance with provisions of the Workers' Compensation Program.
- 11.3.8 The Employer shall provide each unit member with a monthly statement of the amount of sick leave accrued.
- 11.3.9 The Superintendent may require a unit member to verify the claimed reason for absence by submission of a physician's statement or other verification when it is believed that no valid grounds exist for the unit member's claim for absence. For implementation refer to "Agreement" in Appendix 711.3.9)" \s "Appendix 7" \c 8.

11.4 Family Sick Leave

Unit members shall be entitled to use up to a maximum of one-half of their annual entitlement to sick leave to attend to an illness of a child, child of a domestic partner, parent, spouse or domestic partner of the unit member. Unit members shall comply with the procedures governing the use of sick leave set forth in the contract and shall indicate on the leave form that the employee is using "Family Sick Leave." This entitlement does not extend the maximum period of leave to which a unit member is entitled under the Family Medical Leave Act or the California Family Rights Act.

11.5 Extended Sick Leave

Extended sick leave is authorized absence, with partial pay, but with no loss of benefits, resulting from an illness, accident, or other disabling physical condition which prevents a unit member from performing their normal duties after regular sick leave has been exhausted.

- 11.5.1 Unit members may use their accumulated sick leave for any illness. When a unit member has used all their accumulated sick leave, and remains absent, the unit member is entitled to one hundred (100) days of Extended Sick Leave per year.
 - 11.5.1.1 The unit member shall receive their regular salary, minus the cost of a substitute to fill the position, for a period extending no longer than five continuous school months.
 - 11.5.1.2 Substitute costs shall be deducted based on the lowest daily substitute rate.
- 11.5.2 Extended Sick Leave shall be appropriately prorated for part-time unit members.
- 11.5.3 A unit member who has exhausted all Sick Leave and Extended Sick Leave for which they are eligible, and is unable to fully perform their contractual duties, shall be entitled to remain on unpaid leave for the rest of that school year. Or, the unit

- member may apply for the STRS Disability Allowance, the OUSD Disability Plan, or Catastrophic Leave where eligible and applicable.
- 11.5.4 Return to duty is dependent upon the physician's statement of recovery and clearance by an Employer-designated physician (See Section 11.3.6).

11.6 Catastrophic Leave

A bargaining unit member who is eligible for Extended Sick Leave may apply for and receive Catastrophic Leave if they previously donated sick leave credit to the Bank. Such use shall be pursuant to the following provisions:

- 11.6.1 The unit member shall have suffered a severe incapacitating illness or injury which is expected to be for an extended period of time, as certified by the attending physician, and which prevents the unit member from properly performing their District duties.
- 11.6.2 The time off work must create a financial hardship for the unit member because he or she has exhausted all personal sick leave, personal leave, extended sick leave, and any other paid time.
- 11.6.3 If the unit member's status with the Employer is such that the unit member is not eligible for Extended Sick Leave, then the Employer shall not accept a donation and the unit member shall not be eligible to become a participant in the Catastrophic Leave Program.
- 11.6.4 To join the Catastrophic Leave Program, unit members must have at least twenty (20) days of accrued sick leave remaining after donating to the Bank.
- 11.6.5 Catastrophic Leave may not be used for an illness or disability, which qualifies the unit member for Workers' Compensation benefits.
- 11.6.6 A Joint Association Employer committee comprised of two representatives and an alternate of each party must determine and certify that the unit member is eligible for Catastrophic Leave in accordance with 11.6.1 and is unable to work due to the severity of that personal illness, and only after adequate proof of illness has been provided in accordance with Education Code Section 44043.5, and pertinent rules and regulations of the Employer. The Association representatives shall be participants in the Catastrophic Leave Bank.
- 11.6.7 The Joint Committee shall have the responsibility of maintaining the records of the Catastrophic Leave Bank (Reserve), verifying the validity of requests, approving or denying the requests, and communicating its decision, in writing, to the participants and the Employer.
- 11.6.8 Catastrophic Leave credits may be used only for the remainder of the school year in which the extended sick leave is exhausted, plus the remainder of the following school year but in no event longer than twelve (12) consecutive calendar months following the start of the catastrophic leave.
- 11.6.9 No unit member may utilize any Catastrophic Sick Leave benefit unless they have previously donated sick leave credit to the Reserve. There shall be a 45 calendar day waiting period between a unit member's donation of sick leave credit(s), and their utilization of Catastrophic Leave credits. Sick leave and extended sick leave must be exhausted before said utilization.
- 11.6.10 All procedures for requesting Catastrophic Leave, and donating said leave credits shall be established by the Joint Committee provided for in Section 11.6.6 above.

The Joint committee shall make recommendations for continuation or termination of the Catastrophic Leave Bank to the Employer and the Association based on number of credits donated in a year, use of benefits by unit members and credits remaining in the Bank.

- 11.6.11 Catastrophic Leave may begin upon the exhaustion of the statutory period of earned and accumulated sick leave and extended sick leave benefits, depending on the unit member's request and the Joint Committee's decision. The Employer shall continue all fringe benefit contributions.
- 11.6.12 The annual period for donating sick leave credits shall be September 1, through October 31, of each school year.
- 11.6.13 Transfers of eligible leave credits are irrevocable, and shall be donated and utilized in not less than half-day increments.
- 11.6.14 A unit member on Catastrophic Leave shall not accrue any other Employer-paid leave.
- 11.6.15 The Employer and the Association shall instruct their appointees to the Joint Committee to make a good faith effort to maintain confidentiality regarding donations and utilization of the sick leave credits contemplated herein. However, there shall be no liability or recourse if said confidentiality is not maintained.
- 11.6.16 The receipt of a donated sick leave credit through Catastrophic Leave as defined herein, when combined with other District income, shall not provide the recipient with a greater daily District income/monthly fringe benefit contribution than they received immediately prior to the receipt of Catastrophic Leave.
- 11.6.17 The Catastrophic Leave recipient, because they remain in paid District status, shall continue to receive District fringe benefit contributions for the duration of said leave.
- 11.6.18 The approval or denial of Catastrophic Leave requests by the Joint Committee shall not be subject to the provisions of Article 14 Grievance Policy of this Agreement.
- 11.6.19 If and when this Catastrophic Leave provision is terminated by mutual agreement between the Employee and the Association, any unused sick leave credits in the Catastrophic Leave Bank shall be returned on a proportionate basis to enrollees currently employed by the District who did not utilize Catastrophic Leave benefits.

11.7 Occupational Leave

- 11.7.1 Unit members, who are absent from duty because of illness or injury resulting from an accident or condition deemed to be work related under the Workers' Compensation Program, shall be granted occupational leave. A work-connected injury or illness shall be so considered when properly certified as compensable for Workers' Compensation benefits. Absences not verified shall be charged against unit members' sick leave or other leave as appropriate.
- 11.7.2 Eligible unit members shall be granted Occupational Leave subject to the following conditions:
 - 11.7.2.1 Allowable Occupational Leave shall not exceed sixty (60) working days for the same accident.
 - 11.7.2.2 Allowable leave shall not be cumulative from year to year.
 - 11.7.2.3 Occupational Leave shall commence on the first day of absence and continue for a period certified by the physician that the unit member is

- disabled because of the work-related injury or illness, or for a maximum of sixty (60) working days, whichever occurs first.
- 11.7.2.4 Payment for wages lost on any day shall not, when added to a benefit granted the unit member under the Workers' Compensation Program, exceed the normal wage of the day.
- 11.7.2.5 Occupational Leave shall be reduced by one day for each day of absence, regardless of compensation award made under the Workers' Compensation Program.
- 11.7.2.6 When an industrial accident or illness occurs at a time when the full 60 days overlap into the next fiscal year, the unit member shall be entitled, for the same illness or injury, only the amount remaining at the end of the fiscal year in which the injury or illness occurred.
- 11.7.2.7 When entitlement to Occupational Leave has been exhausted, entitlement to other sick leave will be used. However, if the unit member is receiving Workers' Compensation, they shall be entitled to use only so much of the accumulated or available sick leave, vacation, or other available leave which, when added to the Workers' Compensation award, provides a full day's wage or salary.
- 11.7.2.8 Periods of absence for Occupational Leave shall not be considered a discontinuation in service of the unit member.
- 11.7.2.9 Unit members on Occupational Leave shall endorse to the Employer any wage loss benefit checks received under Workers' Compensation Program. The Employer, in turn, shall issue the unit member appropriate warrants for payment of wages or salary and shall deduct normal retirement and other authorized contributions.
- 11.7.2.10 Absences paid under the Occupational Leave policy including certain childhood communicable diseases deemed to be work connected, and which causes doctor imposed quarantine, are not charged to a unit member's sick leave benefits; if verified by a doctor's certificate and in compliance with provisions in the Workers' Compensation Program.

11.8 Other Absences Chargeable to Sick Leave

Unit members may use their accumulated sick leave balance in a school year for one or more of the following reasons of personal necessity:

- 11.8.1 Death of a member of a unit member's immediate family when additional leave is required beyond that provided in funeral leave.
- 11.8.2 An accident involving a unit member's person or property or the person or property of a member of the immediate family.
- 11.8.3 Appearance in court or before an administrative tribunal as a litigant, party, or witness, under subpoena or any order made with jurisdiction.
- 11.8.4 Appearance in court or before an administrative tribunal as an interested party with direct involvement.
- 11.8.5 An emergency caused by an illness of a member of the unit member's immediate family that requires the unit member's absence from work.
- 11.8.6 Emergency delay in travel.

11.9 Leaves Related to Maternity, Paternity, and Adoption

- 11.9.1 Pregnancy Disability Leave Procedures to be followed when applying for a pregnancy disability are:
 - 11.9.1.1 A letter from the unit member's physician verifying pregnancy and approximate delivery date shall be filed in the Human Resources Services and Support Office.
 - The unit member shall have their physician verify the period of time they are disabled and cannot perform the functions of their assignment. During the period of pregnancy disability leave, the District shall use/deduct the employee's accumulated Sick Leave (section 11.3.2) and, if necessary, the 100 days of Extended Sick Leave (section 11.5) or vacation (non-scheduled days) where applicable. When accumulated or extended sick leave is used for the purpose of pregnancy disability, it is a separate entitlement from the twelve (12) weeks of Child Bonding Leave (section 11.9.2) or leave under the Family Medical Leave Act/California Family Rights Act (section 11.9.3).
 - 11.9.1.3 The use of sick leave for pregnancy disability shall be treated the same as any other disability for which sick leave is granted.
 - 11.9.1.4 At any time a unit member is absent as a result of their physical disability arising out of their pregnancy, the Employer at its expense may request a doctor's verification of their inability to render service to the Employer.
 - 11.9.1.5 In order to use sick leave for pregnancy disability, the unit member must have been actually rendering paid service to the Employer and not on unpaid leave immediately preceding the disability.
 - 11.9.1.6 A unit member temporarily disabled as a result of pregnancy, termination of pregnancy, or childbirth may return to duty when they are physically able to render full and complete service to the Employer.
 - 11.9.1.7 Upon returning to duty, the unit member shall provide a doctor's verification that they are physically able to render full and complete service to the Employer.
 - During this period of leave, the Employer's contributions to cover fringe benefits for health, dental, vision care, and life insurance shall continue for unit members.

11.9.2 Child Bonding Leave

- 11.9.2.1 Child Bonding Leave: Unit members may elect to utilize up to 12 weeks of child bonding leave occasioned by the birth of the employee's child, or the placement of a child with the employee in connection with the employee's adoption or foster care of the child as provided by the California Family Rights Act (CFRA). If both parents of a child who are entitled to family care leave under section 11.9 of this contract are employees of the District, each is entitled to take the full amount of leave specified in 11.9.2.1.
- 11.9.2.2 For birthing mothers, the 12 week child bonding leave shall commence (but not immediately follow) the completion of any pregnancy disability leave on a date designated by the unit member and must be completed

within the one year of the birth, adoption, or foster care placement of a child.

- 11.9.2.3 For non-birthing parents, the 12 week child bonding leave shall commence on the first day of such leave.
- 11.9.2.4 Pursuant to Education Code section 44977.5, if an employee exhausts their accumulated sick leave (section 11.3.2) prior to expiration of the 12 week child bonding leave, they shall be entitled to differential pay as defined in section 11.5.1 for the balance of the 12 week period. The school district shall make every reasonable effort to secure the services of a substitute employee.
- Pursuant to the CFRA, child bonding leave must be completed within one year of the birth, adoption, or foster care placement of a child. The balance of any unused child bonding leave shall roll over into the subsequent school year and is usable provided the leave is taken within one year of the birth, adoption, or foster care placement of a child.
- 11.9.2.6 The District must be provided with at least thirty (30) days prior notice of intent to take child bonding leave if the need for the leave is foreseeable based on an expected birth, placement for adoption or foster care. In all other cases, notice shall be given as soon as practicable.
- 11.9.2.7 Pursuant to Education Code section 44977.5, in order to qualify for child bonding leave, employees must have completed one year (twelve months of service for the District, but are not required to have at least 1,250 hours of service during the previous one year (twelve months) period.
- 11.9.2.8 Leave in this section may be taken non-consecutively. The minimum duration of the leave shall be two weeks, except that employers must grant a request for a leave of less than two weeks' duration on any two occasions and may grant requests for additional occasions of leave lasting less than two weeks.
- 11.9.3 Family Leave: Family Medical Leave Act (FMLA) and California Family Rights Act (CFRA)
 - 11.9.3.1 Employees who have completed one year (twelve months) of service for the District, and at least 1,250 hours of service during the previous one year (twelve months) period, have the right to an unpaid leave of absence for up to 12 workweeks within a rolling 12-month period for the purpose of the employee's own serious health condition, caring for a new baby, a newly adopted baby, or a newly placed foster child or for a child, spouse, or parent with a serious health condition.
 - 11.9.3.2 Family leave under this section shall be unpaid unless it is taken pursuant to section 11.9.1 (pregnancy disability), 11.9.2 (child bonding [if the employee elects to exhaust their sick leave]), above or is taken due to the employee's own serious health condition.
 - 11.9.3.3 The employee's accumulated sick leave (section 11.3.2) and the 100 days of extended sick leave (section 11.5) shall run concurrently with FMLA or CFRA leave when it is used for purposes of the employee's own serious health condition, other than pregnancy disability.

- 11.9.3.4 When this extended sick leave is used for the purpose of pregnancy disability, it is a separate entitlement from CFRA leave.
- 11.9.3.5 There is no carry-over of unused leave from one 12-month period to the next 12- month period.
- 11.9.3.6 "Parent" means a biological, foster or adoptive parent, a step-parent, a legal guardian, or other person who stood in loco parentis to the employee when the employee was a child. "Child" means a biological, adopted or foster child, step- child, a legal ward, or a child of a person standing in loco parentis who is either under 18 years of age or an adult dependent child who is incapable of self-care because of a mental or physical disability.
- 11.9.3.7 The provisions of Article 11.2.4 shall not apply to family care leave other than for purposes of child bonding.
- 11.9.3.8 The employee shall provide reasonable advance notice to the District of the need for family care leave, the date the leave will commence, and the estimated duration of the leave. If the need for a leave becomes known more than thirty (30) days prior to the date a leave is to begin, the employee must provide at least thirty (30) days written advance notice.
- 11.9.3.9 If verification is required by the District to verify the serious illness of the child, spouse or parent, upon District request, the employee will provide medical verification by the treating health professional.
- 11.9.3.10 Health insurance coverage shall be maintained during family care leave and paid for by the District for the duration of the leave not to exceed 12 workweeks in a 12-month period.
- 11.9.4 Additional Unpaid Leave. In addition to the above, a unit member may request unpaid leave prior to and following pregnancy leave or childbirth for a period of one year. A unit member whose spouse or partner is pregnant or involved in adopting a child may request an unpaid leave for a period of one year. Leaves may be extended for a one-year period upon request and approval by the Employer for a maximum total of three years. Unit members on a non-paid maternity leave shall have the same fringe benefits as stated in Pregnancy Leave for that period of time the physician certifies they are disabled.

11.10 Military Leave

The Employer shall abide by the provisions of Education Code Section 44800, Military and Veteran's Code Section 389, 394, 395, 395.01 through 395.08, 395.1 through 395.9, and all other applicable state and federal laws.

11.10.1 Active Duty:

- 11.10.1.1 Any unit member ordered to Active military duty (Active Duty) shall be granted military leave with pay for the first 30 calendar days of ordered military duty. The unit member requesting such leave shall submit official verification requiring the military duty.
- 11.10.1.2 Following the first 30 calendar days, any unit member in the service of the Employer at least one year prior to the date on which they are called to active duty, shall be entitled to an additional five (5) calendar months of

partial salary representing the difference between the employee's regular District salary and the employee's military salary. The employee is required to submit proof of the military salary they received for each month claimed.

11.10.2 Military Training Duty:

- 11.10.2.1 Any unit member in the service of the Employer for at least one year who is ordered to participate in military training duty (temporary military duty) shall be granted military leave with pay for the first 30 calendar days of such training duty. Any unit member in the service of the Employer for less than one year who is ordered to participate in military training duty shall be granted an unpaid leave of absence for the duration of the ordered-military training period. A unit member requesting leave (paid or unpaid) for military training shall submit official verification requiring the training duty.
- 11.10.2.2 A unit member working less than a twelve-month schedule is requested to arrange for military training duty to be scheduled during school vacation periods whenever possible. A claim of extenuating circumstances necessitating training duty during other periods shall be referred to the Superintendent or the Superintendent's designee for a decision.
- 11.10.2.3 Military Leave with pay as stated above shall only be provided during the work year as defined in the Agreement. An employee is not entitled to Military Leave with pay during periods outside the work year when the employee would not otherwise be compensated by the Employer.
- 11.10.2.4 Following the receipt of Military Leave with pay under this article, and during any period outside the work year, the employee is entitled to Military Leave without pay for the remainder of such ordered military service.

11.11 Funeral Leave

- 11.11.1 For each death which occurs in the immediate family of the unit member, the unit member upon request shall be granted a funeral leave not to exceed three working days with pay up to a maximum of five workdays with pay should out-of-state travel be required. Unit members will not have any deduction from accumulated sick leave for taking funeral leave.
- 11.11.2 Funeral leave not to exceed one day with pay may be granted to a unit member to attend the funeral of other close relations. Established close relatives are defined as uncle, aunt, first cousins, nephew and niece.

11.12 Jury Duty

Employees shall be granted leave for jury duty. To receive salary from the Employer, the unit member shall submit a certificate of jury service and the endorsed juror fee check (if no expenses are included) or personal check, or money order for the amount of the juror's fee, exclusive of jury duty expenses.

11.13 Emergency Leave

The Superintendent is authorized to excuse unit members from regular work for emergency reasons for periods of an hour or less without loss of salary. Unit

members may be excused for longer periods of time provided there is no cost to the Employer.

An emergency is defined as an unforeseeable or unanticipated circumstance requiring immediate action.

11.14 Sabbatical Leave

The Board of Education may grant a sabbatical leave to a permanent unit member who has rendered at least seven full-time consecutive years of satisfactory service immediately preceding the sabbatical leave. Adult Education and ECE unit members with less than full-time contracts would be eligible for sabbatical leave after seven consecutive years of satisfactory service.

- 11.14.1 The grant shall be on condition that the unit member agrees in writing to render a period of service to the Employer following return from the leave that equals twice the period of the leave.
- 11.14.2 The leave of absence may be taken as a continuous leave not to exceed one year, or in separate six-month periods, provided the leave is begun and completed within a three-year period.
- 11.14.3 A permanent unit member may be granted one sabbatical leave in each seven-year period.
- 11.14.4 Compensation shall be paid to the unit member, while on leave, in the same manner as if the unit member were teaching in the District, provided the unit member furnish a suitable bond indemnifying the governing board of the Employer against loss, in the event the unit member fails to render the agreed period of service in the employ of the District following the return of the unit member from the leave.
- 11.14.5 Should a unit member not serve for the entire period of service agreed upon, compensation paid for the leave shall be reduced by an amount which bears the same proportion to the total compensation. The amount of time not served bears the total amount of time upon which agreement was reached.
- 11.14.6 The maximum number of sabbatical leaves granted each year to unit members shall be eighteen. The unit member shall be compensated at two-thirds of their annual salary. All fringe benefits shall remain in full force and effect. At least one of these leaves shall be granted to Early Childhood Education (ECE) unit members.

11.15 Leave for Educational Meetings and Conferences

- 11.15.1 Depending upon resources budgeted for this purpose and with the prior approval of the Superintendent, unit members (including the Adult Education and Early Childhood Education) authorized to represent the Oakland District at educational meetings and conferences, may be permitted to attend such meetings with no loss in salary and with up-to-full expenses compensated by the Employer.
- 11.15.2 Unit members who are not official representatives of the District but who seek professional improvement, may upon approval, which shall not be unreasonably withheld, be allowed to attend recognized educational meetings and conferences with no loss of salary and in some instances, compensation for expenses. Unit members on an unpaid status are not eligible for these leaves.

11.16 Personal Leave

Requests for absence without loss of pay and without cost of substitute deducted for urgent personal business to a maximum of five days each year shall not be

- unreasonably denied by the Superintendent or designee. Such requests shall be submitted in advance in writing, except where circumstances make an advance request impossible. If the request is denied, the member may request a written explanation that shall be provided within 5 days.
- 11.16.1 Such leaves are limited to one day at a time and shall not be used to extend vacation leaves or holidays, except in cases of emergency.
- 11.16.2 Any unused personal leave days shall be converted to accumulated sick leave at the end of each school year.

11.17 Temporary Permits for Statutory Leave (TPSL)

- 11.17.1 Should the district employ certificated staff who hold Temporary Permits for Statutory Leave (TPSL), it shall:
- 11.17.2 Notify OEA that it has applied to the California Commission on Teacher Credentialing for a Teaching Permit on behalf of a qualified individual.
- 11.17.3 Within thirty (30) days of receiving notice, the parties shall meet to negotiate the conditions of works for TPSLs including, but not limited to, compensation, training, support, and any other areas if not covered elsewhere in this agreement.

ARTICLE 12 - ASSIGNMENT/TRANSFER/VACANCY/CONSOLIDATION POLICY

12.1 Definitions

Assignment: A position at a school or work site.

Consolidation: Reduction in the number of bargaining unit members at a given site or in a particular department or grade level of such a site.

Position Description: A description of a vacancy, which includes information such as specialized training and/or experience requirements and other site-specific needs.

Position List: A list of certificated vacancies which identifies the site, location, job title, grade level or subject area designation and credential requirements.

Reassignment: A change of a bargaining unit member's position within a school or site. Reassignment is not considered a transfer.

Seniority Date: The unit member's first date of paid probationary service.

Temporary Contract: Contract of employment for a specific period of time not to exceed one year.

Transfer: Change from one site or work location to another within the district.

Employer-Initiated Transfer: Change in a unit member's site or work location initiated by the Principal/Site Leader/site administrator and/or other administrators.

Employee-Initiated Transfer: Change in a unit member's site or work location, initiated by the unit member.

Vacancy: An authorized position that does not have a unit member, other than a substitute, assigned to it.

12.2 Vacancies

- 12.2.1 Transfer requests for staffing subsequent school years shall be filled using the following process:
- 12.2.2 Vacancies for the Subsequent School Year

Vacancies for the subsequent school year shall be filled according to a multi-phase process as set forth in this Article.

12.2.2.1 General Provisions

- **12.2.2.2 The Process**: The process for filling vacancies according to the "match process" defined in this Article is comprised of five phases which take all of the following into account:
 - a. The needs of the District to fill vacancies in a timely manner for the benefit of students:
 - Recognizing the past service of teachers who have been displaced from a position due to external factors over which they have no control: and
 - c. Empowering teachers at school sites with a significant and meaningful role in the selection process for filling vacancies.

12.2.2.3 Personnel Committee:

- a. Unit members at each school will elect a Personnel Committee ("PC") of not more than seven (7) members with the simple majority (at maximum) of members being teachers at the site. For example, a seven (7) member PC shall have four (4) teacher members, and a five (5) member PC shall have three (3) teacher members.
- b. The PC members shall, as much as feasible, reflect the needs of the grade levels, departments, and specialized programs. The PC shall call upon individuals to provide information, experience, and input in particular areas of expertise as needed. At the secondary level, at least one unit member with knowledge and/or experience in the content area, grade level, or pathway of the position for which a transfer applicant/candidate is being considered shall participate in the process.
- c. By the second Friday in January, the PC will designate up to two teachers of the PC to be trained by Human Resources.
- d. All teacher members of the PC will receive a stipend of \$250. Up to two teacher members shall receive an additional \$100 stipend for being designated as being available to participate in the summer hiring process during Phase 4.
- **12.2.2.4 Distribution of Information**: As vacancies are identified for the subsequent school year, the Site Leader/Principal shall request from the Human Resources Services and Support a list of eligible candidates who have submitted timely voluntary transfer requests or who are Talent Pool Teachers as defined in section 12.2.2.5.2(a) below returning from extended leave of absence, and qualified external applicants. This information shall be provided to the Site Leader/Principal and the PC in a manner that facilitates timely implementation of the provisions of this Article
 - a. The Human Resources Department shall update the vacancy list which shall be made available to voluntary transfer applicants and Talent Pool Teachers. The District shall promptly provide alerts as new vacancies occur.
 - b. The Human Resources Service and Support shall notify candidates, either electronically or in writing, when their names have been submitted to the site for consideration for transfer to that site or location.

12.2.2.5 The Process for Filling Vacancies

The District and the Association share an interest in facilitating placement of unit members in school programs for which there is a match between unit member desire and program need. For this reason, employee initiated transfers shall be granted upon the approval of Human Resources Services and Support (HRSS). Unit members may submit a request for transfer to HRSS as provided in this Article.

- **Phase 1 (Open Hiring)**: Commencing with the second semester, the District may fill vacancies with voluntary transfer applicants, teachers assigned to Instructional Support roles, teachers returning from extended leave, teachers exiting a school/program pursuant to Articles 12.9, 27, or 28 and external candidates hired by the District up to the date "Phase 2" opens, which shall not be later than April 1.
 - a. Criteria for selection shall be based on required credentials/qualifications and experiences and hiring criteria, defined in section c below if applicable (hereafter referred to collectively as "the match criteria").
 - The PC shall interview and observe (if possible), voluntary transfer applicants and require demonstration lessons of all other Phase 1 applicants.
 - c. Individuals selected to fill vacancies shall be determined by a "match" between the stated preference of the applicant(s)/candidate(s), and the recommendation of both the PC and the Site Leader/Principal.
 - d. Teachers participating in Phase 1 as voluntary transfer applicants must notify Human Resources in writing (including email) not later than the end of the second (2nd) workweek in January. This date shall also be the deadline for the first early declaration of separation. Unit members who wish to be considered for assignment to a vacancy and have not submitted a timely transfer request as required herein, must submit an application (either electronically, on-line, by mail, or in person) for the position listed on the Position List to the Human Resources Services and Support.
 - e. Each voluntary transfer applicant shall indicate in the written notice one of the following choices:
 - i. The employee is leaving their current assignment in order to participate in Phase 1; OR
 - ii. The employee may transfer depending on the vacancy opportunities, but is not leaving their current assignment at this time.
 - f. Voluntary transfer applicants who submit a section (e)(i) notice above shall be considered for known vacancies first, along with other teachers and candidates described in paragraph "Phase 1" above.
 - g. After all "(e)(i) teachers" have been considered, voluntary applicants who submitted a section (e)(ii) notice above shall then be considered for known vacancies, along with other teachers and candidates described in paragraph "Phase 1" above.
 - h. The "(e)(ii) teachers" referred to in section (g) immediately above will receive written or electronic notification of selection or non-selection. The non-selected candidate may arrange a conference within five (5) days from the notice with the Human

Resources Director or designee. The "(e)(ii) teachers" referred to in section (g) immediately above may choose among the following options:

- i. Accept a "match" and voluntarily transfer; or
- ii. Decline "match" opportunities for transfer and retain their current assignment; or
- iii. Leave their current assignment, and participate in the transfer process according to the remaining Phases set forth in this Article.
- **12.2.2.5.2 Phase 2 (Priority Advisory Matching)**: Beginning not later than April 1 and extending for twenty (20) teacher workdays, vacancies shall be filled exclusively by teachers in the Talent Pool through the following process.
 - a. The Talent Pool is comprised of teachers whose current positions have been consolidated due to reasons including overstaffing, school redesign, school closure, unassigned Instructional Support teachers, teachers exiting a school/program pursuant to Articles 12.9, 27, or 28, and unassigned teachers returning from an extended leave of absence.
 - b. In order to be considered for placement, a teacher in the Talent Pool must participate in the process set forth in this section. Failure to do so shall result in a teacher foregoing the priority placement rights set forth in Phase 2.
 - i. "Participate" means attend the school site showcase, present a demonstration lesson (only for teachers returning from extended leave) and engage with the PC. Thereafter, Talent Pool Teachers must submit a list of up to five (5) preferred schools to the Human Resources Department by the deadline established by HR.
 - ii. Talent Pool Teachers shall be offered two (2) days of substitute coverage to "participate" during Phase 2 after the school site showcase.
 - c. In order for the PC to exercise the right to provide the Advisory Matching Feedback Form to a Talent Pool teacher who has included a school among the list of preferences, the school must participate in the process with such teacher. Failure to do so shall result in the PC foregoing its right to provide the Form to such teacher, and the teacher shall be deemed to be a "match" for that school.
 - "Participate" means attend the school site showcase, offer visits to the site, engage with interested teachers, watch demonstration lessons (only for teachers returning from extended leave), and conduct classroom observations (if possible).
 - d. The OEA President and Superintendent/designee may determine that capacity and time constraints necessitate modifications of the

levels of "participation" provided in section 2(b) and 2(c) above. This could mean, for example, that there may be insufficient time for a teacher to visit every school and/or for the PC to observe, engage with or conduct a demonstration lesson for every interested teacher.

- e. The PC's recommendation shall be based on the match criteria to determine if there is a "match."
- f. An external candidate or a voluntary transfer applicant who elected to leave their current assignment and remain in the transfer process (section Phase 1(h)(iii) above) may be selected to fill the vacancy during Phase 2 only if there is no Talent Pool teacher who meets the match criteria, in which case, sections 2(h) through (j) below shall not apply.
- g. Placement in vacancies shall be determined by a "match" between the stated preferences of the Talent Pool teachers and the recommendation of both the PC, and the Site Leader/Principal.
- h. The PC shall provide a list of teacher(s) it is recommending to the Site Leader/Principal who shall then indicate agreement or disagreement with the recommendation(s).
- i. If the Principal/Site Leader disagrees with any PC recommendation(s) on the list, the Principal/Site Leader and the PC shall be afforded an opportunity to reach agreement. In the absence of agreement, there shall be a review by the Superintendent and OEA President, whose decision shall be final.
- j. The final list will then be forwarded to the Human Resources Department which shall facilitate the placement of Talent Pool Teachers at any of each individual's up to five (5) preferred schools.
 - i. For example, a less senior teacher could be placed at "matched" school "A" for which a more senior teacher did not "match.' Instead the more senior teacher could be placed at school "B" for which there was a "match."
 - ii. In any event, a Talent Pool Teacher who has not been "matched" with any of their up to five schools, shall be placed at one of those five schools based on seniority, or may elect one of the following options in section (k) immediately below.
- k. A Talent Pool Teacher who does not elect to be placed at a school as provided in section (j)(ii) above may instead exercise one of the following options:
 - Be placed in available Instructional Support roles. These teachers shall be placed by the District in order of seniority, with the District giving due consideration to the teacher's preference, strengths, and interests; OR
 - ii. Elect to remain in the Talent Pool.

- **12.2.2.5.3 Phase 3 (Advisory Matching)**: Following the close of Phase 2 above and extending through June 30, the District may fill vacancies from teachers remaining in the Talent Pool, remaining voluntary transfer applicants who elected to remain in the transfer process (section Phase 1 (h)(iii) above), and external candidates, subject to the match criteria and the following process.
 - a. Teachers remaining in the Talent Pool at the close of Phase 2 shall be considered for new vacancies, providing they communicate to the Human Resources Department in writing (including email) their interest in specific listed vacancies with in three (3) teacher work days after a vacancy is posted to the vacancy list. Talent Pool Teachers and schools with vacancies must "participate" as defined in Phase 2 above, provided, however, there will be no school site showcase.
 - i. For new vacancies arising after the end of the teacher work year through June 30, participation by the PC shall be by the designated "summer" PC members, and the three (3) days referred to in (a) immediately above shall be days on which the District Office is open for business.
 - b. Individuals selected to fill vacancies shall be determined by a "match" between the stated preference of the applicant(s)/candidate(s), and the recommendation of both the Personnel Committee ("PC") and the Site Leader/Principal.
 - i. For Teachers remaining in the Talent Pool who meet the time line in section (a) above, the determination of "matches" shall be made and reported to Human Resources not later than eighteen (18) teacher work days following the posting of the vacancy.
 - ii. As soon as possible thereafter, Human Resources will inform the participating Talent Pool Teachers in writing (email included) whether a "match" was determined.
 - iii. A Talent Pool Teacher who has not been matched may arrange a conference with the Human Resources Director or Designee by written request within five (5) days of being notified of not being a match.
 - iv. Following notification from Human Resources, a Talent Pool Teacher who has been "matched" with the preferred vacancy shall be placed at the school based on seniority if they notify Human Resources in writing (email included) of this election within two (2) teacher workdays, or may elect one of the following options in section (c)(i) and (c)(ii) immediately below.
 - v. Following notification from Human Resources, a Talent Pool Teacher, who has not been "matched" with the preferred vacancy shall be placed at the school based on seniority if they notify Human Resources in writing (email included) of this election within two (2) teacher workdays,

- or may elect one of the following options in section (c)(i) and (c)(ii) immediately below.
- vi. As required by Education Code section 35036, voluntary transfer applicants who were not placed in Phase 1 may elect one of the options provided in section (d)(i) and (ii) immediately below.
- c. Talent Pool Teachers meeting the time line set forth in (a) above who have not been placed may instead exercise one of the following options:
 - Be placed in available Instructional Support roles. These teachers shall be placed by the District in order of seniority, with the District giving due consideration to the teacher's preference, strengths and interests; OR
 - ii. Elect to remain in the Talent Pool.
- d. Voluntary transfer applicants who elected to remain in the transfer process may exercise one of the following options:
 - Be placed in available Instructional Support roles. These teachers shall be placed by the District with due consideration given to the teacher's preference, strengths and interests; OR
 - ii. Elect to remain in the Talent Pool.
- e. The District may fill vacancies through the Phase 3 "match" process with external candidates immediately if no qualified Talent Pool Teachers apply for the vacanc(ies).
- **12.2.2.5.4 Phase 4 (Summer Hiring)**: Following the close of Phase 3 above and extending until two (2) weeks before the first student attendance day, the District may fill vacancies from both current employees still waiting to be placed and external candidates, subject to the "match" criteria and the following.
 - a. The designated "summer" PC member(s) shall be available to participate in the hiring process over the summer.
- **12.2.2.5.5 Phase 5 (New Year)**: Following the close of Phase 4 above and extending into the upcoming teacher work year, the District may fill vacancies from both the Talent Pool and external candidates, subject to the "match" criteria and the following process.
 - a. The District shall place teachers remaining in the Talent Pool or external candidates may be placed into site vacancies. Talent Pool teachers not placed into a site vacancy will be placed into Instructional Support Roles. These teachers shall be placed by the District in order of seniority and with due consideration of the teacher's school site preference, strengths and interests.

12.2.2.6 Hiring Criteria

As used in this Article, "hiring criteria" are comprised of site-determined and District recognized programmatic elements or Site and Teacher

Agreements which require fulfillment of specific teacher responsibilities essential to the operation of the school.

- 12.2.2.6.1 These programmatic elements can apply to individual schools and programs integrated within a larger school and must be deeply integrated into the organizing principles of the school or program such that the specialized program could not function without them.
- 12.2.2.6.2 Site and Teacher Agreements must be clearly linked to the mission and vision of the specialized program and clearly identify essential instructional practices and commitments that all staff hold and which are different from traditional educational programs or courses of study.
- 12.2.2.6.3 The site based requirements referred to in sections 12.2.2.6.1 and 12.2.2.6.2 above must be identified and implemented by the school or program in order to serve as a criterion to be used in the hiring and placement process set forth in this Article.
 - a. Schools with the following identified and District-recognized organizing principles are deemed to have complied with the requirements of this section and may use hiring criteria:
 - i. Dual Language Immersion
 - ii. Linked Learning
 - iii. Newcomers Program
 - iv. Expeditionary Learning
 - v. Big Picture
 - vi. Blended Learning
 - vii. Lab Schools associated with Teacher Training Programs
 - b. In addition, schools that complete the process set forth in Article 27 may utilize their approved hiring criteria.

12.2.2.7 Affirmation Agreement:

The matters set forth in section 12.2.2.6 above shall be reduced to writing in an "Affirmation Agreement" that shall be provided to each transferred Talent Pool Teacher or external candidate at the inception of their employment at a school or in a program that meets the requirements of section 12.2.2.6 above.

- 12.2.2.7.1 All applicants/candidates are required to sign such document as a condition of working at such school or program.
- 12.2.2.7.2 Not later than the end of the first semester, the Principal/Site Leader may determine in writing that a teacher has not satisfied the terms of the Affirmation Agreement at the new site or program within a site.
- 12.2.2.7.3 In such case, the Principal/Site Leader and teacher shall discuss the written reasons supporting the determination that the Affirmation Agreement has not been satisfied. The Principal/Site Leader and teacher shall also discuss the option of the teacher's voluntary return to the Talent Pool and participation in the transfer procedure at Phase 1 for the subsequent school year.

- 12.2.2.7.4 If the teacher does not select the option to return voluntarily to the Talent Pool, the following process shall apply:
 - a. The Principal/Site Leader and the PC shall be afforded an opportunity to discuss the Principal/Site Leader's written reasons supporting the determination that the Affirmation Agreement has not been satisfied.
 - b. The teacher may also elect to meet with the PC to discuss the Principal/Site Leader's written reasons supporting the determination that the Affirmation Agreement has not been satisfied.
- 12.2.7.5 If the PC agrees with the Principal/Site Leader's determination, the teacher will automatically be placed in the Talent Pool and begin participating in the transfer procedure at Phase 1 for the subsequent school year. In the absence of agreement, there shall be a review by the Superintendent and OEA President, whose decision shall be final. This process, including the review if any, shall be completed not later than the last teacher work day in February.

12.2.2.8 Instructional Support Roles:

- 12.2.2.8.1 Teachers serve in school site roles under the supervision of the Site Leader/Principal or other supervising administrator, to provide support that may include the following:
 - a. Co-teaching
 - b. Substitute coverage for absent teachers at the school site
 - c. Push in/pull out small group instructional support
 - d. Curriculum development
 - e. Special projects assistance
 - f. Remediation
 - g. Internship support
 - h. Classroom coverage to support evaluation procedures
 - i. Differentiated Instruction Support
 - j. Prep Teacher at priority schools
- 12.2.2.8.2 Composition of the teacher's role(s) shall be identified at the time of assignment.
- 12.2.2.8.3 These assignments shall be limited to a maximum of two (2) school sites.
- 12.2.2.8.4 Priority in assigning these roles shall be given to schools with a 90% or more unduplicated student count.
- 12.2.2.8.5 Absent agreement between the teacher and the Site Leader/Principal, and approval by the District, teachers shall serve in Instructional Support Roles for only one (1) school year, upon the completion of which they shall re-enter the transfer process at Phase 1 for the subsequent school year.

- 12.2.2.9 Positions of unit members on leave for one (1) semester or less shall not be considered vacancies. In the event that the unit member extends their leave beyond one (1) semester, does not return, or chooses another assignment, the unit member's former position shall be declared vacant and shall be filled in accordance with this Article.
- 12.2.2.10 Positions of bargaining unit members serving as Consulting Teachers in the PAR program shall not be considered vacancies. The Consulting Teacher is considered on leave to a categorically funded program from their regular assignment (See Article 25 PAR Program).

12.2.3 Vacancies Occurring During The School Year

- 12.2.3.1 Unit members who are unassigned because of consolidations, employer-initiated transfers and/or who are returning from leaves shall be given the first opportunity to accept vacancies for which they are qualified.
- 12.2.3.2 All vacancies occurring during the school year shall be filled first by unassigned staff who have the required credentials, qualifications and experience. Remaining vacancies shall be filled by substitutes and new hires having qualifications for the positions.
 - 12.2.3.2.1 In no case shall the Employer hire an applicant with only an emergency credential until all applicants with preliminary or clear credentials have been considered.
 - 12.2.3.2.2 The Employer will indicate on the monthly listing of new hires provided to the Association those unit members hired with emergency credentials, intern credentials, and/or preliminary or clear credentials.
- 12.2.3.3 Vacancies occurring after the start of the school year, except in Adult Education or Early Childhood Education, may be filled for the remainder of the school year at the direction of the Superintendent or designee. The vacancy shall subsequently be posted according to the procedures in this Article and shall be filled effective with the start of the succeeding school year.
- Positions of unit members on leave for one (1) semester or less shall not be considered vacancies. In the event that the unit member extends their leave beyond one (1) semester, does not return, or chooses another assignment, the unit member's former position shall be declared vacant and shall be filled in accordance with this Article.
- 12.2.3.5 Positions of bargaining unit members serving as Consulting Teachers in the PAR program shall not be considered vacancies. The Consulting Teacher is considered on leave to a categorically funded program from their regular assignment. (See Article 25 PAR Program.)
- 12.2.3.6 Position descriptions shall include the following statements: "The Employer does not discriminate in employment on the basis of age, creed, gender, race, ethnic background, marital status, sexual orientation, national origin or disability."
- 12.2.3.7 The Employer does not discriminate on the basis of sex or handicap in educational programs and activities. (Title IX of the Education Agreement of 1972; OUSD Affirmative Action Plan, adopted March 30, 1976; Section

504, 1973 Rehabilitation Act; and the Americans with Disabilities Act, 1990.)

12.3 Application Process

12.3.1 Applications shall be available at District sites, the Administration Building on-line and provided to the Oakland Education Association. The parties encourage all unit members to file applications using the electronic process.

12.4 Assignment Factors

- 12.4.1 The Principal/Site Leader of each school site, Personnel Committee, and heads of central office divisions requesting teacher personnel shall supply the Human Resources Services and Support with the basic data upon which to determine the eligibility of a candidate for a position on the basis of the following assignment factors:
 - Possession of the appropriate California Teaching credential;
 - Qualifications and Experiences:
 - Hiring Criteria (if applicable)
- 12.4.2 Seniority in the District shall determine assignments according to the process set forth in this Article.
- 12.4.3 In recognition of the importance of experience in the District, the Employer will give full consideration to current credentialed and qualified temporary and substitute teachers who are eligible applicants for a posted vacancy according to the process set forth in this Article.
- 12.4.4 Candidates not selected for a position will be notified either electronically or in writing of the reason(s) for non-selection after a selection has been made and a candidate has accepted. The non-selected candidate may arrange a conference within five (5) days of the notice with the Human Resources Director or designee to discuss the ineligibility.

12.5 Transfers

12.5.1 General Provisions

- 12.5.1.1 A transfer is defined as an assignment change from one site to another within the District.
- Assignment changes for elementary school instrumental music instructors, EEIP teachers, special education teachers, speech and language pathologists, nurses and psychologists are not considered transfers in that such assignments are subject to change because of fluctuations in enrollment and program requirements. Prior to such changes being made, except in an emergency or for extenuating circumstances, unit members shall be consulted regarding possible changes, provided a minimum of five (5) days notice and the provisions of Article 10.2.3 shall apply.
- 12.5.1.3 Unit members shall not be transferred more than once during their probationary period except in an emergency or for extenuating circumstances. Probationary teachers may, however, volunteer for consideration for consolidation.

12.6 Employee-Initiated Transfer (Voluntary)

- 12.6.1 The Employer and the Association share an interest in facilitating placement of unit members in school programs for which there is a match between unit member desire and program need. For this reason, employee initiated transfers shall be granted upon the approval of Human Resources Services and Support (HRSS) using the process for filling vacancies in Article 12.2.2.5.
- 12.6.2 The unit member may discuss with the immediate supervisor the reasons for the transfer if he/she desires.
- 12.6.3 If a unit member's transfer request is denied, the unit member will be notified either electronically or in writing as to the reasons why. Upon written or electronic request, a non-selected candidate may arrange a conference within five (5) days of the notice with the Human Resources Director to discuss the ineligibility.
- 12.6.4 A unit member whose transfer request has been approved will be expected to remain in the new assignment for at least two (2) years, unless an emergency situation or extenuating circumstances makes it desirable for another transfer to take place.

12.7 Employer-Initiated Transfer (Involuntary)

- 12.7.1 If the Principal/Site Leader and/or other administrators initiate a transfer, the administrator shall arrange a conference with the unit member to discuss the reasons a transfer is being proposed. Prior to the conference, the unit member will be notified in writing that he/she may elect to have an Association Representative present at the conference. An Association Representative shall receive a copy of the notice. The unit member may propose alternatives to the transfer at the conference or in writing within five (5) days after the conference. At the conference, the unit member shall be provided the Position List referenced in Section 12.1. If, at the conclusion of the conference, it is determined that a transfer is desirable, the Employer may proceed with the transfer and shall provide a copy of the transfer request to the unit member and the Association listing the reasons for the transfer.
- 12.7.2 The unit member shall select in order of preference, up to his or her first five (5) choices from the Position List. The Unit member's preference shall be honored unless there is a conflict with the assignment factors previously enumerated in this Article.
- 12.7.3 Except in cases of immediate need or emergency, unit members shall be given notice of employer-initiated transfers by the first Friday in December for a transfer that is to take effect in the second semester; and by the first Friday in April for a transfer that is to take effect at the beginning of the first semester.
- 12.7.4 A unit member subject to transfer under this section may appeal the transfer as follows:
 - 12.7.4.1 Within five (5) business days from written notice of the decision, a notice of appeal must be delivered to Human Resources Services and Support.
 - 12.7.4.2 Within five (5) business days of such notice, a Joint Appeals Panel (Panel) shall be convened and will consist of two (2) members selected by the Association and two (2) members selected by the District. The

Panel shall review and consider all pertinent facts prior to rendering its decision.

- 12.7.4.3 The Panel shall render its decision within five (5) business days. If no decision is rendered within five (5) business days, or the unit member and Association dispute the decision, the Association on behalf of the unit member may appeal the decision of the Panel to expedited arbitration.
- 12.7.4.4 If the Association proceeds to arbitration, it shall notify the District in writing within five (5) business days. Within five (5) business days of such notification, representatives of the District and the Association shall select a mutually acceptable arbitrator. The selection of the arbitrator shall proceed under the mutual strike method from a list of arbitrators provided by State Mediation and Conciliation Services.
- 12.7.4.5 The arbitrator's decision shall be in writing and shall set forth the findings of fact, reasoning, and conclusions of the issues submitted. Any award of the arbitrator shall be binding on the grievant, the Association and the District. The arbitrator and arbitration proceeding shall otherwise be subject to the scope of authority and conditions enumerated in Section 14.9.5 and Section 14.9.6.
- 12.7.4.6 The Employer shall have the same duty to defend and indemnify Joint Panel members participating in the appeals process who are acting within the course and scope of their designated functions as it has to other district employees pursuant to Division 3.6, section 810 et seq., of the Government Code.
- 12.7.4.7 Functions performed by teacher Joint Panel members pursuant to the appeals process shall not constitute either management or supervisory functions as defined by subdivision and of section 3540.1 of the Government Code.
- 12.7.4.8 The Human Resources Services and Support shall immediately notify the unit member in writing of the new assignment.
- 12.7.5 Employer-initiated transfers shall not be initiated for reasons of a punitive or disciplinary nature, but rather to address specific program needs of the Employer as they relate to the specific unit member.
- 12.7.6 Unit members transferred under this section shall not be subject to another transfer for two (2) work years (the first year following the transfer and the subsequent year).

12.8 Consolidations

- 12.8.1 Prior to the initiation of this consolidation procedure, site administrators shall discuss employee-initiated transfer options with the affected staff members.
- 12.8.2 A consolidation is defined as a reduction in the number of unit members at a given site or in a particular department or grade level of such a site. Consolidations can be made due to a decrease in enrollment, curriculum change, student program change, budgetary limitation, or other circumstances producing a similar effect upon unit member assignments.
- 12.8.3 Factors to be considered in selecting a unit member to be consolidated are:
 - Credential and legal qualifications

- 12.8.3.1 All the above factors being equal, seniority in the District shall be given preference.
- 12.8.3.2 In addition, at the secondary level, major/minor fields and highly specialized skills relating to the subject area shall be considered.
- Before any consolidation actually takes place, the Principal/Site Leader shall discuss in detail with the unit member, who is being consolidated, the necessity for the consolidation of such position. Such conferences shall take place within five (5) days of written notification. The form to record the conference shall be attached to this contract. (Appendix 15 (F1)12.8.4)" \s "F1" \c 8)
- 12.8.5 The unit member to be consolidated may request an appointment to be granted within ten (10) school days, with the Director of Human Resources or designee to discuss all known vacancies and any possible future vacancies.
- 12.8.6 School site consolidations will be effected as early in the school year as possible.
 - 12.8.6.1 First Semester: Written notification to affected unit members of possible consolidation shall be given by October 1. By the date of the end of the first marking period for secondary schools, all school site classroom consolidations, including traditional elementary schools, shall be completed.
 - 12.8.6.2 Second Semester: Written notification of possible consolidation shall be given by the first Friday in January. The final consolidation plan for the second semester for all schools shall be completed no later than the second Friday in January.
 - 12.6.3 Consolidations in Early Childhood Education Centers shall be made by the last duty day in January or on the last duty day in June.
 - 12.6.4 Consolidations in year-round schools shall not occur until fifteen (15) school days after the start of the last track.
- 12.8.7 Non-classroom consolidations may take place at any time during the school year.
 - 12.8.7.1 The Employer will determine the necessity of a consolidation.
 - 12.8.7.2 When specific consolidations are to take place, the administrators of the sites involved will inform their staff and discuss, if requested, the staffing modifications required.
- 12.8.8 Unit members to be reassigned because of consolidation of their positions shall be given the opportunity to accept current vacancies for which they are qualified.
- 12.8.9 The Executive Officer of Human Resources or designee shall notify the unit member in writing of the new assignment, as soon as possible, and provide one working day of release time for moving from one site to another. Additional time, if needed, may be granted by the new site administrator.
- 12.8.10 Consolidated unit members shall have the option of returning to the school from which they were consolidated if an equivalent position for

which the unit member is qualified and credentialed becomes available. The Executive Officer of Human Resources Services and Support or designee shall notify such unit members of this option if said position is available.

- 12.8.11 Unit members shall not be consolidated more than once during their probationary period except in emergency or extenuating circumstances. Probationary teachers may, however, volunteer for consideration for consolidation.
- 12.8.12 The Employer will provide information to the Association prior to consolidations. Such information will include the sites where consolidations will take place, and any other pertinent information, which may be available.

12.9 Transfer/Consolidation Due To School Closure/Replacement

- 12.9.1 Unit members according to their seniority will have the option of being assigned to schools to which students from the closed school have been placed if positions are created due to the attendance of students from the closed school.
- 12.9.2 In the event all unit members cannot follow the students from the school due to changes in enrollment, the process set forth in this Article shall be followed.
- 12.9.3 Should the unit member not exercise this option, they will follow the process set forth in this Article.
- 12.9.4 If closure is based on inability to use the facility, when the facility is rebuilt, all unit members who were in the original school shall have first opportunity to be assigned to the new facility. If more unit members desire to return than there are positions available, the assignment factors shall be considered.
 - 12.9.4.1 After the unit members in the original school have had an opportunity to be assigned to the new facility, if vacancies still remain, then procedures for filling a vacancy under this Article shall be followed.
- 12.9.5 Unit members assigned to a school prior to grade reconfiguration (grade level changes) shall have the option of remaining at the school after reconfiguration. An exception would be allowed if the unit members do not have the credential required for the new grade level configuration.
 - 12.9.5.1 In the event that all the unit members cannot remain after the reconfiguration due to enrollment decreases, the consolidation factors will be used to determine who is to be consolidated.
 - 12.9.5.2 Unit members who wish to follow their students to another school, due to enrollment shifts required by reconfiguration, shall be granted the opportunity to do so, in accordance with the process set forth in this Article.
 - 12.9.5.3 Unit members who do not want to continue in their assignments, due to reconfiguration, shall have the first opportunity to accept current vacancies for which they are qualified according to the process set forth in this Article.
- 12.9.6 Should the unit member not exercise the option to be considered or not be selected for a position in the New School, he/she may select a position from the Position List referenced in Section 12.1, unless such selection conflicts with the assignment

factors (contained in Section 12.4). The Position List will be presented to the unit member before the end of the current school year. The unit member shall select in order of preference, up to his or her first five (5) choices from the Position List. If more than one unit member selects the same position, the unit member with the most seniority shall have preference.

12.9.7 A unit member not selected for a position in the New School under this section may appeal the decision pursuant to the appeal procedures enumerated in Section 12.7.4 above.

12.10 Classroom Teacher Assignment

- 12.10.1 Classroom teachers shall be given written notice of their tentative assignment for the following school year no later than ten (10) days prior to the last student day of the school year. Such tentative assignment shall include tentative grade, class, subject, and room assignment.
 - 12.10.1.1 The Principal/Site Leader shall have an individual conference with the teacher, regarding their assignment in the event of changes.
 - 12.10.1.2 Changes in assignment shall be made on a voluntary basis whenever possible and must be in writing.
 - 12.10.1.3 The elementary school draft grade level teacher assignments and the secondary school draft master programs including teacher assignments, shall be completed with input from the Faculty Council and all interested staff by fifteen (15) days prior to the last student day of the school year. The Faculty Council at each high school and middle school shall receive a copy of the draft Master Schedule by fifteen (15) days prior to the last student day of the school year and shall provide any recommendations regarding the draft Master Schedule not later than ten (10) days prior to the last student day of the school year.
 - 12.10.1.4 Should it be necessary to change the assignment during the summer or Intersession, the unit member shall be notified in writing at their address of record by the school administrator/designee. Written notification shall be made to an alternative location if requested by the unit member. Subsequently any changes in the grade level or final master program shall be reviewed with the Faculty Council.
 - 12.10.1.5 Room assignment decisions shall be made in consultation with the Faculty Council. Necessary changes in room assignments, which occur during the summer or intersession, shall subsequently be discussed with the Faculty Council.
 - 12.10.1.6 If a teacher is reassigned to another grade level or subject area, that teacher shall not be assigned another grade level or subject area for at least two (2) years, unless by mutual agreement.
 - 12.10.1.7 Volunteers shall be considered before involuntarily assigning members to combination classes. Unit members shall not be involuntarily assigned to a combination class in consecutive years except in an emergency or for extenuating circumstances.
- 12.10.2 The desirability of minimizing the number of different preparations is recognized. Except in exigent circumstances and appropriate to the educational program, the

preparations for elementary classroom teachers shall be limited to official enrollees in one class. For secondary teachers, the preparations shall be limited to two.

- 12.10.2.1 For secondary teachers, preparations shall be determined as follows:
 - 12.10.2.1.1 A secondary teacher who teaches a core class to a single grade level, and another separate single class, will be deemed to have 2 preparations.
 - 12.10.2.1.2 Multiple grade levels in a single class for a single subject will be counted as one preparation.
 - 12.10.2.1.3 Music and Art teachers at all levels are deemed to have 2 preparations.
 - 12.10.2.1.4 Teaching different levels of ELD, PE, or sheltered courses does not count as different preparations.
 - 12.10.2.1.5 A sheltered class taught as part of a regular course (e.g. Sheltered Algebra I and Algebra I) will count as a single preparation.
 - 12.10.2.1.6 Any preparation that results in a unit member being paid a stipend is not deemed a preparation.
 - 12.10.2.1.7 The list above is illustrative, not exhaustive.
- 12.10.2.2 If there are no exigent circumstances, or any additional preparations are not appropriate to the educational program, a unit member shall be paid for more than two preparations as follows:
 - 12.10.2.2.1 For one additional preparation: in the ensuing year the teacher will not have any more than two preparations.
 - 12.10.2.2.2 For two additional preparations: The District will pay the teacher an additional \$500 per semester.
 - 12.10.2.2.3 For three additional preparations: The District will pay the teacher an additional \$1000 per semester.
 - 12.10.2.2.4 The amounts to be paid will be prorated by semester.
- 12.10.3 To ensure students are taught by teachers working within their area of competence; a teacher shall not be assigned outside the scope of their teaching certificate and/or their major or minor fields of study.
- 12.10.4 Teachers whose schedules include assignment to more than one school shall not be required to engage in an unreasonable amount of inter-school travel. Teachers who are required to travel shall be eligible for mileage reimbursement as set forth in Section 24.17. Such teachers shall be notified of any changes in their schedules as soon as practical.
- 12.10.5 The Superintendent shall assign all newly appointed personnel to specific positions within that subject area and/or grade level for which the Board has authorized employment. New employees shall receive notice of assignments as soon as practical.

12.11 Assignment For Adult Education Program

12.11.1 Positions in the Adult Education program shall be made available to qualified certificated personnel, except in cases where vacancies are filled by administrative transfers or by unassigned staff who have the required credentials.

- 12.11.1.1 First consideration shall be given in the selection process for posted vacancies in the Adult Education Program to Adult Education unit members.
- 12.11.1.2 Vacancy openings will be posted in accordance with established District procedures unless modified by further provisions of this Agreement.
- 12.11.2 Adult Education unit members shall not be released from their assignments without prior consultation with the Principal/Site Leader, who shall state the reason for the change in assignment.
- 12.11.3 In an effort to reduce Adult Education teachers' class preparations, assignments of one or more teachers at the site may be changed. Changes made in assignment shall be voluntary.
- 12.11.4 The District's letter of employment and assignment to Adult Education unit members will be the District's commitment to employment for the specific service to be rendered. The revised letter of employment and assignment is used in the offer of employment.
- 12.11.5 A unit member's date of hire in the Adult Education Program shall determine seniority.
- 12.11.6 Contracted Adult Education unit members shall be exempt from any requirements to attend circuit or department meetings, or to participate in professional activities, or perform, or be assigned professional duties beyond the regular workday without additional compensation provided that unit members will be required to attend regular faculty meetings in the same manner and to the same extent as required of K-12 unit members.
- 12.11.7 All positions in Adult Education shall be posted. The posting period for said vacancies shall be neither less than five (5) working days nor more than fifteen (15) calendar days. This provision does not apply to vacancies that require immediate placement of unit members due to special needs of outside agencies.
 - 12.11.7.1 Adult Education unit members presently employed, who apply and are qualified, shall be given first consideration for these vacancies or new positions before a new employee is hired.
 - 12.11.7.2 All listings shall include days, hours, and duration of assignment and total hours.
 - 12.11.7.3 The Adult Education Office shall keep on file requests by unit members for additional hours in areas for which they qualify.
- 12.11.8 Presently employed Adult Education unit members with less than a full-time contract shall be given first consideration for filling positions in Adult Education.
 - 12.11.8.1 Ranking for consideration within the presently employed Adult Education unit members shall be according to subject area training, and experience as appropriate for the positions and according to seniority in the program.
 - 12.11.8.2 Presently employed Adult Education unit members with less than full-time contracts who agree to waive the upward adjustment that would result in additional yearly contracted hours will receive full consideration for all temporary positions.
- 12.11.9 By June 8, Adult Education unit members shall be given written notice of class, subject, site assignment and a schedule of the teaching days, hours to be worked

- each day, non-teaching days and planning and prep days, for the subsequent adult school year. Upon request, Adult Education unit members shall receive tentative assignments on May 15, or as soon thereafter as possible.
- 12.11.10 Assignment of work to a qualified, credentialed hourly teacher shall not be considered subcontracting or diversion of bargaining unit work provided such assignment does not violate the provisions of Section 12.11.10.1.
 - 12.11.10.1 The Employer and Association will continue to work together to enhance comprehensive Adult and Career Education program offerings and increase Adult and Career Education enrollment and utilize Adult Education and Career Education contracted FTEs. There shall be no minimum number of contracted FTE required in the Adult Education program. The Employer and Association will continue to work together to enhance comprehensive Adult and Career Education program offerings and increase Adult and Career Education enrollment and utilize Adult and Career Education contracted FTEs.

12.12 Assignment of "Teachers on Special Assignment" (TSAs)

- 12.12.1 Teachers on Special Assignment shall be given notice, in writing, of their tentative assignments when the school program is completed in June for the ensuing school year.
- 12.12.2 A TSA shall be consulted individually regarding any change in their assignment due to unanticipated circumstances after the initial assignment, and shall be informed of any changes in writing. The appropriate administrator shall hold a conference with said TSA to discuss any change of assignment.
- 12.12.3 Schedules of TSAs who are assigned to more than one school shall be arranged so that no TSA shall be required to engage in an unreasonable amount of inter-school travel. TSAs who are required to travel shall be eligible for mileage reimbursement as set forth. (See Section 24.17.) Such TSAs shall be notified of any changes in their schedule by October 1, of each school year, except in cases of emergency.
- 12.12.4 In the event eleven- or twelve-month TSAs are to be reassigned to ten-month positions for a subsequent work year, notice of possible reassignment shall be given by certified letter on or before March 15th of the current year. This reassignment shall be considered a consolidation.
- 12.12.5 The Employer shall make every reasonable effort to provide TSAs with adequate workspace, equipment, and supplies.
- 12.12.6 All TSAs shall work at least eighty percent (80%) of their work assignment time with students or in some teacher support role.

12.13 Early Childhood Education Center Transfers

12.13.1 Employer may transfer credentialed teachers to regular K-12 teaching positions. Such transfers shall be made into existing classroom vacancies within the teacher's credential. ECE credentialed unit member who do not wish to transfer shall remain in ECE and be paid on the ECE permit salary schedule.

12.14 Review Process for Electronic Forms of Communication

12.14.1 The electronic forms of communication delineated in this Article shall be subject to review by all parties involved. Both parties will meet to evaluate the effectiveness of the program.

12.14.2 The electronic program will continue to be reviewed bi-annually in May and January of each subsequent year throughout the term of this contract.

ARTICLE 13 - PERFORMANCE EVALUATION

13.1 The Evaluation System

The Employer shall provide each unit member with an electronic copy of the current Evaluation Guidelines as found in this Agreement by the 10th workday of each school year. Sample copies of evaluation plan/SMARTe goals, observation forms, evaluation forms A and B, reports and an evaluation calendar shall be included with the Evaluation Guidelines (See Appendix 613.1)" \s "Appendix 6" \c 8).

All forms shall be included in the Certificated Employee Evaluation Guidelines and in the Evaluation Handbook(s) and shall not be changed unless mutually agreed to by the parties to this Agreement.

- 13.1.1 Purposes A uniform system of evaluation and assessment of the performance of certificated personnel is required by law. The primary purpose of such evaluation is to, support, and develop, and assist the certificated Employee and the Employer to improve the quality of education offered in the District. Such evaluation may also serve as a legal basis for determining the lack of professional competence of a certificated employee.
- 13.1.2 *Criteria* Evaluation criteria shall include, but shall not necessarily be limited to, the following elements:
 - 13.1.2.1 District standards of expected student achievement at each grade level in each area of study.
 - 13.1.2.2 Assessment of certificated personnel competence as it is related to the established standards, including but not limited to the Oakland Effective Teaching Framework (OETF), for individual certificated personnel.
 - 13.1.2.3 Assessment of other duties normally required to be performed by certificated employees as an adjunct to their regular assignments.
 - 13.1.2.4 Procedures and techniques for ascertaining that the certificated employee is maintaining proper control, and is preserving a suitable learning environment. Such procedures and techniques shall include:
 - Implementation of the Core Curriculum.
 - Increased student achievement as measured by various assessment measures, such as test scores.
 - The use of curriculum embedded assessment.
 - Increased student attendance.
 - Reduced student discipline (such as suspensions and expulsions).
 - Increased parent/caregiver contacts by letter, telephone or conference.

13.1.3 Definitions

- 13.1.3.1 The Evaluator may be the unit administrator or management employee on the administrative/supervisory salary schedule designated by and working under the direct supervision of the unit administrator.
- 13.1.3.2 Evaluatee is defined as the certificated employee who is to be evaluated. The evaluatee shall be apprised of their evaluator at the beginning of the evaluation cycle, or upon change of the evaluator.
- 13.1.3.3 Unit is defined as the assigned site, department, or office.

13.1.3.4 Observation is a viewing of an employee's performance, which is reduced to writing.

13.1.4 Selection of Evaluatee

- Tenured employees shall be evaluated at least every two years except as provided in this section and as described in 13.1.4.2 below. A random method of selection shall be used to determine the evaluatees for odd and even years. Probationary employees shall be evaluated annually. The need for consecutive annual evaluations may be challenged by the evaluatee through the procedure described in Section 13.5. Consecutive annual evaluations shall be for just cause only.
- To support continuous improvement of unit members and contributions to colleagues, the District and the Association agree to the following alternative evaluation cycle: For the duration of this agreement, pursuant to Education Code section 44664, unit members with permanent status who have been employed at least 10 years with the school district, who are also highly qualified as defined in 20 U.S.C. Section 7801(23), and whose previous evaluation rated the employee as meeting or exceeding standards (proficient or exceeding under the OETF), shall be evaluated at least every five years if the unit member and evaluator consent in writing to such a modified schedule. Unit members on such alternate schedule shall complete an alternative to evaluation as specified below. Should either the evaluator or evaluatee withdraw consent, notice shall be given prior to the first meeting scheduled under Article 13.3.2.

Unit members with an alternate evaluation cycle pursuant to this section shall complete a mutually agreed upon alternative project. Examples include but are not limited to: project based learning, curriculum development, pursuit of advanced degree with research topic to support profession/District/nation, and project based learning to impact target group identified in District's strategic plan. If mutually agreed, the project period may exceed one year, however, it is not the intent of the parties to increase the frequency of evaluation beyond the provisions of Article 13.1.4.1 above.

13.2 Implementation of Evaluation System

- 13.2.1 It shall be the administration's responsibility to advise each certificated employee of the specific assignment on which they will be evaluated, and to designate who will be the evaluator of the unit member.
- 13.2.2 It shall be the responsibility of the unit administrator to hold one or more staff meetings to review evaluation policies and procedures, and to review the evaluation calendar for the year (normally provided by the Human Resources Services and Support).
- 13.2.3 It shall be the evaluator's responsibility to:
 - 13.2.3.1 Meet with the evaluatee to initiate the evaluation process.
 - 13.2.3.2 Meet with evaluatee to establish the specific plan for the evaluation program in accordance with Section 13.6.2 herein. For first and second year teachers, the evaluator shall assist the evaluatee in identifying a primary focus on two (first year) or three (second year) of the standards

- identified in order to target support necessary for the evaluate (See Section 13.6.2).
- 13.2.3.3 Review and approve the evaluation plan submitted from the evaluatee.
- 13.2.3.4 Provide assistance to the evaluatee to achieve agreed upon objectives as stated in evaluation plan /SMARTe goals including providing support to new teachers/evaluatees in the targeted areas identified pursuant to Section 13.2.3.2.
- 13.2.3.5 Make scheduled and unscheduled observations, and meet with evaluatee for post-observation conferences and completion of observation forms and evaluation report.
- 13.2.3.6 Meet legal and District calendar dates as they relate to staff evaluation as stated in this article and in Appendix 6 Evaluation Handbook.
 - 13.2.3.6.1 Assess progress.
- 13.2.3.7 Meet with evaluatee for final review by the last workday in April. The evaluator shall report evaluation findings both in consultation and in writing to the employee so as to:
 - Commend outstanding performance ("exceeding").
 - Assist evaluatee with a satisfactory ("proficient") rating to improve.
 - Assist evaluatee with unsatisfactory ("developing" or "beginning") rating by identifying areas of deficiency and providing assistance to improve.
 - Notify evaluatee whose performance remains unsatisfactory ("developing" or "beginning") of a pending reassignment or dismissal in order to meet all legal requirements.
- 13.2.4 It shall be the evaluatee's responsibility to:
 - 13.2.4.1 Meet with evaluator to initiate the performance evaluation process as described herein.
 - 13.2.4.2 Meet with the evaluator to establish the specific plan for the evaluation program. The plan should include specific performance standards and concomitant criteria that address the following:
 - Planning and preparing rigorous standards-aligned lessons.
 - Building a supportive and challenging learning environment.
 - Teaching to ensure ownership and mastery for ALL students.
 - Reflecting and revising to improving student outcomes.
 - Engaging and supporting all students in learning.
 - Understanding and organizing subject matter for learning, planning and designing instruction, and learning experiences for all.
 - Assessing student learning, developing as a professional educator, and other duties and responsibilities.
 - For first and second year teachers, the plan should have a primary focus on two (first year teacher) or three (second year teacher) of the standards identified herein in order to target support necessary for the evaluatee.

- 13.2.4.3 Record tentative standards of performance and activities on the District evaluation form.
- 13.2.4.4 Submit the tentatively completed evaluation plan (SMARTe goals/objectives), as referenced in the Evaluation Handbook, review the stated standards of performance and objectives and relate them to District standards, objectives, and participate in the scheduled conference(s) with the evaluator.
- 13.2.4.5 Conduct appropriate programs for meeting standards of performance and completing the activities.
- 13.2.4.6 Seek assistance from evaluator to achieve agreed upon SMARTe goals/objectives as stated in the evaluation plan.
- 13.2.4.7 Assess progress.
- 13.2.4.8 Meet with evaluator for final post-observation conference and completion of observation form and evaluation report.
- 13.2.4.9 Meet all applicable legal and District calendar dates relating to evaluation.
- 13.2.4.10 Meet with evaluator for final review and report by the last workday in April as scheduled by the evaluator.
- 13.2.4.11 Willful absence and/or failure of the evaluatee to cooperate shall not prevent completion of the evaluation process.
- 13.2.5 It shall be the joint responsibility of evaluator and evaluatee to:
 - 13.2.5.1 Meet and review the evaluation process to be followed and to reach an agreement.
 - 13.2.5.1.1 Should agreement not be reached on the evaluation process or on the appropriateness of the designated evaluator, the procedures for resolution of differences shall be followed (See Section 13.5).
 - 13.2.5.2 Meet periodically to assess progress.
 - 13.2.5.3 Meet and review in detail the final evaluation report. Should the evaluatee not agree with the report, the procedures for resolution of differences shall be followed.

13.3 Evaluation Sequence

13.3.1 Evaluation Process

The evaluation process begins on the first day of the employee's work year. Each evaluatee shall specify their plan to achieve District standards and any District initiatives. District standards serve to establish the broad purposes of our schools and to guide each employee in the planning and evaluation of their work. The evaluation process outlined in Appendix 1 of this Article shall be followed.

- 13.3.2 The unit administrator holds one or more staff meetings to review evaluation policies, procedures and the evaluation calendar at the beginning of the employee's work year.
- 13.3.3 Each work site develops work site objectives through staff participation.
- 13.3.4 Assistance in defining individual objectives may be found by reference to the California Standards for the Teaching Profession, the Oakland Effective Teaching

- Framework, District initiatives, unit objectives, District standards, site or departmental plans, individual job description, areas of pupil achievement and environmental control, performance areas needing improvement, and additional assigned duties and responsibilities.
- 13.3.5 For each SMARTe goal/objective, the evaluatee prepares an evaluation plan form. SMARTe goal/objective are prepared for cognitive items and may be prepared for effective items related to pupil progress, suitable learning environment, instructional strategies and techniques, adherence to curricular objectives, and other duties and responsibilities.
- 13.3.6 Each evaluatee prepares an evaluation plan/ SMARTe goal forms. The body of the forms are completed as follows:
 - 13.3.6.1 Items I through VI are required evaluation items.
 - 13.3.6.2 Item VII, Other Enter any additional items that are considered appropriate for evaluation.
 - 13.3.6.3 Support Requirements Enter the required administrative and/or logistic support.
 - 13.3.6.4 Mitigating Factors Enter circumstances, which may limit or inhibit success in achieving satisfactory control and learning environment.
 - 13.3.6.5 Other Duties and Responsibilities:
 - List only those to be evaluated.
 - Mitigating Factors Enter circumstances, which may limit or inhibit success in satisfactorily carrying out duties and responsibilities.

13.4 Completed Forms

- 13.4.1 Evaluatee shall provide the completed evaluation plan (SMARTe goals) to evaluator by the 18th work day after the beginning of the employee's work year or the 18th day after classes begin, whichever is later.
- 13.4.2 Initial conference between evaluator and evaluatee to review standards of performance shall be completed by the 27th work day after the beginning of the employee's work year or the 27th day after classes begin, whichever is later.
- 13.4.3 Evaluator shall have the appropriate Evaluation Report form initiated for evaluatee.
- 13.4.4 Evaluator and evaluatee shall review the evaluation process to be followed and the evaluation plan (SMARTe goals) completed by evaluatee.
- 13.4.5 Agreement between evaluator and evaluatee shall be reached on the number, priority, and appropriateness of the objectives/goals and activities, on standards of performance, on assessment procedure, on mitigating factors, and on common responsibilities and support that will be available to the evaluatee.
- 13.4.6 It is to be understood that there shall be periodic reviews of progress and changes in objectives/goals, in standards of performance, and procedures as conditions change.

13.5 Disagreement

13.5.1 When agreement cannot be reached between evaluator and evaluatee at the first conference, the second conference between the two shall be completed by the 30th workday after the beginning of the employee's work year.

- 13.5.2 When an agreement cannot be reached at the conclusion of the second conference, there may be a conference of both parties with the unit administrator, provided they are not the evaluator, or the optional step may be invoked. In either event, this step shall be completed by the 33rd workday after the beginning of the employee's work year.
- 13.5.3 The unit administrator has the option at this time to replace the designated evaluator, or to assume the evaluator role.
- 13.5.4 When a different person becomes the evaluator, the two parties must start from the beginning of the evaluation cycle.
- 13.5.5 If the optional step is chosen, each party may select a conferee. The two parties and the conferee shall meet jointly at the site to resolve differences and/or affect a compromise. Selected conferees shall be certificated personnel currently employed by the District. Conferees shall not be persons in direct line of responsibility, or persons who are official representatives of employee groups.
- 13.5.6 Should disagreement persist after the above steps have been taken, a written report shall be submitted by the evaluatee to the Associate Superintendent or to the Superintendent, as appropriate, who will review and make the final decision. Said review shall be completed by the 37th workday after the beginning of the employee's work year.

13.6 Activities During Evaluation Period

- 13.6.1 During the evaluation period, the evaluator shall:
 - 13.6.1.1 Notify evaluatee that there will be scheduled and unscheduled observations. A variety of techniques may be used, but they should be of sufficient duration and quality to make significant contributions to the process of performance evaluation.
 - 13.6.1.2 Confer with evaluatee within five (5) workdays following the observation. Evaluator and evaluatee shall discuss the observation and complete the Observation Form. Appropriate entries shall be made on the appropriate Evaluation Report/Observation Form.
 - 13.6.2 The evaluator may also wish to enter the following on the Evaluation Report/ Observation Form: comments, remarks on evaluation plan, review of progress, changes in objectives, etc.
 - 13.6.2.1 Evaluation Report includes those items which have an effect on the plan, e.g., "the evaluatee was absent for a month," "the evaluator did not receive the support needed to assist the evaluatee," etc.
 - 13.6.2.2 Review of progress refers to meetings of evaluator and evaluatee to assess progress, other than to discuss observations.
 - 13.6.2.3 "Changes in objectives" refers to any change agreed to by evaluator and evaluatee, including the addition or deletion of objectives.
 - 13.6.2.4 Evaluatee, prior to the final Evaluation Report, shall complete "Degree of Achievement" on Evaluation Plans/ SMARTe Goals 1 and 2. Comments may include reasons for variations from the Evaluation Plan/ SMARTe Goals.
 - 13.6.2.5 Evaluations shall be conducted in private conferences.

13.7 Probationary Interim Report

- 13.7.1 A Formal mid-year conference for first-year probationary personnel is to be completed by the second Friday in December.
- 13.7.2 The Evaluator shall prepare two sets of Evaluation Report forms at the beginning of the evaluation period. Entries shall be made on both sets of forms; one set to be used for the Interim Assessment Report for probationary-first year employees; one set to be used for the final Evaluation Report.
- 13.7.3 By the second Friday in December, the evaluator shall meet with the evaluatee and review the Interim Assessment Report. They shall discuss the strengths and/or weaknesses of the evaluatee's performance, with commendations and/or recommendations for improvement, and shall identify the procedures they will follow to facilitate such improvement.

13.8 Formal Evaluation Report

- 13.8.1 By the last workday of April, the evaluator shall have met with evaluatee and reviewed the Evaluation Report.
- 13.8.2 The Evaluation Plan, as outlined in the SMARTe Goal Forms (see Appendix 1) shall be reviewed. Ratings on the Evaluation Report, including any comments, are based on the Evaluation Plan developed at the beginning of the evaluation period and in accordance with Section 13.2.4.2 herein.
 - 13.8.2.1 Engage/support all students in learning are rated in relationship to the established standards of performance, the Oakland Effective Teaching Framework ("OETF") domains, contained in the Certificated Evaluation Observation Form.
 - 13.8.2.2 Create/maintain effective student learning environment is rated in relationship to the established standards for individual certificated personnel contained in the Certificated Evaluation Observation Form.
 - 13.8.2.3 Understand/organize subject matter for learning is rated in relationship to the items contained in the Certificated Evaluation Observation Form.
 - 13.8.2.4 Plan/design instruction/learning experiences for all are rated in relationship to the item contained in the Certificated Evaluation Observation Form.
 - 13.8.2.5 Assess student learning is rated in relationship to the items contained in the Certificated Evaluation Observation Form.
 - 13.8.2.6 Develop as a professional educator is rated in relationship to the items contained in the Certificated Evaluation Observation Form.
 - 13.8.2.7 Other duties and responsibilities are rated in relationship to the items contained in the Certificated Evaluation Observation Form.
 - 13.8.2.8 Comments are to be used to commend an exemplary performance, to assist a proficient evaluatee to improve, and to provide assistance to facilitate improvement of an evaluatee who has a beginning or developing rating.
 - 13.8.2.9 The evaluator shall discuss any recommended personnel action with the evaluatee.

13.9 Review of Unsatisfactory Rating

- 13.9.1 When an evaluatee wishes to review a "beginning" and/or "developing" rating, an additional conference with the evaluator shall be held by the third Friday in April. If the evaluator, however, is not the site administrator, the evaluatee and the evaluator shall confer with the site administrator by the fourth Friday in April.
- 13.9.2 When an evaluation contains one or more beginning and/or developing rating (unsatisfactory rating), the evaluatee, after the additional conference with the evaluator and/or unit administrator, may submit a letter of rebuttal to be attached to the final evaluation report.

The letter of rebuttal shall be submitted to the evaluator by the first Friday in May. The final evaluation report with the letter of rebuttal attached shall be forwarded immediately to the Director of Human Resources or their designee for review and response.

- 13.9.3 Review and response to the letter of rebuttal by Associate Superintendent or Superintendent shall be completed within ten (10) workdays after the final date for submitting the letter of rebuttal.
 - 13.9.3.1 The Director of Human Resources or their designee's response shall be attached to the final report. Copies of the complete record (Evaluation Report, letter of rebuttal, official response) shall be filed in the unit member's personnel file.

13.10 Probationary and Tenured Personnel

Probationary and tenured personnel whose retention is doubtful shall be so notified in writing. Subsequently, the Employer shall proceed in accordance with the provisions of the collective bargaining agreement between the parties and Education Code provisions.

13.10.1 Personnel Files

- 13.10.1.1 The Human Resources Services and Support shall establish and maintain file(s) for each member of the unit. The file(s) shall be the official District repository for evaluation records.
- Materials in personnel files of unit members, which may serve as a basis for affecting the status of their employment, are to be available for the inspection of the member involved. Such material is not to include ratings, reports, or records which were obtained prior to the employment of the person involved, were prepared by identifiable examination committee members, or were obtained in connection with a promotional examination.
- 13.10.1.3 An employee's personnel file will be accessible for review on the following basis:
 - 13.10.1.3.1 The employee shall notify the Human Resources to schedule an appointment.
 - 13.10.1.3.2 The appointment shall be scheduled based on a twenty-four hour notice.
 - 13.10.1.3.3 Emergency exceptions to the above will be allowed based on the individual scheduling of the Department and the employee.

- 13.10.1.4 Every member of the unit shall have the right to inspect such materials upon request, provided that the request is made at a time when such member is not actually required to render services to the employing District. A representative of the employee's choosing may, at the request of said employee, accompany the employee in this review. The review shall be made in the presence of the administrator or the administrator's designee responsible for the safekeeping of this file.
- 13.10.1.5 Information of a derogatory nature, except material mentioned in 13.10.1.2 above, shall not be entered or filed unless and until the employee is given notice and an opportunity to review and comment thereon.
 - 13.10.1.5.1 Such review shall take place during normal business hours and the employee shall be released from duty for this purpose without salary reduction. An opportunity shall be provided within 15 workdays of the date of receipt of such material by the administrator. Any derogatory material regarding evaluation of an employee cannot be used in a unit member's incompetency case after four years. Such material shall be removed from the personnel file after four years from date of entry, and shall be sealed and deposited in the Superintendent's Office.
- An employee shall have the right to enter, and have attached to any such derogatory statement, their own comments thereon. The unit member shall acknowledge that they have read such material by affixing their signature on the actual copy to be filed, with the understanding that such signature merely signifies that the material to be filed has been read, and does not necessarily indicate agreement with its contents. In the event that an employee refuses to affix their signature to the document, a statement to this effect, together with that of a witness, shall be attached to the document by the Director of Human Resources Services and Support or appropriate administrator.

13.11 Complaints Against Employees

- 13.11.1 Criticisms shall be conducted in a private conference.
- 13.11.2 An administrator receiving a formal complaint, which may lead to disciplinary action against any employee under their supervision, shall notify the employee of said complaint in writing within ten days.
- 13.11.3 All information forming the basis for disciplinary action shall be made available to the employee.
- 13.11.4 Material that may, as a result of any proven complaint, be entered into a unit member's personnel file, shall be processed in accordance with Section 13.10.1.
- 13.11.5 In the event of a complaint or charges of a derogatory nature which is ultimately not proven, no record shall be kept.
- 13.11.6 When charges and complaints are made directly to the Board, it shall in turn direct the Superintendent to notify the employee(s) of such charge or complaint.
 - 13.11.6.1 The Superintendent shall investigate the complaint and report the results to the Board of Education with a recommendation. Prior to taking any action adverse to the employee's interests, the Board shall afford such employee(s) an opportunity to discuss the matter directly with the Board.

13.12 Site Instructional Assistance Program (SIAP)

13.12.1 Purposes

- 13.12.1.1 To improve overall instruction and programs at each District site.
- 13.12.1.2 To provide input and feedback for the improvement of instruction at each grade and/or department level.
- 13.12.1.3 To provide multiple sources of assessment to achieve these purposes.
- 13.12.2 Tenured employees shall participate in the SIAP at least every two years alternating with their Evaluation year, probationary employees at least once each year during the probationary period.
- 13.12.3 Examples of the forms and questionnaires to be used in this procedure shall be appended to this Agreement as Appendix 813.12.3) " \s "Appendix 8" \c 8.
- 13.12.4 After the first year of use, a review committee composed of four teachers to be appointed by OEA and four administrators will review, and if necessary, revise the forms.
- 13.12.5 Implementation of the Instructional Assistance Program
 - 13.12.5.1 Site administrators will obtain valid student or parent input into the assessment of Educational Programs and site personnel. District approved instruments and procedures based upon professionally recognized models and samples, including the use of student surveys conducted by the teacher in grades four and above, and by parent surveys in grades three and below.
 - 13.12.5.2 Teachers and the site administrator shall jointly plan the means by which the implementation procedure will be employed.

13.12.6 General Provisions

- 13.12.6.1 Program assessment data and information shall be kept in a confidential file at the school site, and shall only be available to the unit member and their union representative, and the site administrator(s).
- 13.12.6.2 Said data and information shall not be made available to any other District personnel without the prior written approval of the unit member and the site administrator(s).
- 13.12.6.3 Said data and information shall not be used in any unit member evaluation procedure, disciplinary process, or in any other punitive action.
- 13.12.6.4 If the unit member believes that said data and information has been used in a subsequent evaluation, they may challenge the evaluation, and if they can show that said evaluation was based upon or connected to the data/information, said evaluation shall be destroyed.
- 13.12.6.5 The SIAP shall not be implemented unless it is coupled with a process by which teachers and other unit members shall assess the various site administrators. To improve overall site administration, the frequency of which in each school shall be the same as the application of the SIAP for teachers and other unit members.

ARTICLE 14 - GRIEVANCE POLICY

14.1 General

- 14.1.1 It is the policy of the Employer to develop and practice reasonable and effective means for resolving difficulties arising among employees, to reduce potential areas of grievance, and to establish and maintain recognized two-way channels of communication between staff and administration.
 - 14.1.1.1The purpose of this article is to set forth the procedures provided in the District for the prompt and equitable adjustment of differences, which may arise, and to provide equitable solutions to those problems at the lowest possible administrative level. The parties agree that confidentiality at any level shall be maintained.
- 14.1.2 This article provides procedures for consideration of grievances as they relate solely to contract disputes, and as they relate to this Agreement.
- 14.1.3 Provisions in this article shall not preclude administrative responsibility and authority for implementing Employer policies and operating schools.
- 14.1.4 The Association has the right to file a grievance on behalf of a group of affected bargaining unit members that have a complaint on identical issues.

14.2 Definitions

- 14.2.1 Problem is defined as a concern on the part of one or more employees, which arises from the application or interpretation of a provision of this Agreement.
- 14.2.2 Grievant is defined as a unit member who is filing a grievance.
- 14.2.3 Employee is defined as an employee receiving compensation and belonging to the unit herein. Independent contractors are not included.
- 14.2.4 Party in Interest is defined as a person who might be required to take action or against whom action might be taken in order to resolve the claim.
- 14.2.5 Conferee is defined as another District employee or representative from the exclusive organization (bargaining agent).
- 14.2.6 Immediate Administrator is defined as the administrator with immediate jurisdiction over the employee who is filing the grievance.
- 14.2.7 Day is defined as duty day for the unit member relating to time frames applicable to the grievant and/or duty day for the administrator responsible for responding to the grievance. In the event that the administrator responds to a grievance at a time when the grievant is not on duty, the time line shall be suspended until such time as the grievant returns to duty, unless the grievant requests otherwise.

14.3 Informal Resolution of the Problem

- 14.3.1 Unit Members are encouraged to ask their immediate administrator for assistance on matters that relate to their duties. A problem that arises should be discussed in a conference between the unit member and the administrator to whom they are immediately responsible.
- 14.3.2 At least one conference between the unit member and the immediate administrator concerning the problem shall be held before formal grievance procedures are invoked. The individuals should make every effort to resolve the difficulty at the

point of origin and should not hesitate to obtain information or advice from other appropriate sources in an attempt to arrive at a satisfactory solution.

14.4 Step A - Pre-Formal Stage

- 14.4.1 A Unit Member who believes they have a problem, which could result in a grievance, and who wishes to follow the prescribed procedure, is required to request an appointment to present the problem to the immediate administrator within 20 days after the unit member knew or reasonably should have known of the circumstances, which formed the basis for the problem.
- 14.4.2 The immediate administrator will schedule at least one private conference with the unit member within five days after request for the appointment, and every effort should be made to resolve the matter within five days from the date of the conference.
- 14.4.3 The final determination at this stage is an oral resolution of the problem between the unit member and the immediate administrator.
- 14.4.4 If the unit member is not satisfied with the administrator's response or if there is no response in the five-day period, the unit member may file a formal grievance no later than seven days after hearing the response or lack of a response.

14.5 Level I

- 14.5.1 A problem not resolved at the pre-formal stage may be presented by the employee in the form of a written statement of grievance to the administrator to whom he/she reports. After this grievance procedure has been invoked, continued discussions between the unit member and the administrator are encouraged.
- 14.5.2 Effective within sixty days following ratification of this agreement, the parties shall agree upon a jointly developed electronic grievance form and tracking system to be maintained by the employer. Until this occurs, the language in the 2014-17 agreement shall be followed. Once the new electronic grievance form and tracking system is implemented, the written Level 1 grievance number, date filed, grievant name, and name of the person against whom the grievance is filed shall be entered on an electronic form accessible to the grievant, the OEA grievance chair, the OEA President, CTA staff, the site leader or other individuals involved in the grievance, the immediate supervising Central Administrator (e.g. Executive Director/Assistant Superintendent/Network Superintendent), the Deputy Chief of Talent, the Executive Director of Labor Strategy, and the Superintendent.
- 14.5.3 If the pre-formal procedure was not utilized, there shall be a conference with the Level I administrator and the grievant.
- 14.5.4 If the pre-formal procedure was utilized, either party may waive the conference.
- 14.5.5 The unit member and administrator may have a conferee present during all conferences.
- 14.5.6 The administrator must respond in writing within seven (7) days after receipt of the grievance indicating the decision and supporting reasons. If the administrator fails to render a written decision within ten (10) days, the grievance shall be escalated to Level II.

14.6 Level II

14.6.1 A grievance not resolved at Level I, or escalated from Level I, shall be presented by the grievant within seven (7) days from receipt of the Level I decision to the unit

member's Central Administrator (e.g. Leadership Director/Assistant Superintendent/Network Superintendent).

14.6.2 The Central Administrator (e.g. Executive Director/Assistant Superintendent/Network Superintendent) or designee shall have a conference with the grievant within ten (10) days of receipt of the grievance in an effort to resolve the grievance, unless mutually waived. The Central Administrator (e.g. Executive Director/Assistant Superintendent/Network Superintendent) shall respond to the grievant in writing indicating the decision and supporting reasons within ten (10) days of the conference or within ten (10) days of receipt of the grievance if the conference is mutually waived. If the administrator fails to render a written decision within twenty-one (21) days, the grievance shall be escalated to Level III.

14.7 Level III

- 14.7.1 A grievance not resolved at Level II or escalated from Level II shall be considered as presented to the Superintendent by the grievant within seven (7) days after receipt of Level II decision. The Superintendent or Superintendent's designee shall review and investigate the matter, and may engage in a conference with the aggrieved unit member.
- 14.7.2 The Superintendent or designee shall respond in writing within seven days after receipt of grievance indicating decision and supporting reasons. If the Superintendent or designee fails to render a written decision within ten (10) days, the matter shall be placed on the agenda for the regularly scheduled meeting between the OEA President or designee and the Superintendent.
- 14.7.3 Upon mutual agreement, the grievance will be referred to mediation. In the event there is no agreement, the grievance will be referred to arbitration.

14.8 Mediation

The intent of this section is to provide a mechanism to mediate and resolve grievances in a satisfactory manner. The mediator shall be selected from a mutually agreed upon panel and shall facilitate dialog and help the parties reach a mediated settlement. The mediator shall have no power to add, modify or delete any provision of the collective bargaining agreement.

- 14.8.1 Recommendations of the mediator shall be advisory and non-precedent-setting. Neither party shall cite the recommendation(s) in future grievances.
- 14.8.2 The mediator shall present recommendations in writing to both parties in the form of a proposed settlement agreement within sixty (60) days of the mediation. Upon receipt of the mediator's recommendations, either party may appeal the recommendation by referring the matter to arbitration. If neither party appeals, the recommended settlement will be implemented.

14.9 Binding Arbitration

- 14.9.1 In the event the grievant is not satisfied with the mediator's decision at Level III, he/she may, within seven (7) days, request of the Association that the grievance be submitted to a neutral arbitrator from list in 14.9.3.
- 14.9.2 The Association, by written notice to the Superintendent or designee within fifteen (15) days after receipt of the request from the aggrieved person, may submit the grievance to binding arbitration. The Association and the employer shall select an arbitrator using the procedures in Section 14.9.3.

- 14.9.3 Upon invocation of arbitration, the parties shall attempt to seek an agreement on a suitable arbitrator. If no agreement is reached within 10 days of the District's receipt of the Association's request for arbitration, the parties will jointly request a list of arbitrators from the State Mediation and Conciliation Service (SMCS) and will attempt to reach agreement from this list. If the parties are unable to reach such agreement within five (5) days of receipt of the SMCS list, an arbitrator shall be selected as follows:
 - The Association will strike one arbitrator from the SMCS list.
 - The District will strike one arbitrator from the SMCS list
 - The parties will continue alternating strikes in this manner until one arbitrator remains, who then is the selected arbitrator.

The party striking first from the SMCS list shall alternate with each grievance, beginning with the first grievance on which the Association requests arbitration following ratification of this agreement.

- 14.9.4 Any award of the arbitrator shall be binding on the grievant, the Association and the District.
- 14.9.5 It shall be the function of the arbitrator to make an award, if necessary, which will resolve the grievance. The arbitrator shall be subject to the following limitations:
 - 14.9.5.1 The arbitrator shall have no power to add to, alter, subtract from, disregard, change, or modify any terms of this Agreement; but shall determine only whether or not there has been a violation, misapplication, or misinterpretation of this Agreement as alleged by the grievant.
 - 14.9.5.2 The arbitrator, when necessary for interpretation of the contract, may use accepted rules of interpretation or construction of contracts.
 - 14.9.5.3 The arbitrator shall have no power to establish salary structures, but has power to interpret words or phrases in the contract and related to the salary structure when necessary to resolve disputes.
 - 14.9.5.4 The award of the arbitrator shall be based solely upon the evidence and arguments presented in the presence of the parties, and upon any post-hearing briefs of the parties.
 - 14.9.5.5 The arbitrator shall have no power to change any practice, policy or rule.
 - 14.9.5.6 The standard of review for the arbitrator is whether the employer acted in violation, misapplication and misinterpretation, was negligent or intentional in nature.
 - 14.9.5.7 The arbitrator shall not consider any issue raised by the parties unless it has been raised by the parties at Level III of this Grievance Policy.
 - 14.9.5.8 The arbitrator shall have no power to recommend or resolve the termination of services or failure to reemploy any member of the unit.
- 14.9.6 All fees and expenses of the arbitrator shall be shared equally by the employee and the Association.
- 14.9.7 If the Employer claims that a grievance should be dismissed; for example, it falls outside the scope of the procedure or was filed or processed in an untimely manner,

such claim shall be heard and promptly ruled upon by the arbitrator prior to any hearing on the merits of the grievance. Upon the request of either party, there shall be a stay/continuance, the duration to be determined by mutual agreement between such a ruling and any further proceedings, which may be necessary. If a stay/continuance is not requested, the arbitrator may immediately conduct any further proceedings, which may be necessary.

- 14.9.8 If the employer should choose to refuse to arbitrate a dispute, nothing in this section shall preclude the Association from seeking, through appropriate administrative or judicial proceedings, to compel the District to proceed to arbitration.
- 14.9.9 The arbitrator shall have no power to render an award before thirty (30) days after the effective date of this Agreement. If a grievance is filed before the termination of this Agreement, it shall be continued until its completion pursuant to this Agreement.

14.10 General Provisions

- 14.10.1 For purposes of efficiency, grievances involving similar issues may be consolidated by either party.
- 14.10.2 Unit member and all other persons appropriately involved in a grievance shall be free from restraints, interference, coercion, discrimination, or reprisal. Employer grievance forms shall be used for processing all grievances. Each written statement of grievance shall provide the following information and bear the signature and signature date of the grievant.
 - 14.10.2.1 Listing of provisions of the contract alleged to have been violated or misapplied.
 - 14.10.2.2 Description of the specific grounds of grievance, including names, dates and places necessary for complete understanding.
 - 14.10.2.3 Listing of reasons as to why the proposed resolution provided at the prior level is unacceptable, with specific and factual data in support thereof.
 - 14.10.2.4 Listing of specific actions requested of the Employer to remedy the problem.

14.10.3 Time Limits

- 14.10.3.1 Grievance at Level I must be filed within twenty days after the employee knew or reasonably should have known of the circumstances that formed the basis for the grievance.
- 14.10.3.2 Appeals to Levels II and III must be filed no later than seven (7) days following the date of receipt of a decision at the prior level. Grievances filed at Levels II and III pursuant to Section 14.10.3.1 must be filed within twenty (20 days after the grievant knew or reasonably should have known of the alleged violation or misapplication of the contract provisions(s). Appeals to arbitration must be filed no later than twenty-two (22) days following the date of receipt of a decision at the prior level.
- 14.10.3.3 Time limits may be extended by mutual consent due to the inability to schedule conferences and responses within the above time frames.
- 14.10.3.4 A decision rendered at any step in these proceedings shall become final unless appealed within the time limits specified.

- 14.10.3.5 When it is not possible to conclude Levels 1, 2 or 3 before the end of the school year, either party may stay the grievance until the start of the next school year.
- 14.10.4 A unit member may present grievances to the appropriate administrator and have such grievances adjusted without the intervention of the exclusive representative as long as adjustment is reached prior to arbitration. Such adjustments are not inconsistent with the terms of a written agreement then in effect. Provided that the unit member will not agree to a resolution of the grievance until the exclusive representative has received a copy of the grievance and the proposed resolution and has been given the opportunity to file a response within five (5) days.
- 14.10.5 The grievant may terminate a grievance at any time by giving written notice to the administrator involved at the current level of the grievance.
- 14.10.6 A unit member may terminate any grievance conference until a conferee may be present. Securing a conferee shall not be construed as violating any time limits. Termination of the conference shall be confirmed in writing by the unit member.
- 14.10.7 A unit member may be represented at all stages, including Section 14.4, Step A Pre-Formal Stage of the grievance procedure, by himself/herself, or at their option, with a representative selected by the Association. The Association shall have the right to be present and state its views at all stages of the grievance procedure.
- 14.10.8 If a grievance arises from an authority higher than a Level 1 administrator, e.g., leave requests, payroll errors, insurance program benefits; the grievance may be filed at the appropriate step of the grievance procedure.
 - 14.10.8.1 Personnel grievances shall be filed with the Director of Human Resources or their designee.
 - 14.10.8.2 The Director of Human Resources shall be considered the Level 2 respondent for personnel grievances.
 - 14.10.8.3 The Level 2 decision shall be appealable directly to Level 3, the Superintendent or their designee.
- 14.10.9 The Association may have the right to file a grievance in those areas that speak specifically to Association rights.
- 14.10.10 If a grievance is denied at any level prior to Level 3 on the basis that it is not considered as a grievance within the definition contained in this article, it may then be appealed in its entirety to the next level for consideration of the question of its grievability and its merits.
- 14.10.11 Failure to submit an appeal to the next level within the specified time limit indicates that the grievant waives the right to appeal further on the grievance at issue. Failure of unit member to present a grievance within the specified time limit shall render the grievance null and void.
- 14.10.12 Documents relating to the processing of grievances shall be filed separately from a unit member's personnel file. All records of proceedings shall be retained in the office of the Human Resource Division and shall be held confidential.
- 14.10.13 Grievance forms, which have been mutually agreed upon by the employer and the Association, will be prepared by the employer and made available at the Human Resource Division and the Association office.

- 14.10.14 Meetings shall be held at the local site or office for Levels I, II and III. At arbitration other localities may be designated by the arbitrator.
- 14.10.15 Every reasonable effort shall be made to schedule meetings for the processing of grievances at times when they shall not interfere with the instruction of students.
- 14.10.16 The Employer shall permit a reasonable amount of release time without loss of compensation for designated Association representatives for the purpose of processing grievances.
- 14.10.17 In the event a grievance is filed at such a time that it cannot be processed through all the steps in this grievance procedure by the end of the school year and, if left unresolved until the beginning of the following school year could result in harm to the aggrieved employee, the time limits set forth herein shall be reduced by mutual agreement so that the procedure may be completed prior to the end of the school year or as soon as practical.
- 14.10.18 Written responses by the administrator to the grievant shall be hand-delivered to the employee at the employee's worksite or sent by certified mail to the employee's address of record.
- 14.10.19 Evaluations are not subject to the grievance procedures unless the evaluation procedures have not been followed.
- 14.10.20 A copy of the grievance form is found in Appendix 15 (F5)14.10.20)" \s "F5" \c 8 of this Agreement.

ARTICLE 15 - CLASS SIZE AND CASELOADS

15.1 Definitions

- 15.1.1 Elementary Schools: Class size is the number of enrolled pupils assigned to each classroom teacher.
- 15.1.2 Secondary Schools: Class size is the number of enrolled pupils assigned to each classroom teacher on a daily basis.
- 15.1.3 Adult Schools: Class size is determined by the number of students in attendance each hour the class is scheduled.

15.2 Class Size Procedures

- 15.2.1 Beginning Grace Period: Elementary Schools at the beginning of the first semester only, monitoring and achieving maximum class size shall be completed within ten (10) school days.
- 15.2.2 Secondary Schools: At the beginning of the first semester, the District shall take action to achieve maximum class size within fifteen (15) school days. At the beginning of the second semester, the District shall take action to achieve maximum class size within ten (10) school days.
- 15.2.3 In the event the class size maximums are exceeded, the District agrees to take one or more of the following steps:
 - 15.2.3.1 Reclassification of the school where the class size maximums have been exceeded.
 - 15.2.3.2 Transfer of pupils to their neighborhood schools.
 - 15.2.3.3 Transport of pupils to schools that do not have overcrowding problems.
 - 15.2.3.4 Nothing in this subsection shall relieve the District of its obligations to maintain the class size maximums contained herein.

15.3 Exceptions to Maximum Class Size

Exceptions to class size maximums may be taken when space is inadequate and/or when health and safety of students could be imperiled. This exception shall only exist until arrangements are made pursuant to Section 15.2 above to maintain class size maximums.

- 15.3.1 Emergency Exceptions (See "Emergency" definition in Article 3). In the event of an emergency, the individual class size maximums may be exceeded by up to two (2) students provided the following procedure is followed:
 - 15.3.1.1 The Principal/Site Leader may assign a student to an individual teacher's class.
 - 15.3.1.2 Principals/Site Leaders shall make every effort to maintain all classes within a grade level as nearly equal in size as possible.
 - 15.3.1.3 Class maximums must be re-established as soon as the emergency ceases.
 - 15.3.1.4 The Joint OEA/OUSD Special Education Committee shall make recommended changes to this section regarding special needs students assigned to general education classes and the nature of inclusive practices. Those

recommendations shall be communicated to the bargaining teams for consideration as amendments to this section.

15.4 Categories – Elementary

The Employer shall not exceed the following class size maximums.

15.4.1 Elementary Schools - Maximum Class Sizes Base Program

TK/Kindergarten	27
Grades 1-3	30
Grades 4-6	31

15.4.2 Transitional Kindergarten (TK) through Grade 3

Under the current State Local Control Funding Formula (LCFF), school districts must make annual progress toward average class sizes of 24:1 per school site across grades TK through 3, and must reach that target by 2020-21. Commencing with the 2015-2016 school year, average class sizes across grades TK through 3 shall be 24:1 per school site based on the current 2020-21 LCFF target. Thereafter, class size for grades TK through 3 may be determined by future state laws and regulations.

Effective with the 2016-17 school year, the District shall make every effort regarding staffing and effective allocation of resources in a manner intended to achieve individual class sizes in schools with an unduplicated pupil percentage in excess of ninety (90) which should not exceed the following:

TK/Kindergarten	24
Grade 1-3	27
Grade 4-5	30

15.5 Categories - Secondary Schools - Maximum Class Sizes

Maximum Pupil Contact

	Per Day
Basic Program	Class Size
English	32 x 5 = 160
Foreign Language	32 x 5 = 160
Mathematics	32 x 5 = 160
Social Science	32 x 5 = 160
Non-Lab Science	$32 \times 5 = 160$
Lab Science	31 x 5 = 155
Business Education (word processing)	Available Work Area or
	34 x 5 = 170
Business Education (Other)	33 x 5 = 165
Industrial Arts	Available Work Area
Homemaking	29 x 5 = 145
Music	$52 \times 5 = 260$
Physical Education	52 x 5 = 260

Arts - Crafts	$27 \times 5 = 135$
Arts - Fine	$30 \times 5 = 150$
ROTC	35 x 5 = 175

15.5.1 Beginning with the 2019-20 school year, the class size maximums in sections 15.4 and 15.5 above shall be reduced by 1 student at schools with an unduplicated pupil percentage in excess of ninety (90).

a. Elementary Schools - 2019-20 Maximum Class Sizes Base Program

Schools with an unduplicated pupil percentage in excess of ninety (90)

TK/Kindergarten	26
Grades 1-3	29
Grades 4-6	30

b. Categories - Secondary Schools - 2019-20 Maximum Class Sizes

Schools with an unduplicated pupil percentage in excess of ninety (90)

	Maximum Pupil Contact
	Per Day
Basic Program	Class Size
English	31 x 5 = 155
Foreign Language	31 x 5 = 155
Mathematics	31 x 5 = 155
Social Science	31 x 5 = 160
Non-Lab Science	31 x 5 = 155
Lab Science	30 x 5 = 150
Business Education (word processing)	Available Work Area or
	33 x 5 = 165
Business Education (other)	32 x 5 = 160
Industrial Arts	Available Work Area
Homemaking	28 x 5 = 140
Music	51 x 5 = 155
Physical Education	51 x 5 = 155
Arts - Crafts	26 x 5 = 130
Arts - Fine	29 x 5 = 145
ROTC	34 x 5 = 170

15.5.2 Effective June 30, 2021, unless otherwise mutually agreed by the parties, the class size maximums in sections 15.4 and 15.5 above shall be reduced by 1 student at all schools beginning with the 2021-22 school year and carrying forward (This provision would increase the reduction to two (2) students, inclusive of 2019-20 school year reduction of 1 student, at schools with 90% or higher unduplicated pupil percentage.)

a. Elementary Schools - Maximum Class Sizes Base Program Effective June 30, 2021

TK/Kindergarten	26
Grades 1-3	29
Grades 4-6	30

b. Categories - Secondary Schools - Maximum Class Sizes Effective June 30, 2021

	Maximum Pupil Contact Per Day
Basic Program	Class Size
English	31 x 5 = 155
Foreign Language	31 x 5 = 155
Mathematics	31 x 5 = 155
Social Science	31 x 5 = 160
Non-Lab Science	31 x 5 = 155
Lab Science	$30 \times 5 = 150$
Business Education (word processing)	Available Work Area or
	33 x 5 = 165
Business Education (other)	$32 \times 5 = 160$
Industrial Arts	Available Work Area
Homemaking	28 x 5 = 140
Music	51 x 5 = 255
Physical Education	51 x 5 = 255
Arts - Crafts	$26 \times 5 = 130$
Arts - Fine	29 x 5 = 145
ROTC	34 x 5 = 170

c. Elementary Schools - Maximum Class Sizes Base Program Effective June 30, 2021

Schools with an unduplicated pupil percentage in excess of ninety (90)

TK/Kindergarten	25
Grades 1-3	28
Grades 4-6	29

d. Categories - Secondary Schools - Maximum Class Sizes Effective June 30, 2021

Schools with an unduplicated pupil percentage in excess of ninety (90)

	Maximum Pupil Contact
	Per Day
Basic Program	Class Size
English	$30 \times 5 = 150$
Foreign Language	$30 \times 5 = 150$
Mathematics	$30 \times 5 = 150$
Social Science	$30 \times 5 = 150$
Non-Lab Science	$30 \times 5 = 150$
Lab Science	29 x 5 = 145
Business Education (word processing)	Available Work Area or
	32 x 5 = 160
Business Education (other)	31 x 5 = 155
Industrial Arts	Available Work Area
Homemaking	27 x 5 = 135
Music	$50 \times 5 = 250$
Physical Education	$50 \times 5 = 250$
Arts - Crafts	25 x 5 = 125
Arts - Fine	28 x 5 = 140
ROTC	33 x 5 = 165

15.6 Adult Education - Maximum Class Sizes

Basic Adult Education Classes $36 \times 5 = 180$

15.7 Secondary Schools

- 15.7.1 In the secondary schools, exclusive of adult education classes, the total student contact per day is based on five (5) periods at the maximum class size times five (5). An individual period may exceed the maximum per period stated in Section 15.5 above by three (3) students provided that total student contact per day does not exceed the stated maximums in item Section 15.5 above.
- 15.7.2 The parties agree to work together when reconfiguration occurs in order to reduce class size in the middle grades and examine other related issues.

15.8 Combination Classes at Elementary Schools

When an elementary school combination class is being established for the start of a school year, the smallest class within the affected grade levels shall be the combination class, unless the teacher of the combination class prefers otherwise.

15.9 Potential Staffing Allocation and Class Size Reductions

The Employer and the Association shall collaboratively explore ways to achieve staffing allocations and class size reductions to be funded from alternative revenue sources, throughout the term of this contract. These sources may include but shall

- not be limited to Morgan-Hart secondary school class size reduction monies and/or a parcel tax directed to class size reduction.
- 15.9.1 These efforts will be especially designed to address class size reductions not specifically reduced in the preceding provisions of this article.
- 15.9.2 School Improvement Program (SIP) funds shall be utilized for class size reductions in schools not eligible for Desegregation funding if legally eligible for that purpose.

15.10 Special Day Class (SDC)

15.10.1 Effective with the 2017-18 school year, the District shall make every effort regarding staffing and effective allocation of resources in order to make assignments in a manner intended to achieve reasonably balanced caseloads and class sizes assigned to members, which should not exceed the amounts below:

TK-12:

Mild to Moderate: 13
Moderate to Severe 10

- 15.10.2 The District shall monitor class sizes and caseloads. On at least a monthly basis the district and the Association representative(s), designated for this purpose shall meet. When an individual's class size approaches the amount designated above, the following steps will occur:
 - 15.10.2.1 The District shall exhaust all available options to prevent the assignment of additional students that would cause the amounts above to be exceeded.
- 15.10.3 If a member's class size exceeds the amounts above, a written support plan shall be implemented within 20 days from the date a student is assigned to the unit member's class._This written support plan may include, but must not be limited, to one or more of the following:
 - i. Evaluation of the level of special education support staff according to the needs of the class.
 - ii. Identification of school or other environmental resources, etc.
 - iii. Additional planning/preparation time; and/or
 - iv. Additional compensation subject to written approval of the Executive Director of Special Education.
- 15.10.4 To the extent possible, the support plan shall be based on mutual agreement.
- 15.10.5 A monthly written report after each OEA and District meeting will be produced tracking the monitoring of the class sizes/caseloads and actions decided on and followed through with.
- 15.10.6 In no event shall any provisions of this section be construed to support or result in the denial of placement of or services to students in any classroom.
- 15.10.7 The classifications in District document #13 dated November 6, 2014 shall be incorporated as an Appendix, number to be determined at insertion.

ARTICLE 16 - COMMITTEES

16.1 Joint Study Committees

Joint Study Committees (JSC) may be established by mutual agreement of Association and Employer to investigate topics of mutual interest.

- 16.1.1 The purpose of each study may be to gather and evaluate pertinent information and, in some instances, to develop recommendations.
- 16.1.2 A JSC shall function only in an advisory capacity.
- 16.1.3 The findings, conclusions, recommendations and final report of all such committees shall be submitted to the Association and Employer for review.
- 16.1.4 Sizes of JSC may vary inasmuch as the topics for study may differ considerably in complexity. Equal numbers of members and an appropriate number of alternates shall be appointed by the Employer and the Association to serve on such committees. District members and alternates shall be appointed by the Superintendent or designee. Association members and alternates shall be appointed by the Association President.
 - 16.1.4.1 Sizes of JSC may vary inasmuch as the topics for study may differ considerably in complexity. Equal numbers of District and Association members and an appropriate number of alternates shall be appointed.
- 16.1.5 The work of the JSC shall terminate at the time that a final report of its findings and recommendations has been submitted for review, unless the report is referred back to the committee for further consideration.

16.2 Consultation Committees (CC)

The Association has the right to consult on the definition of educational objectives, determination of content, courses and curriculum, selection of textbooks, to the extent such matters are within the discretion of the public school employer under the law.

- 16.2.1 The Association shall delegate proportionately its unit member representatives to the total committee when established by the employer in these areas.
- 16.2.2 All CC reports on major curriculum issues shall be made available to each school Principal/Site Leader for review and comment prior to submission to the Board for approval.

16.3 Special Education Committee

16.3.1 Special Education Committee Purpose

In an effort to improve communication and collaborative problem-solving, the District and the Association shall establish a Special Education Committee made up of bargaining unit members and Programs for Exceptional Children (PEC) administration.

16.3.2 Composition

There shall be not more than fifteen (15) unit members serving on the committee comprised of 75-90% unit members assigned to PEC and 10-25% assigned to general education positions in the District. By October 1st of each year, the Association will provide the District a list of unit members to serve on the

committee for the year. Unit members may not serve on the committee for more than three consecutive years.

The District participants on the committee will include the Deputy Chief of Programs for Exceptional Children and/or designee(s) which may include other administrators and staff.

16.3.3 Special Education Committee Function

It shall be the function of this committee to identify and discuss any opportunities or challenges of common interest related to PEC. The specific function of the Committee shall be to facilitate a culture of collaborative problem-solving.

16.3.4 Logistics

The committee will hold regularly scheduled meetings, no less than every other month during the school year. Meeting times will be based on mutual agreement. The unit members shall elect a chairperson to facilitate communication regarding committee matters, including joint preparation of meeting agenda with the Deputy Chief. Minutes of items discussed and decisions made by the committee shall be reviewed for accuracy by the chairperson and by the Deputy Chief or designee(s). The Deputy Chief shall ensure that the official minutes are posted for review by all District staff. Each party shall be responsible for notifying their respective constituents of the availability of the minutes.

ARTICLE 17 - SAFETY AND SECURITY CONDITIONS

17.1 General

The District and the Association are jointly committed to provide for the safety and security of all staff. Bargaining unit members shall not be required to work under unsafe or hazardous conditions or environments, or to perform tasks that endanger their health, safety, or well-being.

- 17.1.1 The parties support Oakland's status as a sanctuary district. Board Resolution 1617-0089 shall be incorporated herein as Appendix 15.
- 17.1.2 The parties agree that when it comes to student discipline, restorative practices are preferable to punitive ones and that interaction with the criminal justice system is to be avoided whenever possible.
- 17.1.3 Effective July 1, 2019, unit members shall participate in restorative justice and positive behavior intervention system training on an annual basis. Such trainings shall take place on a) one of the three (3) staff development day included in the work year for 186 day unit members (formally known as buy-back days); or b) during professional development time occurring on Wednesdays or other minimum days. Further, should additional one-time or on-going revenues from public or private sources become available for this purpose such funds may be used to restore the cuts in Restorative Justice Programs in Board Resolution 18-2727.

17.2 Fire and Disaster

Neither the Employer nor the unit members shall knowingly violate the provisions of the California Fire Code and/or the Oakland Fire Code. Fire and other emergency action plans have been developed at each work site and unit members should familiarize themselves with details of such plans, including building evacuation routes, the locations to which students are to report in the event of an air raid or other civil disaster, and the administrator in charge in the event of an emergency.

17.3 Disruptive Actions by Students

Unit members may send to the appropriate administrator those students whose actions are disruptive to their classroom instructional program. Should the student refuse to comply, the administrator shall be so notified and appropriate action shall be taken to remove the student from the immediate environment. In response to student behavior under this section, unit members retain the right to exercise a two-day student suspension under Education Code section 48910. Prior to the student being returned to the unit member's class, the administrator shall communicate with (provide feedback to) the unit member to discuss the student's conduct.

17.4 Assault, Attack or Threat Against a Unit Member by Students

17.4.1 The unit member shall report the incident immediately to the administrator in charge or designee, who shall immediately report the incident to the police and notify the

- student's parent, guardian or other responsible adult immediately. In addition, the unit member retains the right to notify the police directly.
- 17.4.2 The unit member shall prepare a written statement concerning the incident and present it to the site administrator who will then add their remarks, if any, to the statement, and then forward it immediately to the Oakland Police Department.
- 17.4.3 A unit member who is the victim of an assault, attack or threat, verbal or physical, and who so requests, shall be immediately granted the remainder of the day of the assault, attack or threat released from duty. These release days shall not be charged against any of the unit member's accrued or accumulated leaves specified in Article 11 Leaves.
- 17.4.4 Students involved in a physical assault, attack or verbal threat, against a unit member, shall not be returned to the unit member's class(es) until a meeting between the unit member and an administrator is held, and the consequences to the student(s) for the misconduct has been determined. When a student is referred to the District' Disciplinary Hearing Process (DHP) the student shall not be returned to the classes of the bargaining unit member from which the student was removed until the DHP is completed.
- 17.4.5 If it is finally determined that a student is the perpetrator of an assault, attack or threat, verbal or physical, against a unit member, that student shall not be returned to the unit member's class(es) or caseload for the remainder of the instructional year.
- 17.4.6 The Employer shall comply with and implement all provisions of Education Code section 49079 regarding prior student conduct to unit members.

17.5 Civil Disorder

Should conditions of civil disorder or student unrest develop to the extent that, in the Superintendent's judgment and in consultation with the FC's where practicable, a school or schools cannot be kept open with reasonable assurance of safety to pupils and unit members, the Employer shall declare an emergency closing of the school for a period of time considered necessary by the Employer. Teaching days lost at the site because of such closure shall be counted as if they were days served within the unit member's contracts, with no loss of salary. After school has been closed, a faculty meeting shall be called by the site administrator to plan for the orderly reopening of the school.

17.6 Unauthorized Person(s)

- 17.6.1 When unauthorized person(s) are observed on school premises, they shall be reported to the administrator or designee, who shall enforce the "no loitering" rules of the Employer by investigating the report; and, if appropriate, ask that the person leave and, if necessary, call the Police Department.
- 17.6.2 The parties acknowledge the importance of campus security. The Employer will seek to implement measures to provide a secure campus learning environment at each school site. Recommendations from the FC and the Employer Safety Committee shall be sought. (See 7.2.)

17.7 After School

The Employer has established procedures to provide for the safety of students and staff members participating in, attending, or supervising after school events.

Procedures for pre-planning and safety determination of after-school events have been established for local sites as well as District-wide events. At the school site, the administrator and FC (see 7.2) meet for the purpose of pre-planning and review of the school's after school athletic or other events for which there may be a question of student or staff safety. Should the Principal/Site Leader and the FC not be in agreement on a safety-related topic, the FC may appeal the administrator's decision to the Employer Events Safety Council.

- 17.7.1 Difference of opinion on matters under consideration by the Council may be appealed to the Superintendent and if deemed appropriate to the Board of Education, whose decision shall be, considered final.
- 17.7.2 Should the unit member believe an assigned duty to be unsafe, the unit member shall discuss the matter with the site administrator, who will endeavor to develop a satisfactory solution to the problem. Should the administrator's solution to the problem be considered as unsatisfactory, the unit member may appeal the problem, in writing, to the appropriate Associate Superintendent for further consideration.

17.8 Procedures Regarding Unsafe or Harmful Conditions

Unit members shall report in writing unsafe or harmful or potentially unsafe or harmful working conditions to the immediate administrator who shall investigate and take action to remedy the unsafe or harmful condition. If, within five days, the site administrator cannot remedy the conditions through the Buildings and Grounds services, they shall notify the appropriate Assistant Superintendent and the unit member by the end of the fifth day.

Within five (5) days, the appropriate Assistant Superintendent shall initiate action to remedy the condition and monitor the progress to ensure that the condition is corrected within a reasonable period of time. The determination of a reasonable period of time shall be made by the appropriate Assistant Superintendent, but shall not be longer than thirty (30) days.

- 17.8.1 If the condition has not been corrected, the appropriate Assistant Superintendent may take whatever steps are necessary to ensure the safety of the affected students and staff, including closing that portion of the school site.
- 17.8.2 Employer shall provide, publish, and post rules for safety and prevention of accidents, and spread of communicable diseases, and provide protection devices where they are required for safety of unit members.
- 17.8.3 Unit members shall work with the site administrator to maintain safe and sanitary conditions in their teaching or work areas of responsibility. Unit members shall have the opportunity to make suggestions and recommendations to the site administrator affecting the safety at the site. Unit members, as employees of the Employer, are protected in the event of injury to their person during the performance of assigned duties to the extent defined in the Workers' Compensation provisions.
- 17.8.4 Unit members shall seek the assistance of the site administrator in moving heavy equipment.
- 17.8.5 The Employer agrees, to the fullest extent possible within the current maintenance budget, to provide lighting at night functions, which are approved school activities.
- 17.8.6 The Principal/Site Leader, in consultation with the FC (see 7.2), shall develop and implement necessary security provisions for night functions, which are approved school activities.

17.8.7 Every effort will be made to perform painting, roofing, and other repairs on days when schools are not in session.

17.9 School Site Discipline Committee

A school site discipline committee shall be established at each school representing parents, unit members and students, if appropriate, to assist in the development of school site procedures relating to student conduct and school site safety. School site discipline committees shall be advisory to the Principal/Site Leader and shall report in writing to the appropriate Assistant Superintendent or their designee annually.

17.10 Safety Information

The Employer shall annually utilize a portion of one of the initial planning days to review with unit member's key sections of this Agreement, administrative bulletins, and laws, which pertain to safety/discipline.

17.11 Classrooms

The Employer shall provide every classroom and unit member work area on Oakland Unified School District (OUSD) property with:

- 17.11.1 A working telephone, or intercom, or other means of electronic communication with office personnel available to receive calls during the unit member's work day, and
- 17.11.2 Peepholes or windows in classroom doors to provide visibility into the hallways or outdoor area.

17.12 Maintenance

Regularly scheduled classroom cleaning, trash removal and maintenance of school grounds at each site shall be maintained.

17.13 Employer Safety Committee

In order to confront increasingly significant district wide problems of safety at school sites: The parties agree to the formation of a cooperative Safety Committee made up of twelve (12) members, four to be appointed by the Association who will represent all Unit members including ECE, Pre-school/elementary, middle school and secondary/adult school sites unit members, four to be appointed by the Superintendent representing Legal and/or Labor Relations, Budget and school-site administration and four to be appointed by the Local Control Accountability Plan (LCAP) Parent-Student Advisory Committee (PSAC). The Committee will gather concerns and suggestions relating to issues of student and school staff safety including but not limited to environmental hazards, natural disaster preparation, fire safety, threats of violence, and issues impacting school climate, including the efficacy of restorative practice. The Committee will arrive at recommendations for action through consensus. Committee recommendations for action will be directly presented to the Superintendent. The Committee will issue monthly reports back to the site Faculty Councils and site administrators, with copies to the Association. In addition, the Committee shall issue an annual report to the Governing Board, the OEA and the LCAP PSAC no later than April 30 of each school year regarding the concerns received and their resolution.

17.14 Vehicle

17.14.1 In recognition of the special circumstances facing unit member in Oakland schools, the District will reimburse unit members when their personal vehicles have been

vandalized on or in the immediate vicinity of school site property while the unit members is engaged in professional duties, subject to the following conditions:

- 17.14.2 Reimbursement will be determined on a case by case basis, as indicated below:
 - 17.14.2.1 Reimbursement will not exceed the amount of insurance deductible or three hundred dollars whichever is less.
 - 17.14.2.2 Reimbursement will be based upon proof of damage and cost of repair to the vehicle due to vandalism during duty hours, including police reports, personal affidavits, witness statements and/or photographs.
 - 17.14.2.3 Reimbursement will be based upon proof of loss if submitted to the insurance carrier and proof of the amount of the policy deductible.
 - 17.14.2.4 Reimbursement shall be made within thirty days of submission of claim, or as soon as practicable.
 - 17.14.2.5 Unit members will cooperate with the Employer in connection with any rights of subrogation accruing to the District by virtue of any payment made pursuant to this article.

17.15 Teacher Initiated Suspension Form

The OUSD Teacher Initiated Suspension Form shall be found in Appendix 15 (F6)17.15)" \s "F6" \c 8 of this Agreement.

17.16 Use of Recording Devices

- 17.16.1 The District may employ the use of video and/or audio recording devices in non-classroom areas in order to enhance school safety. The District agrees that such use shall fully comply with the Education Code 51512 regarding the use of recording devices in classrooms.
- 17.16.2 In gymnasiums and other multi-use facilities, the District agrees to use signage or to otherwise notify unit members working in proximity to such locations of the existence of video recording devices and the possibility of activities being recorded. Such notice does not constitute a waiver of rights under Education Code 51512.

ARTICLE 18 - ORGANIZATIONAL SECURITY

18.1 General

- 18.1.1 Upon written authorization from a unit member, the Employer shall deduct the appropriate amount from the salary of the unit member, make the appropriate remittances for annuities, credit union, savings bonds, charitable donations, or any other plans or programs for which such deductions are designated by the unit member and authorized by the Board of Education.
- 18.1.2 Remittances shall be made within ten (10) workings days of the issuance of the pay warrants, unless it is beyond the control of the Employer.

18.2 Membership Dues

- 18.2.1 The Association shall deliver on or before September 1 of each school year, to the Employer a list of names of all unit members who are Association members and who have signed for authorized deductions of the annual unified membership dues.
 - 18.2.1.1 The Employer shall deduct one tenth or one twelfth of such dues from the regular pay warrants of the unit members, depending on the number of said warrants he/she receives annually.
 - 18.2.1.2 The Association shall provide verification to the Employer monthly of members who sign such dues authorizations after the opening of the school year. Their deduction amounts shall be appropriately prorated to complete the annual dues payments by the end of the school year.

18.3 Agency Fee

The Employer shall deduct an annual Agency Fee from the monthly pay warrants of unit members who are not members of the Association.

- 18.3.1 Unit members new to the Employer shall be apprised of the Organizational Security provisions of this Agreement by the Association and the District.
- 18.3.2 The agency fee amount shall be equal to the unified membership dues of the Association.
- 18.3.3 The agency fee can be made in a lump sum cash payment to the Association or can be authorized for payroll deductions.
- 18.3.4 In the event no cash payment or payroll deduction is authorized, the Employer shall immediately begin automatic payroll deductions as provided in Education Code section 45061. The automatic deductions shall begin with the first pay warrant issued.
- 18.3.5 There shall be no charge to the Association for the mandatory agency fee deductions.

18.4 Agency Fee Exception

- 18.4.1 Any unit member who is a member of a religious body whose traditional tenets or teachings include objections to joining or financially supporting employee organizations shall not be required to join or financially support the Association as a condition of employment.
 - 18.4.1.1 Said unit members shall pay, in lieu of the agency fee, sums equal to the agency fee to one of the following non-religious, non-labor organizations, under Title 26 of the Internal Revenue Code:

- Marcus A. Foster Educational Institute
- Foundation to Assist California Teachers (FACT)
- The CTA Martin Luther King Scholarship Fund
- 18.4.1.2 Arrangements for such payment shall be made on or before October 31 of each school year, except for new hires, who shall make payment within 60 days of date of hire.
- 18.4.1.3 Proof of payment and a written statement of objection, along with verifiable evidence of membership in said religious body shall be made on an annual basis to both the Employer and the Association as a condition of continued exemption from the provisions of Sections 18.2 and 18.3.
 - 18.4.1.3.1 Proof of payment shall be in the form of receipts, payroll deduction stubs and/or cancelled checks indicating the amount paid, date of payment and to whom payment in lieu of the agency fee has been made.
 - 18.4.1.3.2 If such proof is not provided as prescribed in Section 18.4.1.3.1 above, monthly payroll deduction of the agency fee shall begin with the November pay warrants.
 - 18.4.1.3.3 Such deduction shall be prorated to complete full fee payment by the end of the school year.
- 18.4.2 Any unit member making payments as set forth in Section 18.4.1 above, and who requests that the grievance or arbitration provisions of this contract be used in his or her behalf, shall be responsible for paying the reasonable cost of using said grievance or arbitration procedures.

18.5 Reporting and Payments

- 18.5.1 The Employer agrees to promptly remit on a bi-weekly and/or monthly basis dues/agency fee monies to the Association accompanied by alphabetical lists of unit members for whom such deductions have been made.
 - 18.5.1.1 The Employer agrees to categorize the list as to members and non-members.
 - 18.5.1.2 The Employer shall indicate any changes in personnel (additions or deletions) from the lists previously furnished.
- 18.5.2 The Association agrees to furnish in a timely manner any information needed by the Employer to fulfill the provisions of this article.
- 18.5.3 The Oakland Education Association agrees to pay to the Employer all reasonable legal fees and legal costs (including damages) incurred in defending against any court or administrative action challenging the legality of the Agency Fee provisions of this Agreement or their implementation.
 - The Association shall have the exclusive right to decide and determine whether any such action referred to above shall or shall not be compromised, resisted, defended, tried or appealed.

ARTICLE 19 - SAVINGS PROVISION

19.1 General

- 19.1.1 Should any provisions of this Agreement be held contrary to law by a court of competent jurisdiction, such provisions shall not be deemed valid and subsisting except to the extent permitted by law, but all other provisions shall continue in full force and effect.
- 19.1.2 Within thirty (30) days of receipt of a court's decision, negotiations shall commence regarding matters related to such provisions defined in this article. The parties may mutually agree to an extension of this time period.

ARTICLE 20 - CONCERTED ACTIVITIES

20.1 General

- 20.1.1 The Association agrees that it will not authorize, engage in or support any sanction, strike, work stoppage, or other concerted refusal to perform assigned duties by any members of the unit for any reason during the term of this Agreement.
- 20.1.2 The Association and the Employer recognize the duty and obligation of this contract and shall make every effort toward inducing all unit members and their supervisors to do likewise.
- 20.1.3 OEA agrees that the No Strike Clause prescribed above be continued without a window if the parties reach a closed three-year (3) Agreement.
- 20.1.4 Alternatively, if the parties do agree to reopen salaries and/or other economic items each year of the three-year (3) term Agreement, then the No Strike provision shall be lifted during such period after the exhaustion of all impasse procedures, if any, in each year.

ARTICLE 21 - SPECIAL SERVICES AND SPECIALIZED ASSIGNMENTS

21.1 General Provisions

The parties recognize the importance of providing adequate facilities for unit members providing special services at each work site. Toward that end the following will be implemented.

- 21.1.1 A complete survey of such facilities/supplies will be under-taken at each school site. FCs (Section 7.2) will be vital resources in accomplishing a complete survey. The survey will be completed no later than January 1, of each year.
- 21.1.2 The needed repairs/facilities will be listed by priority and presented to the Superintendent for recommendation for Board funding by February 1, of each year. OEA will be consulted prior to submission of the priority list to the Superintendent.
- 21.1.3 The listed items will be addressed to the extent permitted by funds available for this purpose according to listed priority.
- 21.1.4 In conjunction with the foregoing, Association and Employer will simultaneously and jointly work to publicize and bring to the voters of Oakland a bond initiative for the purpose of providing funding for comprehensive repair of existing school site buildings and grounds.
- 21.1.5 The evaluation of teachers providing special services shall be developed as part of Article 13 Performance Evaluation.
- 21.1.6 Nurses, psychologists, counselors, teachers of the visually impaired, Language, Speech and Hearing Specialist (LSHS), Resource Specialists (RS) and Music Instructors shall have a permanent room at each site which includes a locking door, and a locking file, wherever possible. These spaces may be shared with other certificated unit members.

21.2 Counselors

21.2.1 The following provisions shall apply to the allocation of counselors to every site serving sixth (6th) grade and above. "Counselors" as used in this article refers to unit members appropriately credentialed to provide social and academic interventions and counseling support for students. The site allocation for counselors shall be based upon a District-wide ratio of 550:1 (effective July 1, 2019) and 500:1 (effective July 1, 2020).

21.2.2 Student/Counselor Ratio

- 21.2.2.1 The student/counselor ratio shall be based upon the number of secondary students (Grades 6-12) enrolled as of the last Friday in October.
- 21.2.2.2 No more than 64 additional students shall be assigned to the total counseling staff at a school before additional counseling periods shall be provided, except by mutual agreement between site administrator and the counselor(s).
- 21.2.2.3 The District may alter the ratio provided in section 21.2.2.1 above only as a result of actual implementation of a statutory reduction in force (layoff) prompted by budgetary needs, subject to the following:
 - 21.2.2.3.1 The District will notify OEA in advance of any potential decision to reduce counseling services. The parties shall meet to discuss

alternatives that might mitigate or eliminate the need for such reduction in services.

- 21.2.2.3.2 Alteration of the ratio set forth in section 21.2.2.1 above due to a reduction in force shall not result in a District-wide ratio greater than 600:1 (effective July 1, 2019) and 550:1 (effective July 1, 2020).
- 21.2.3 In the senior high schools, data processing, testing and scholarship services shall be provided. In the middle schools, data processing and testing services shall be provided.
- 21.2.4 When possible, a credentialed substitute counselor will be provided for a counselor whose absence exceeds five consecutive school days; if such absence has been determined to run for said length in advance.
- 21.2.5 The Employer may utilize the services of retired counselors to assist regular counselors during periods of time when it is determined that there is a heavy demand for counselors' services (i.e. opening and closing of the school year).
 - 21.2.5.1 All retired credentialed counselors that are so utilized shall be paid the current substitute rate.
- 21.2.6 The Employer agrees that all counselors, who possess valid credentials and are affected by the new provisions which require a student-counselor ratio of 600:1 (effective July 1, 2019) and 550:1 (effective July 1, 2020), will be offered positions in the classroom by the Employer.
- 21.2.7 Every site serving sixth (6th) grade and above shall be ensured monthly counseling services.
- 21.2.8 Individual counselors shall not be required to provide services to more than three (3) school sites as part of their regular assignment.
- 21.2.9 Every effort shall be made to make counselor assignments in a manner intended to achieve reasonably balanced workloads.

21.3 Coaches

The parties agree that coaching assignments are considered unique extra-duty responsibilities. Any coach whose assignment is to be changed or discontinued during the next school year shall be notified by the Principal/Site Leader in writing by May 30. Such notification shall include verifiable reasons for the change or removal. Decisions to change or remove shall be made on an objective good faith basis after proper notice and warning.

21.4 Bilingual Education

21.4.1 BCLAD Stipend

- Beginning with the 2015-2016 school year, a BCLAD stipend of \$1500 shall be paid to all unit members who hold BCLAD certification. The stipend shall be added to the annual salary amount and paid out in ten equal increments.
- 21.4.2 The District shall provide bilingual secondary and elementary Resource Specialists, when possible.
- 21.4.3 District Bilingual Advisory Committee
 - 21.4.3.1 The OEA shall appoint five (5) unit members to the District Bilingual Advisory Committee.

21.5 Home and Hospital Instruction

Home and Hospital Teachers shall have the choice of teaching additional hours above their contract for one year at the substitute rate. For the following year at the teacher's option, their contract shall be increased by the same number of hours (percentages) as he/she taught above their contract in the previous year.

- 21.5.1 Full-time home and hospital teachers shall receive one preparation/conference period per week. Such preparation time shall be pro rata for less than full-time home and hospital teachers.
- 21.5.2 The Employer shall provide adequate travel time between assignments as part of the regular workday.
- 21.5.3 A home and hospital assignment shall not require a teacher to work beyond their requested hours.

21.6 Instrumental Music Program

There shall be a secure, designated place for storing musical instruments at each school site wherever possible.

- 21.6.1 Class size shall be maintained per current contract.
- 21.6.2 In middle schools, music teachers who develop a program subject to the approval of the Principal/Site Leader, or appropriate administrator shall be compensated at the extra-duty hourly rate up to a maximum of 15 hours per site when such duty exceeds the designated five (5) hours per month. Programs related to this duty must be in accordance with site needs and consistent with the basic academic program.
 - 21.6.2.1 High school teachers will be paid as stipulated in Article 24 Compensation.

21.7 Consumer and Homemaking Education

Mileage incurred by consumer education and homemaking teachers while shopping for supplies to meet program requirements shall be subject to the contract provisions of Article 24.

The mileage must be incurred within a reasonable radius of a teacher's school site.

21.8 Librarians

A maximum number of pupils assigned to a library period will be established at each school site by the school Principal/Site Leader with input from the librarian. Except in emergencies, librarians shall not be assigned duties (e.g.: substituting) that are not in their general job description.

21.9 Nurses

- 21.9.1 The maximum caseload of a nurse shall not exceed 1:1350.
- 21.9.2 Time spent by nurses complying with State mandated requirements beyond the regular work day shall be considered administratively assigned and shall be subject to the Extra Pay Provisions of Section 10.2 above.
- 21.9.3 Subject to the approval of the Principal/Site Leader, the nurse will develop, at the beginning of the school year, a work schedule inclusive of record-keeping time based on school/student needs. During this scheduled record-keeping time, students will be seen only in cases of emergency.

21.9.4 The District and OEA shall establish monthly joint meetings of a Nurses' Faculty Council/Joint Committee, to facilitate communication between representatives of the Nurses and District administrative personnel. The Nurses' Faculty Council shall conduct a joint study to inform best practices for recruitment and retention of school nurses and management of district-wide caseload overages.

21.10 Psychologists

When psychologist positions become vacant, psychologists presently employed by Oakland Public Schools for less than full-time shall be given first consideration in filling these positions.

- 21.10.1 Effective July 1, 2019, the maximum caseload for each psychologist shall not exceed 725:1. Alteration of the ratio set forth in this section due to a reduction in force shall not result in a District-wide ratio greater than 750:1.
- 21.10.2 Effective July 1, 2020, the maximum caseload for each psychologist shall not exceed 700:1. Alteration of the ratio set forth in this section due to a reduction in force shall not result in a District-wide ratio greater than 725:1.
- 21.10.3 The District may utilize Psychologist interns, if available, for up to 12 months. If Psychologist interns are utilized, then any supervising psychologist will be provided a minimum of two (2) hours of time per week for the supervision of their intern.
- 21.10.4 The District will make every effort to utilize temporary or substitute School Psychologists to cover sites where there is a vacancy due to extended leave or mid-year resignation.

21.11 Speech and Language Pathologist Specialists (SLP)

Each SLP shall be provided two (2) release days to complete mandated paperwork. The dates shall be scheduled by mutual agreement between the SLP and immediate supervisor.

When justified by student need, SLP shall have the option to schedule all assigned students for two sessions per week.

SLP shall have a secure, designated place for storing their instructional materials at each school site, whenever possible.

- 21.11.1 SLP who travel shall be provided adequate time for travel between assignments.
- 21.11.2 Each full-time (1.0 FTE) SLP shall be reimbursed annually by the District for fees paid for obtaining and retaining the American Speech Language and Hearing Association (AHSA) Certificate of Clinical Competence. The reimbursement amount shall be prorated appropriately for less than full-time SLPs. Requests for reimbursements must be submitted to the District by January 31.
- 21.11.3 Utilization of SLP shall be consistent with state law.
- 21.11.4 SLP Joint Committee provisions of Section 16.1 may be utilized to address programmatic needs and concerns.
- 21.11.5 The Employer will review on a quarterly basis Speech and Language caseloads and make adjustments to ensure equitable workloads: taking into account age of and number of students, number of assigned schools, and identified needs on student IEP's.

21.11.6 Effective with the 2019-20 school year, the SLP caseload for an individual SLP shall not exceed the average of 55:1. Effective with the 2020-21 school year, the caseload for an individual SLP shall not exceed the SELPA-wide average of 50:1.

21.12 Teachers of the Visually Impaired and Orientation and Mobility Specialists

Teachers of the Visually Impaired and Orientation and Mobility Specialists shall have a secure, designated space for storing their instructional materials at each site.

- 21.12.1 Teachers of the Visually Impaired and Orientation and Mobility Specialist who travel shall be provided adequate time for travel between assignments.
- 21.12.2 Extra-duty responsibilities for teachers of the Visually Impaired and Orientation and Mobility Specialists shall be the same as those provided in Section 10.2, Workday, and Section 10.4 Faculty Meetings.

21.13 Special Day Class Teachers (SDC)

Through the Individualized Education Plan (IEP) process the team may recommend SDC student placements to the coordinator. The coordinator or designee shall consider such recommendations in determining the placement of students.

- 21.13.1 Class Composition: Non-severely handicapped (NSH) students shall be placed through the IEP process in classes with peers of comparable abilities and educational objectives whenever possible.
- 21.13.2 SDC teachers shall have, on an equal basis with teachers at the site, access to school site equipment and instructional material designated for general use by regular classroom teachers whenever possible.
- 21.13.3 With the prior approval of the Principal/Site Leader, the SDC teacher may visit other SDC sites to conduct relevant business during non-instructional time.
- 21.13.4 Special Education teachers shall not be required to perform medical procedures for students or assume responsibility for the performance of such procedures except for the routine administering of medication. However, voluntary training for Special Education teachers shall be provided, at Employer expense, if specialized physical/health services are required.
- 21.13.5 Extra-duty responsibilities for SDC teachers shall be the same as those established for all other regular classroom teachers.
- 21.13.6 The SDC teacher may work with their supervisor to prioritize the classroom services of the paraprofessional(s) consistent with the job description for the paraprofessional(s).
- 21.13.7 SDC teachers shall be invited to participate in the interview process to fill paraprofessional vacancies in their programs. It is understood that this language does not apply if a permanent SDC teacher is not currently assigned to this position.
- 21.13.8 Each SDC teacher shall submit to the Site Leader recommended work schedule and work responsibilities for any paraprofessional assigned to their program.
 - 21.13.8.1 Under the direction of and support from the Site Leader and/or appropriate program personnel, SDC teachers are responsible for providing program guidance, including orientation to paraprofessional(s) assigned to their program.
- 21.13.9 A paraprofessional position shall be allocated to each established SDC. The preceding sentence shall not limit the Employer's ability to layoff or otherwise

- manage paraprofessionals. SDC teachers may submit a written request if they do not wish any paraprofessional positions allocated to their class. The Employer may honor such requests if it does not jeopardize the ability to implement student IEPs.
- 21.13.10 If the teacher of a NSH SDC Class believes that the majority of their students are severely handicapped, he/she may request a meeting with the appropriate Special Education coordinator to review the appropriateness of the class designation as NSH.
- 21.13.11 The Employer will review SDC student rosters on a quarterly basis and redirect new student placements to ensure equitable work loads: taking into account age and number of students, disabilities of students, identified needs on student IEP's, size of classrooms, parent preference and specific class program needs.

21.14 Resource Specialists (RS)

The RS shall have access to the copy machine(s) at each site, where available.

- 21.14.1 Extra-duty responsibilities for RS shall be the same as those provided in Section 10.2, Workday, and Section 10.4 Faculty Meetings.
- 21.14.2 Secondary and Elementary RS
 - 21.14.2.1 RS who travel shall be provided adequate time for travel between assignments.
 - 21.14.2.2 RS shall have a secure, designated place for storing their instructional materials at each school site, wherever possible.
 - 21.14.2.3 If mitigating circumstances prevent an RS from performing their responsibilities, the coordinator, Principal/Site Leader, and RS shall meet to reassess those responsibilities.
 - 21.14.2.4 Elementary RS who travel shall not have to participate in the site yard duty schedule except in emergencies.
 - 21.14.2.5 Any student who does not have a valid IEP shall not be served by the RS Program.
 - 21.14.2.6 RS shall not be required to supervise Resource Specialist Program students during the student's unscheduled instructional time in the RS program.
 - 21.14.2.7 RS who are assigned to two (2) schools shall attend one faculty meeting per month, on an alternating basis.
 - 21.14.2.8 Utilization of RS shall be consistent with State law.
 - 21.14.2.9 RS will be invited to participate in the interview process to fill IA vacancies in their programs. It is understood that this language does not apply if a permanent RS is not currently assigned to this position.
 - 21.14.2.9.1 Each RS shall submit to the Site Administrator recommended work schedule and work responsibilities for any IA assigned to their program.
 - 21.14.2.10 RS are responsible for providing programmatic guidance including orientation to IA assigned to their program.

21.14.3 Caseload Limit

21.14.3.1 No individual RS shall be assigned more students than the state caseload maximum 28:1. Effective with the 2020-21 school year, the caseload for RS shall not exceed 27:1.

21.14.4 Itinerant Site Assignments

The RS site schedule shall be arranged by the coordinator in collaboration with RS and affected site administrators to minimize instructional and program time loss due to travel time between sites.

21.14.5 Initial Assessments

- 21.14.5.1 Beginning with the 2015-2016 school year, in order to make RS assignments in a manner intended to achieve a reasonable balance among RS's, the number of initial assessments performed by each RS should not exceed 12 annually.
- 21.14.5.2 When an individual RS approaches 12 assigned initial assessments, the following steps will occur:
 - 21.14.5.2.1 The District shall exhaust all available options to prevent the assignment of additional initial assessments that would cause the number in 21.14.5.1 to be exceeded.
 - 21.14.5.2.2 The Special Education Committee (see Article 16.3) shall monitor initial assessments of all RS's on at least a monthly basis to achieve the number in 21.14.5.1.

21.15 Full Inclusion Education

21.15.1 Definition

21.15.1.1 The full-time enrollment as prescribed by IEP, of a SDC student in a general education classroom, which is not staffed by a full-time Special Education teacher.

21.15.2 General Provisions

- 21.15.2.1 The full inclusion student shall count as part of the regular education teacher's class size roll.
- 21.15.2.2 The general education teacher shall receive a copy of the student's IEP.
- 21.15.2.3 Any full inclusion student with an IEP shall be included in the Special Education case carrier's caseload.
- 21.15.2.4 The Special Education case carrier in consultation with the general education teacher(s) shall submit to the Site Administrator and appropriate coordinator a recommended work schedule and work responsibilities for any IA/AHC assigned to their programs.
- 21.15.2.5 The Special Education case carrier in consultation with the general education teacher(s) is responsible for providing program guidance, including orientation to IA/AHC assigned to their programs.
- 21.15.2.6 It is the intention of the Employer to continue the practice of planning full inclusion placements in such a way that no more than two full inclusion students are normally placed in any one self contained general education classroom.

- 21.15.2.7 It is the intention of the Employer to assign full inclusion students to case carriers who have the necessary competencies to meet the needs identified in the IEP.
- 21.15.2.8 The Principal/Site Leader, upon receiving a copy of the IEP, shall assign a full inclusion student to a general education classroom. Among other factors the assignment shall take into consideration physical space needs of the student, scheduling, curriculum designs, existing classroom composition and the unique needs of the student.
- 21.15.2.9 The Special Education case carrier shall make available to the Principal/Site Leader any additional pertinent information regarding the student, which may have a bearing on placement. Any general education teacher who wishes to volunteer to teach a full inclusion student may have such a request considered as one of the placement factors.
- 21.15.2.10 This is for information only. A unit member may request an IEP review to be held pursuant to California Special Education Code Section 56343.
- 21.15.2.11 Under the direction of the Site Administrator and consultation with the appropriate Special Education Coordinator, the Special Education case carrier and the general education teacher(s) will develop a plan for the support. Each general education teacher serving a full inclusion student will receive one (1) day per month of release time via a rotating substitute.
- 21.15.2.12 Preference shall be given to the use of single grade classrooms as one (1) of the primary factors in selecting placements for full inclusion students.

21.16 Physical Education (P.E.) Teachers

Every effort will be made to ensure that no more than 10 percent of all students enrolled in any physical education class are identified emotionally or physically handicapped.

21.17 Journalism Teachers

Journalism unit members who are responsible for the yearbook and/or the school newspaper shall have a daily period for production.

In performing these responsibilities, such unit members shall not be required to work beyond their regular duty day, except as provided in Section 10.2.

21.18 Substitute Teachers

21.18.1 Substitutes are an integral part of the school community and deserve to be treated as such.

The Employer shall send confirmation forms to substitute teachers by August 1 of each year in order to determine the composition of the substitute list for the ensuing school year. Substitute teachers who do not return their confirmations by September 15 will be removed from the list.

- 21.18.2 The daily and Long-Term rates of all substitute teachers are found in Article 24 Compensation.
- 21.18.3 Substitute teachers may purchase benefits from the Employer's Kaiser Health Plan at their own expense. Substitutes shall also be permitted to purchase benefits from

other Employer health and welfare plans if there is no cost to the Employer for processing or administering their participation.

21.18.4 Substitute Assignments

- 21.18.4.1 The unit member shall request a substitute.
- 21.18.4.2 If the unit member is not able to call for a substitute, the Principal/Site Leader or their designee shall call for a substitute.
- 21.18.4.3 If a specific substitute is requested by the teacher or the Principal/Site Leader, the request shall be honored if the substitute is available.
- 21.18.4.4 If there is no request for a specific substitute, assignments will be made in the following order:

21.18.4.4.1	<u>Elementary</u>	Secondary
21.18.4.4.2	a) Elementary credential	a) Fully credentialed in subject area
21.18.4.4.3	b) Emergency credential	b) Emergency credential

- 21.18.4.5 Given the time constraints and the process used for obtaining substitutes, every effort will be made to call the most senior substitutes in each category first. Seniority for substitute teachers shall be based upon the number of days worked as a substitute in the District.
- 21.18.4.6 All substitutes shall be expected to work for a full day unless there is a prior agreement between the Employer and the substitute to work a half-day.
 - 21.18.4.6.1 In no case shall salary be less than a half-day increment, except for Adult Education teachers who work by the hour.
- 21.18.4.7 Substitutes eligible for sick leave pursuant to Article 11.3.3.1 shall have access to sick leave forms made available online for utilization by substitutes with the appropriate instructions.

21.18.5 Duties of a Daily Substitute

- 21.18.5.1 The duties of a daily substitute shall be the same as those of the regular classroom teacher.
- 21.18.5.2 A daily substitute shall have no duties outside of the regular duty day but shall be entitled to any regular break time, or conference time including a duty-free lunch that the regular teacher's schedule calls for. At the elementary level scheduled prep time shall be used for correction of student work when specific instructions and answer keys are left by the teacher.
- 21.18.5.3 A detailed list of the daily substitute's duties shall be included in the Employer's substitute packet that will be given to each substitute at the beginning of the school year.

21.18.6 Duties of a Long-Term Substitute

21.18.6.1 The duties of a long-term substitute shall be the same as the duties of the regular teacher as defined in Article 10 - Hours of Work of the contract except:

21.18.6.2 A long-term substitute shall not be required to submit written lesson plans for a period to exceed twice the number of days of the assignment or for a period longer than any regular unit member is expected to plan, or for a period longer than an expected termination date for the assignment.

21.18.7 Substitute Evaluation

- 21.18.7.1 The Principal/Site Leader or their designees will observe/evaluate substitutes after 18 consecutive days on an assignment, using the Employer's long-term substitute observation/ evaluation form. The evaluator and substitute will meet to discuss the report. Completed forms will be given to the substitute; sent to the Human Resources Services and Support; and, kept at the school site for three (3) years.
- 21.18.7.2 After eighteen full days at a site (not necessarily continuous days) the substitute may request the site administrator to write a letter of assessment/recommendation which shall become a permanent part of the substitute's personnel file.

21.18.8 General Provisions

- 21.18.8.1 At least one mailbox shall be provided at each site for substitute teachers.
- 21.18.8.2 Substitute teachers shall receive twice-monthly paychecks. Each substitute shall receive their check by mail to the last address of record.
- 21.18.8.3At the beginning of each school year, each new substitute teacher shall receive a handbook outlining the general operating procedures of the schools and such other information as is necessary to assist substitute teachers in the discharge of their responsibilities. This information shall include accurate maps covering the geographical area served by the Employer.
- 21.18.8.4 In-service programs for substitute teachers shall be developed by the Employer with input from the Association. At the beginning of each semester, all new substitute teachers will be assigned to attend at least one in-service training session.
- A substitute teacher will be compensated at the Long-Term Rate for any full days of service after the 30th day of service. To be eligible for compensation under this subsection, a substitute teacher must have written verification of completion of the paid minimum half-day in-service program or other professional development as offered by the Employer pursuant to Section 21.18.9.4. This subsection applies only to substitute teaching assignments in K-12 and does not apply to unit members participating in the STIP program identified in Section 21.18.11 Substitute teachers that work 60 days will remain at the long term rate the following school year and will be reviewed annually for eligibility.
- 21.18.8.6 Recognizing the importance of a pool of qualified substitute teachers at the present, as well as in the future, the Employer agrees whenever possible to use this pool from which to hire new teachers.

21.18.9 Requests for a Substitute Not to Return to a Particular School-Site

21.18.9.1 The site administrator and the substitute each have the right to request that a substitute not return to a site for the duration of a school year. Any site administrator or substitute may make such a written request and

submit it to the Substitute Office along with supporting rationale. The Substitute Office shall honor these requests. All records and written requests from this type of action shall be destroyed at the end of a three-year period.

- 21.18.9.2 Requests for a Substitute Not to Return to a Particular Adult Education School Site
 - 21.18.9.2.1 An Adult Education site administrator and an Adult Education substitute each have the right to request that an Adult Education substitute not return to a site for the duration of a school year. Any Adult Education site administrator or Adult Education substitute may make such a written request and submit it to the Director of Adult Education along with supporting rationale. The Director of Adult Education shall honor these requests. All records and written requests from this type of action shall be destroyed at the end of a three year period.

21.18.10 Substitute Teacher Incentive Plan

- 21.18.10.1 A participant in the Substitute Teacher Incentive Plan (STIP) is defined as a substitute teacher who serves every day for the entire period of the STIP contract and is required to report daily to an assignment at any grade level or subject area.
- 21.18.10.2 Qualifications for Participation
 - 21.18.10.2.1 Must have served for one or more years as a substitute with a record of "meets District standards" on evaluations from site administrators.
 - 21.18.10.2.2 Must possess or be eligible for a California regular teaching credential authorizing service in Grades K-12 or an emergency 30- day teaching permit.
 - 21.18.10.2.3 Failure to keep the credential current will invalidate the STIP contract. The Employer will process credential information submissions in a timely manner.
 - 21.18.10.2.4 STIP Substitutes will be assigned and on duty the first instructional day of the school year.
- 21.18.10.3 The District will:
 - 21.18.10.3.1 Recruit a minimum of forty-five (45) STIP Substitutes beginning April 1, for the subsequent school year. STIP applications will be available through the Human Resources Services and Support and the OEA.
 - 21.18.10.3.2 Guarantee work for every instructional day for 180 days. This will include eight (8) days of in-service. STIP Substitutes hired after the opening of the school year will have their contract days adjusted accordingly.
 - 21.18.10.3.3 Beginning with the 2015-2016 school year, provide single coverage for Kaiser Health Plan, dental, and vision benefits commensurate with probationary and permanent unit members. Coverage is for the twelve-month period September 1 through August 31, for each STIP participant.

- 21.18.10.3.4 Provide five (5) days of Sick Leave for each participant. Sick Leave shall be defined and used as provided in Article 11 Leaves.
- 21.18.10.3.5 Determine the home base school for each STIP participant.
- 21.18.10.4 The STIP Substitute will:
 - 21.18.10.4.1 Be prepared to work every instructional day as assigned on a day-to-day or long-term basis, and fulfill the duties as prescribed in Sections 21.18.4 to 21.18.9.
 - 21.18.10.4.2 Sign a temporary contract designed specifically for this purpose.
 - 21.18.10.4.3 Participate in district and/or school site in-services as assigned. The Employer and the Association will jointly design in-service opportunities to improve substitute services provided through this program.

21.19 Year-Round Schools

At present the District does not operate or anticipate operating during the term of this Agreement Year-Round schools. In the event the District determines to operate Year-Round schools in the future, it shall meet and confer with OEA prior making that decision and negotiate the effect of that decision.

21.20 Classroom Teacher Instructional Improvement Program

See Appendix 5. (Sec. 21.20)5

21.21 Adult Education Hourly

21.21.1 Beginning with the 2015-2016 school year, Adult Education Hourly members teaching ten (10) hours a week or more shall receive one (1) hour of paid preparation time per week in addition to their instructional hours. This hour shall count towards eligibility for health benefits.

21.22 Newcomer Supports

21.22.1 Beginning with the 2019-20 school year, schools with 50 or more newcomer students shall be allocated 0.5 FTE to support newcomer students. Schools with 100 or more newcomer students shall be allocated 1.0 FTE.

ARTICLE 22 - EARLY CHILDHOOD EDUCATION CENTERS

22.1 General

- 22.1.1 All articles of this contract are in effect unless specifically modified in this article. Toward this end, the parties expressly agree to modify the following contract provisions:
 - Article 3 Definitions
 - Article 10 Hours of Work
 - Article 12 Transfer/Consolidation
 - Article 15 Class Size and Caseloads
 - Article 17 Safety and Security Conditions
 - Article 21 Special Service and Specialized Assignments
 - Article 24 Compensation
- 22.1.2 Article 23 of this contract shall not apply to Early Education Centers.

22.2 Definitions

- 22.2.1 Center is defined as an Early Childhood Education Center.
- 22.2.2 Site Administrator is defined as the management employee who has responsibility for one or more centers or offices.
- 22.2.3 Teacher is defined either as a CDC Preschool teacher, a CDC School-age teacher or a State Preschool teacher. (A State Preschool teacher is also sometimes referred to as a State Pre-K or State Pre-Kindergarten teacher.)
- 22.2.4 Day is defined as a center workday.
- 22.2.3 Daily Rate is defined as the employee's annual salary plus stipends, divided by the number of contract days of service in a center year.
- 22.2.4 Seniority is defined as number of years of service in the District with required certificate/credential, including leaves of absence for sabbatical and military leave.

22.3 Association Rights

- 22.3.1 The Employer shall provide release time, cost of substitute if required, to be borne by the Association, for up to five (5) unit members who have been elected to serve as official Association representatives, to attend ten (10) Association meetings per year. Additionally, when necessary, up to four (4) unit members who have been so elected may, with the approval of the site administrator, arrange to change shifts or to arrive at other mutually agreeable solutions (including providing substitutes when available) in order to attend these meetings.
- 22.3.2 The Association shall provide the Employer with the names of all official Association representatives and the dates and times of Association meetings.
- 22.3.3 The parties recognize that under normal circumstances unit members are not required to perform routine custodial tasks. Nevertheless, the performance of such tasks may be required from time to time in the interests of teacher/pupil safety and hygiene. Unit members are not required to cook food or obtain food for their pupils unless they agree to do so.

22.4 Safety Conditions (ECE)

22.4.1 General

22.4.1.1 The District, with input from the site administrator and the FC at each school site shall work to ensure that employees will not be required to work under unsafe or hazardous conditions or to perform tasks, which endanger their health, safety, or well being.

22.4.2 Fire and Disaster (ECE)

22.4.2.1 Neither the Employer nor the employee shall knowingly violate the provisions of the California Fire Code and/or the Oakland Fire Code. Fire and other emergency action plans have been developed at each school site and teachers should familiarize themselves with details of such plans. These include building evacuations routes, the locations to which students are to report in the event of an air raid or other civil disaster, and the administrator in charge in the event of an emergency.

22.4.3 Disruptive Action by Students (ECE)

22.4.3.1 Unit members may send to the appropriate administrators those students whose actions are a danger to the teacher's person or property and/or are disruptive to their classroom instructional program. Should the student refuse to comply, the administrator shall be notified and appropriate action shall be taken to remove the student from the immediate environment.

22.4.4 Assault, Attack or Threat Against an Employee

- 22.4.4.1 It shall be the duty of the employee to:
 - 22.4.4.1.1 Report the incident immediately to the administrator in charge or designee, who will immediately report the incident to the Police or immediately notify the student's parent, guardian, or other responsible adult.
 - 22.4.4.1.2 Prepare a written statement concerning the incident and present it to the site administrator who will then add their remarks, if any, to the statement, and then forward it immediately to the Oakland Police Department.
- 22.4.4.2 The District shall provide and select legal counsel for an employee who appears as a witness at any trial or hearing when such request is initiated by the District. The District shall also provide legal counsel to an employee charged in a criminal or civil complaint that results from the performance of duties provided that criteria as set forth the law is met. Such law provides "...act or omission in the scope of employment and the public entity determines that it is in the best interest of the public entity. The employee acted or failed to act in good faith without actual malice and in the apparent interest of the public entity."
- 22.4.4.3 In the case of assault, attack or threat against a student, it shall be the duty of the employee to report the incident immediately to the administrator in charge who shall then implement the District's Discipline Policy.

22.4.4.4 The students involved in conduct prohibited by Sections in 22.4.3 and 22.4.4 above shall not be returned to any class until the site administrator has initiated action in accordance with the District's Discipline Policy.

22.4.5 Civil Disorder (ECE)

- 22.4.5.1 Should conditions of civil disorder or student unrest develop to the extent that, in the Superintendent's judgment, and in consultation with the FC's where practicable, a school(s) cannot be kept open with reasonable assurance of safety to pupils and teacher, the District shall declare an emergency closing of the school for a period of time considered necessary by the District.
- 22.4.5.2 Teaching days lost at the site because of such closure shall be counted as if they were days served within the teaching contracts, with no loss of salary. After school has been closed, the site administrator will call a faculty meeting for the orderly reopening of the school.

22.4.6 Unauthorized Person (ECE)

- When unauthorized persons are observed on school premises, they shall be reported to the administrator or designees, who shall enforce the "no loitering" rules of the District by investigating the report; and, if appropriate, ask that the person leave and, if necessary, call the Police Department.
- 22.4.6.2 The parties acknowledge the importance of campus security. The District will seek to implement measures to provide a secure campus learning environment at each school site. Recommendations from the FC and the District Safety Committee shall be sought.

22.4.7 Safety After School (ECE)

22.4.7.1 The District has established procedures to provide for the safety of students and staff members participating in, attending, or supervising after school events. Procedures for preplanning and safety determination of after-school events have established for local sites as well as District-wide.

22.4.8 Safety Conditions

22.4.8.1 The Employer with the input from the Site Administrator and the FC at each Early Childhood Education Center shall work to ensure that unit members will not be required to work under unsafe or hazardous conditions or to perform tasks which endanger their health, safety, or well being.

22.4.9 School Site Discipline Committee

An Early Childhood Education Center Safety/Discipline Committee shall be established at each center site. Committee members will include the site administrator and an equal number of unit members and parents. The center committee shall meet and develop procedures pertaining to safety, student conduct and discipline. The committee shall function in an advisory capacity and shall make an annual report to the Director of Early Childhood Education Centers. A copy of the report shall be made available to the Association upon request. The Site Administrator shall be

responsible for advising unit members of the requirements of the State law and reporting procedures and responsibilities related to child abuse.

22.4.10 Safety-After Center Hours

22.4.10.1 The employer has established procedures to provide for the safety of students and unit members participating in, attending, or supervising after-center hour's events. Procedures for preplanning and safety determination of after-center hour's events have been established for local sites as well as District-wide events. At the center site, the administrator and FC meet for the purpose of preplanning and review of the student or staff safety. Should the site administrator and the FC not be in agreement on a safety-related topic, the FC may appeal the administrator's decision to the District Events Safety Council.

22.5 Vacancies

- 22.5.1 All known vacancies shall be listed as vacancies occur. All qualified permit teachers in the District may apply. Priority consideration will be given in the selection process to teachers in Child Development Centers. Positions which become available as a direct result of the above mentioned posting may be posted immediately. All vacancies, which have not been posted and have been filled by newly appointed teachers, shall be posted at the next available posting period.
- 22.5.2 Vacancy lists shall be available in all center sites and in the Human Services Division as well as in the Association office lobby. All contracted certificated (Permit) unit members shall be eligible to apply for posted vacancies in the centers.
- 22.5.3 All vacant or newly established positions for which early childhood Permit teachers are eligible to apply shall be listed.
- 22.5.4 All certificated (Permit) unit members shall be eligible to apply. Such eligibility will be clearly stated on the official listing. Applicants who are not selected for a position shall be notified in writing within seven center days of the date the selection is made. Applicants may request that the reasons for denial be given in writing. Such reasons shall be provided upon request. (See Article 12 Assignment/Transfer/Vacancy/ Consolidation Policy.)

22.6 Application Procedure

- 22.6.1 Application forms shall be available in the Center Administration Office and in the Human Resources Services and Support.
- 22.6.2 Unused.
- 22.6.3 It is desirable for the unit member to discuss with the center administrator the reasons for submitting a transfer application if the unit member so desires.
- 22.6.4 Each center administrator or the program director shall supply the Director of Human Resources with the basic data upon which to determine eligibility. The Director of Human Resources shall determine the eligibility of a candidate for consideration for a position on the basis of assignment factors.
- 22.6.5 Ineligible candidates will be notified of their ineligibility. If an interview seems desirable, it may be arranged by either the unit member or the center administrator.
 - 22.6.5.1 The Director of Human Resources shall notify all applicants for a position that a decision has been made. If the transfer request is denied, the unit member may request that the reasons for the denial be given in writing.

- 22.6.6 Notices of vacancies for which unit members are eligible to apply shall be listed, except in cases where vacancies are filled by Employer-initiated transfers.
- 22.6.7 Vacancies not filled as the result of the second listing and those resulting from it shall be filled by unit members who do not have an assignment because of consolidation of their positions, unit members on leave who have not been assigned and other unassigned unit members.
- 22.6.8 All ensuing vacancies during the year shall be filled first by unassigned unit members who have the required certificate/credential qualifications/experience (as defined in past preparation, formal evaluations, and length of service to District). Remaining vacancies shall be filled by applicants (new hires) having qualifications for the position.
- 22.6.9 Positions of unit members on leave for one-half their duty year shall not be considered vacancies.
- 22.6.10 These provisions shall not apply to unit members who extended their leave beyond one-half duty year. In the event the unit member does not return or chooses to select another assignment, the unit member's former position shall be declared vacant and will be listed.
- 22.6.11 Unit members leaving stamped self-addressed envelopes with the Director of Human Resources shall receive OUSD certificated listings during their non-duty days.

22.7 Assignment Factors Considered for Position

- Possession of appropriate California certificate/credential.
- Qualifications experience.
- All the above factors being equal, seniority in the District shall be given preference in granting a transfer request.

22.8 Transfer – General Provision

- 22.8.1 A transfer is defined as an assignment change from one site to another within the District, but does not apply to a change in assignment at a site or to assignment to a particular classroom or age grouping.
- 22.8.2 Unit members shall not be transferred or consolidated more than once during their probationary period except in an emergency or for extenuating circumstances.
- 22.8.3 Unit members whose transfer requests have been approved and newly hired unit members will be expected to remain in the new assignment for at least two years unless an emergency situation or extenuating circumstances make it desirable for a transfer to take place.
- 22.8.4 Transfers and changes in assignment shall be made on a voluntary basis whenever possible. In making voluntary transfers, the convenience and wishes of the teacher shall be given careful consideration.

22.9 Employer - Initiated Transfer

- 22.9.1 If the center administrator initiates a transfer, the center administrator shall arrange a conference and discuss with the unit member the reasons a transfer is being considered.
- 22.9.2 If at the conclusion of the conference it is determined that transfer is desirable, the center administrator shall submit a written request to the Director of Early Childhood

Education listing the reason, a copy of which shall be given to the unit member. Except in cases of emergency, unit members shall be given notice of employer-initiated transfers by the first Friday in January for a transfer that is to take effect the beginning of the spring semester, and the first Friday in June for a transfer that is to take effect between July and September. (Emergency is defined as an unforeseeable or unanticipated circumstance requiring immediate action.) The Director of Early Childhood Education shall confer with the Director of Human Resources to discuss all known vacancies and to consider possible reassignments. When a reassignment or a number of reassignments are identified, it shall be the responsibility of the Director of Human Resources to confer with the unit member and arrange reassignments. This does not preclude the unit member from applying for transfer to a listed vacancy.

- 22.9.3 The Superintendent, Director of Human Resources shall arrange a conference with the unit member receiving an Employer-initiated transfer and discuss the necessity for the transfer. The new assignment will also be discussed at this conference. Reasons for the transfer, if requested by the unit member, shall be put in writing and a copy given to the unit member.
- 22.9.4 Employer-initiated transfers shall not be enacted for reasons of a punitive or disciplinary nature. Any reason considered in relation to Employer initiated transfer must have a direct bearing on the unit member and the unit member's specifically assigned responsibilities. In addition, unit members may request an employer-initiated transfer with reasons to be stated to the appropriate site administrator or Director of Early Childhood Education.
- 22.9.5 If possible, the Director of Human Resources shall notify the unit members in writing of the new assignment by the second Friday in January when the Employer-initiated transfers are to take effect at the beginning of the spring semester and by the first Friday in June when it is to take effect the following program year.

22.10 Consolidations

- 22.10.1 Factors to be considered in selecting a unit member are:
 - Certificate/Credential
- 22.10.2 All of the above given factors being equal, seniority in the District shall be given preference.
- 22.10.3 A staff consolidation is defined as a reduction in the number of certificated unit members at a given center due to a decrease in enrollment, curriculum change, child program change, age grouping change, and budgetary limitation, or other circumstances producing a similar effect upon unit member assignments.
- 22.10.4 When specific consolidations are to take place, the center administrator will inform the unit member and discuss if requested the unit members modifications required.
- 22.10.5 Before any consolidation actually takes place, the center administrator shall arrange a conference and discuss with the unit member who is being consolidated the necessity for the consolidation of the position.
- 22.10.6 The unit member to be consolidated may telephone or request an appointment with the Director of Human Resources to discuss all known vacancies and possible future vacancies to be listed.

Center consolidations shall be effective on either January 31 or June 30. For consolidations that are scheduled to take place on January 31, the affected unit members shall be given appropriate written notice no later than December.

- 22.10.6.1 For consolidations that are scheduled to take place on June 30, the affected unit member shall be given appropriate written notice no later than May 15.
- 22.10.7 Non-classroom consolidations may take place anytime during the school year. The Employer shall determine the necessity of a consolidation. When specific consolidations are to take place, the administrators of the sites involved will inform unit members and discuss if requested, the unit member's modifications required.
- 22.10.8 Unit members to be reassigned because of consolidation of their positions will have preference in applying for listed vacancies in accordance with assignment factors. The District of Human Resources shall notify the unit members in writing of the new assignment as soon as possible, and provide one working day of released time for moving from one site to another.
- 22.10.9 The Employer will provide information to the Association upon request prior to district-wide consolidations.

22.11 Transfer/Consolidation Due to Center Being Closed/Replaced

Unit members will be assigned to centers where the children from the original site have been placed. Unit members will follow the students. In the event all the unit members cannot follow the students due to changes in enrollment, the consolidated factors shall determine who follows the students.

- 22.11.1 When the facility is rebuilt, all unit members who were in the original center shall have the first opportunity to return to the new facility. If more unit members desire to return than there are positions available, the assignment factors may be considered. After unit members in the original center have had an opportunity to be assigned to the new facility and vacancies still remain then procedures for listing a vacancy under the unit members transfer article shall be followed.
- 22.11.2 Unit members assigned to a center prior to reconfiguration shall have the option of remaining at the center after reconfiguration. In the event that all unit members cannot remain after reconfiguration due to enrollment decreases, the consolidation factors will be used to determine who is to be consolidated. Unit members who wish to follow their students to another center due to enrollment shifts required by reconfiguration shall be granted the opportunity to do so in accordance with the assignment factors. Unit members who do not want to continue in their assignment due to reconfiguration shall be granted an Employer-initiated transfer and/or may apply for listed positions.

22.12 Unit Member Assignment

22.12.1 Unit members shall be given written notice of center assignment when the center program is completed in June for the ensuing year. The unit members shall be so informed in writing, and upon request, the center administrator shall have a conference with the unit member. The unit members shall be consulted individually regarding any change in the unit member assignment due to unanticipated circumstances after the initial assignment, and shall be so informed in writing.

- 22.12.2 Unit Members assigned to a center shall work with any age-grouping served by the center as assigned by the center administrator. However, the site administrator will consider requests of individual unit member to work with specific age groups.
- 22.12.3 In order to assure that the pupils are taught by unit members working within their areas of competence, unit members shall not be involuntarily assigned outside the scope of their teaching certificate.
- 22.12.4 The Superintendent shall assign all newly appointed unit members to specific centers for which the Board has authorized employment. New unit members shall receive notice of assignments as soon as practical.

22.13 Hours of Work

- 22.13.1 Work Year CDC Preschool and CDC School-age
 - 22.13.1.1 The work year for CDC Preschool and CDC School-age permit unit members will be 208 days.
 - 22.13.1.2 CDC Preschool and CDC School-age unit members shall schedule their non-scheduled day(s) by mutual agreement with the site administrator. Site administrators shall not unreasonably prevent mutual agreement. Not more than one-half of the certificated unit members may be off work at one time. If these provisions have been met, and two or more unit members wish to be off from work at the same time, seniority in the District shall be the determining factor. Non-scheduled days will be prorated for unit members hired after July 1st of any given year.
 - When two or more teachers request the same non-teaching days and mutual agreement has not been reached, the teachers concerned may request a conference with the site administrator. The site administrator will schedule a conference within five days of the request before seniority is invoked.
 - 22.13.1.4 When an Early Childhood Education site work calendar is mutually agreed upon with the site administrator and approved by the Employer for the following center year it may not be changed, except that unit member shall be entitled to reschedule their teaching days in emergency situations such as extended personal illness or injury or emergencies related to members of their families. This provision shall not be interpreted to require any other unit member to reschedule their teaching days.
 - 22.13.1.5 When requested by the Employer to work beyond the contract year, unit members shall be paid per diem on the basis of their annual scheduled salary. Such additional duty shall be by mutual agreement. Unit members who upon their own request work beyond the contract year shall be paid the regular/non-long-term substitute rate.
 - 22.13.1.6 The Early Childhood Education CDC Preschool-CDC School-age and the State Preschool calendars can be found in Appendix 422.13.1.5)" \s "Appendix 4" \c 8 .
 - 22.13.1.7 There shall be two planning days each year.
 - 22.13.1.8 One half-day per year will be scheduled by the site administrator for all unit members at that individual site to be used for group planning, in-service, or articulation with elementary schools. Whenever possible,

the half-day will coordinate morning and afternoon session unit members so that substitute costs will be minimized.

- 22.13.1.9 The Employer shall provide each unit members at a Center with a copy of the appropriate annual ECE calendar.
- 22.13.1.10 Unit members hired prior to July 1, 2006 may voluntarily request to reduce their work year to the State Preschool work year, provided a State Preschool vacancy exists. Such requests shall not be unreasonably denied.

22.13.2 Work Year - State Preschool Program

- 22.13.2.1 The work year for State Preschool unit members hired on or after July 1, 2006 shall be 183 days, or the number of days that State Preschool students are scheduled for classes, plus three (3) additional days of professional development activities as scheduled by the District for a total of 186 duty days.
- 22.13.2.2 The work year for State Preschool unit members hired prior to July 1, 2006, shall be 208 days, consistent with the Regular CDC Work Year. State Preschool unit members on the 208-day work year shall not be eligible to use non-scheduled days on days on which State Preschool students are scheduled for classes.
- 22.13.2.3 Summer Work for 186-day State Preschool Teachers. Qualified State Preschool teachers shall have priority over substitutes for substitute assignments that occur in ECE summer programs outside of the State Preschool work year. State Preschool teachers shall submit their interest in filling such substitute assignments in writing to the ECE director or designee by March 1st. The District shall provide a copy of the list of volunteers to the Association. Such assignments shall be at the appropriate ECE Permit Teacher substitute rate.
- 22.13.2.4 The unit member shall be guaranteed a duty-free lunch period, preparation and break. If a unit member is required by the site administrator to perform a duty that interferes with the duty-free lunch period, student-free preparation period or break, the site administrator shall reschedule the duty day to provide for such comparable periods of time, or the employee shall be compensated at the extra-duty rate.
- 22.13.2.5 A unit member who is required to work beyond the duty day for more than 15 minutes shall be compensated at the extra-duty rate.
- 22.13.2.6 A Teacher in Charge (Opening) and Teacher in Charge (Closing) shall be assigned by the site administrator. These assignments shall be reduced to writing and these unit members shall be notified of their assignments prior to the start of the center year. Temporary Teachers-in-Charge (Opening or Closing) may be assigned as necessary. These assignments will be reduced to writing when possible.
 - 22.13.2.6.1 Unit members assuming these duties in the absence of the site administrator shall be paid an annual stipend. The stipend shall be paid on a pro rata basis to all of the teachers providing these services. (See Appendix 1 ECE Teachers Certificated/Permit Salary Schedule.)

- 22.13.2.7 Responsibilities of the Teacher-in-Charge include:
 - Opening or closing of ECE Center and
 - · Receiving State documents and
 - Assigning students to individual classes in the event a substitute cannot be secured and
 - Referring parents to the site administrator in case of specific concerns and
 - Responding to emergency situations as necessary
 - Other related responsibilities shall be consistent with Section 22.13.4.
- 22.13.3 Certificated Early Childhood Education unit members who provide substitute coverage to parts of classes for which substitutes are not provided by the Employer under circumstances where a substitute is normally authorized, shall be granted the pay which would have been paid to a daily substitute teacher. Such pay shall be paid proportionately to the teacher involved. Children shall be distributed in the fewest groups possible and in the most appropriate age levels possible.
- 22.13.4 A unit member who temporarily assumes the role of the site administrator without action having been taken by the Board will have administrative responsibilities. The requirements include use of independent judgment, excluding the right to hire, transfer, suspend, layoff, recall, promote, discharge, assign, reward, discipline other employees, to assign work, direct and adjust grievances or recommend such action.

22.13.5 General Provisions

- 22.13.5.1 Lesson preparation, study, student evaluation, conferences, etc. are to be conducted during those periods of time so scheduled. Unit members shall have the flexibility as to location at the site to perform such functions, provided that such performances do not interfere with their regularly assigned duties and those of others at the site.
- 22.13.6 A Labor/Management Advisory Council shall be established that includes the Director of Early Childhood Education, assigned central office staff, and designated site administrators who meet with site representatives and other unit members from various sites. All unit members are eligible to participate in this voluntary activity.
 - 22.13.6.1 The purpose of the council is to improve the dissemination of information and to promote on-going articulation between unit members and the Director. The goal of this partnership should include the improved understanding of program needs in order to foster positive public information skills for all unit members who participate.

22.13.7 Program Expansion/New Facilities

22.13.7.1 All vacancies shall be listed.

22.13.8 Teacher/Child Maximums

22.13.8.1 Teacher/child maximum is defined as the maximum number of pupils who may be in attendance with each unit member. The Employer shall adhere to the following unit member/child maximums in the District Early Childhood Education Centers as established by the State Department of Education:

	Teacher/Child Maximum	Adult/Child Maximum
0 yrs to 2 yrs	1:18	1:3
0 yrs to 3 yrs	1:16	1:4
3 yrs to 6 yrs (Preschool to Kindergarten)	1:24	1:8
6 yrs to 10 yrs (1st to 3rd)	1:28	1:14
10 yrs to 14 yrs (3rd to 8th)	1:36	1:18

- 22.13.9 Exceptions to class size maximums may be taken when space is inadequate, when health and safety of students could be imperiled.
- 22.13.10 Waivers to class size maximums may be made upon mutual agreement of the site administrator and the unit member(s) involved.
- 22.13.11 In the event that the State Department of Education changes its established ratios, the Association and the District shall reopen negotiations within 20 center days to discuss what effect, if any, the change may have on the provisions of this Article.

ARTICLE 23 - SUMMER SCHOOL EMPLOYMENT

23.1 General Provision

Summer School is established on the basis of need in the District and available funding. The following articles of the contract to the extent that they are not specially and explicitly in conflict with the Summer School Program shall govern terms and conditions of employment for Summer School employees: 1, 2, 3, 4, 5, 6, 7, 8, 9, 10 (to the extent that such Articles apply to substitutes), 13, 14, 15, 17, 19, 21 and 24.

In addition to the provisions of the contract applying to Summer School cited above, herein, the following specific terms of employment shall apply:

23.2 Calendar

Summer schools ordinarily start on the Monday, which occurs ten calendar days following dismissal of the regular session.

Elementary summer sessions are ordinarily 29 school days in length; secondary are ordinarily 24. Modifications of this calendar may be made by the Employer at its discretion.

23.3 Staff Selection

Qualified unit members from within the District shall be considered for placement. Should qualified District unit members not be available, the Employer may accept candidates from outside the District.

- 23.3.1 Summer School positions in this unit shall be posted, usually not later than the end of April. Qualifications for the position shall be included in the listing.
- 23.3.2 Applicants are to use the "Certificated Summer School Application Form" in responding to posted openings.
- 23.3.3 Summer school unit members shall be selected based on the following criteria:
 - 23.3.3.1 Qualifications (including for specific needs of specific assignment stated in the posting).
 - 23.3.3.2 Balance of site staff on the basis of a range of certificated services in the District that is roughly reflective of the overall District distribution of experience for certificated service.
 - 23.3.3.3 Unit members shall not normally be selected for more than two consecutive years. If there are no applicants with proper qualifications for a position, unit members may serve for more than two consecutive years. This provision shall be stated on the listing.
 - 23.3.3.4 Unit members applying for Summer School shall be given preference in inverse order from when they last taught summer school within the preceding three-year period. All other factors being equal, seniority shall prevail.
 - 23.3.3.5 Employment shall be contingent upon enrollment.
 - 23.3.3.6 Unit members selected shall be notified in a reasonably timely manner and not later than the last duty day of the regular school year when possible. Unit members selected for the alternate/stand-by list for possible Summer School employment based on enrollment shall also be so notified.

23.4 Evaluation

The Summer School programs and all unit members shall be evaluated on the basis of established goals and objectives and/or the District's standards of achievement.

23.5 Salary

Payment for Summer School work shall be made on the basis of the annual salary for the year preceding, and on a pro rata basis.

23.6 Alternate Teachers

The Employer agrees to employ substitute teachers in the District Summer School Program as alternate teachers. A minimum of one substitute teacher per school site shall be selected. If more alternate teachers are required, substitute or permanent teachers will be utilized. Compensation for alternate teachers shall be the Employer's prorated daily rate.

23.7 Sick Leave

Unit member regularly assigned to the District Summer School program will be permitted to use accumulated sick leave on an hour-for-hour basis to cover two days of absence during summer school caused by illness or injury.

23.8 Site or Program-Specific Summer Programs

- 23.8.1 Site or Program-Specific Summer Programs are differentiated from traditional summer school referenced in Article 23.1 by serving specific students from a specific site or sites within a specific program (e.g. Extended School year for Programs for Exceptional Children).
- 23.8.2 Should the District offer Site or Program-Specific Summer Programs, unit members at the participating site(s) or program(s) shall be selected from qualified, interested unit members at those participating site(s) or program(s). The Site Administrator at the site hosting the program shall select qualified unit member(s) and such member(s) from the participating sites or programs shall be given first priority because of their familiarity with the students and their needs. Where a program serves students from multiple sites, qualified, interested unit members at each of the participating sites or programs shall be considered in accordance with the criteria 23.3.
- 23.8.3 Qualified unit members from other sites or programs shall be given second priority in filling these positions.
- 23.8.4 Statement of Interest forms shall be distributed to all unit members at participating sites or programs by April 30th. All positions unfilled by unit members at a site shall be posted District-wide by May 15th.
- 23.8.5 Unit members shall not be selected for more than two consecutive years unless this is necessary to staff the site or program, or the qualifications for the positions require it. The District shall provide the Association with a list of those selected and the participating sites and programs by the first Friday in June.
- 23.8.6 A list of qualified unit members and the sites from which they were selected will be provided to the Association by October 31st.
- 23.8.7 Teachers shall be paid their hourly per diem rate of pay for Site or Program-Specific Summer Programs.

ARTICLE 24 - COMPENSATION

24.1 Compensation by Fiscal Year

- 24.1.1 Ongoing Increase: To address the high cost of living, all OEA members (with FTE) shall receive the following compensation increases:
 - 24.1.1.1 Effective July 1, 2022, salary schedules for FTE unit members shall be increased by a flat annual rate of \$1,000.
- 24.1.2 One-time Increase(s):
 - 24.1.2.1 FY 2021-22: In connection with the provision outlined in Article 10 of this Agreement (below), OEA FTE unit members (full time equivalents "FTE" of .5 to 1.0, including Early Child Education Teachers) shall receive a one-time off schedule payment of \$2000 no later than 60 days following the ratification of this Agreement by both OEA and the District's Board of Education.
 - 24.1.2.1.1 FTE unit members (including Early Childhood Teachers) with less than .5 FTE shall receive a prorated amount of the above one-time payment consistent with the unit members FTE.
 - 24.1.2.2 FY 2022-23: Additionally, to support retention, FTE unit members who complete and submit an intent to return form per the CBA between the Parties and upon verification of return for the 2022-23 school year, shall receive a one-time off schedule payment of \$1000 on or before October 30, 2022. Unit members must be employed by the District at the time of disbursement.
- 24.1.3 All compensation included in this section is contingent upon a determination by the Alameda County Office of Education that the combined financial impact of all tentative agreements with OUSD labor partners reached in the 2021-2022 school year do not endanger the fiscal well-being of the District. The Board's ratification of the agreement shall occur within 15 days of receipt of the AB 1200 letter from ACOE with such a determination.
- 24.1.4 Premium Extra Duty Rate: Effective May 30, 2022 through October 31, 2022 unit members who participate in Foundational Professional Learning outside of their normal work year and/or workday shall be compensated at a premium extra duty rate of \$45.00 per hour. (See Appendix 2).
- 24.1.5 Effective July 1, 2020 the parties agree to amend the K-12 Salary Schedule as follows:
 - i. One additional step (step 32) shall be included on the K-12 salary schedule.
 - ii. Step 32 shall be 5% higher than Step 31.

- 24.1.6 Effective upon ratification of this Agreement by the Association (March 4, 2019) substitute pay shall be as follows:
 - i. Daily Rate: 1-30 days per academic year: 70% step 1/column 1 of the K-12 salary schedule.
 - ii. Extended Rate: 31 days or more per academic year: 80% step 1/column 1 of the K12 salary schedule.
 - iii. STIPs and Retired OUSD teachers shall be eligible for the Long Term Rate on the first day of service as a substitute.
 - iv. Retention rate: Substitutes who work 60 days or more shall be eligible to continue at the Long Term Rate in the subsequent school year.

24.1.7 Nurse Retention Bonus and Salary Increase

- i. Nurses shall receive a \$10,000 annual bonus in May of 2020 and May of 2021
- ii. Eligibility: All nurses must be currently employed by the District at the time of disbursement to be eligible.
- iii. Effective June 30, 2021 at 11:59pm a separate salary schedule shall be created to increase current nurse salaries by 9% (based on salary placement as of June 30, 2021).

24.2 General Provisions for Placement on Teachers Salary Schedule

24.2.1 Registration of Credentials

- 24.2.1.1 The Education Code provides that no person shall be paid for services in a position requiring certification qualifications until the credential has been registered with the County Superintendent of Schools.
- 24.2.1.2 It is required that certificated unit members shall have on file in the Human Resources Services and Support copies of the credentials authorizing the services, subjects or grades they are teaching. The County Superintendent of Schools issues copies upon the receipt of the unit member's credential in that office.
- 24.2.2 Salary warrants shall be issued on the last working day of each month, as specified in Education Code section 45048.
- 24.2.3 All unit members will be paid in accordance with the appropriate salary found in the Appendices. (See Appendix 1.) Effective July 1, 2015, all Adult Education unit members will be paid in accordance with the "Salary Schedule Adult Contract Teachers" schedule. (See Appendix 1.4)

24.2.4 Step Increase Requirements

- 24.2.4.1 Initial Placement: Credit for service shall be allowed on the Salary Schedule at the rate of one increment (step) for one year of certificated teaching service, up to a maximum of twelve (12) increments.
- 24.2.4.2 All current District employees hired as an Adult Education unit members shall be placed at a step above and closest to the employees current pay rate. Adult Education unit members who are hired from outside the District shall be placed on Step 1 of the salary schedule.

- 24.2.4.3 All unit members shall advance one step on the salary schedule for each year of service in the District except those whose placement is at the maximum step for their class.
- 24.2.4.4 Unit members must work seventy-five (75%) of the regular full-time days that school is in session in order to have the year count as a year of experience on the Salary Schedule. Adult Education unit members must work seventy-five (75%) of the full-time equivalent hours in order to have the year count as a year of experience on the salary schedule.
- 24.2.4.5 Unit members participating in a shared teaching assignment for less than 75% of the regular full-time days that school is in session shall receive credit for a year of experience on the Salary Schedule for each two (2) years in which they participate in such an assignment. Adult Education unit members working less than seventy-five percent (75%), but more than thirty-eight percent (38%) of the full-time equivalent hours shall receive credit for a year of experience on the salary schedule for each two (2) years of service below seventy-five (75%).
- 24.2.5 Upon application for initial employment, a unit member may be granted salary schedule credit, at the discretion of the Director of Human Resources for previous non-teaching experience. The fully verified non-certificated paid service must be deemed by the Employer to be in a shortage field and be directly related to the certificated assignment.
 - 24.2.5.1 Credit will be evaluated on the basis of one (1) step for each two (2) years of acceptable experience within the past ten years. Outside experience credit that is granted is limited to a maximum of five (5) years on the salary schedule.
 - 24.2.5.2 The Employer's decision in granting said non-teaching experience shall be made prior to the candidate's employment. Subsequently, a unit member may apply to the Director of Human Resources for additional non-teaching experience credit within the time frames identified herein.
- 24.2.6 A teaching assignment in the Peace Corps or Vista shall receive year-for-year credit for such service on the salary schedule up to a maximum of five (5) steps. Requests will be reviewed by the Director of Human Resources, who will determine if the experience will receive credit. Active military service will be credited according to the following table.

2-3 years	1 step
4-5 years	2 steps
6-7 years	3 steps
8-9 years	4 steps
10+ years	5 steps

24.3 Column Requirement

24.3.1 All course work approved for initial placement must be verified by official transcripts. Obtaining official transcripts is the responsibility of the unit members. All transcript verifications must be received within sixty (60) days of the signing of the unit member's initial contract.

- 24.3.1.1 Earned degrees received and semester units earned at an accredited college or university shall be allowed for initial placement and subsequent horizontal movement on the Salary Schedule.
- 24.3.2 Units offered to meet requirements for advancement to Columns 2-6 must be completed in an accredited college or university that offers a degree beyond BA.
 - 24.3.2.1 Transcripts must be official and bear the Registrar's seal and signature. All work submitted for salary class advancement must carry upper division and graduate credit. Graduate credit is established by the institution offering the work and is validated by an official transcript.
- 24.3.3 Graduate and upper division semester hours (units) as defined by the particular accredited college, University or institution will be acceptable for placement and/or advancement on the salary schedule. Graduate and upper division quarter hours (units) shall be converted to semester hours (units) by multiplying the total of such hours (units) by two-thirds.
 - 24.3.3.1 Lower division courses taken after the BA is earned and taken at accredited institutions, as stated above, may be accepted for Salary Schedule credit if approved in advance by the Director of Human Resources. The approval shall be on the basis that such hours are in courses, which pertain, to/or may be reasonably expected to enhance the work skills of the unit member.
 - 24.3.3.2 Course work for salary placement of new employees will initially be evaluated by the Human Resource Division, based upon the same criteria as defined herein except that no prior approval shall be required.
- 24.3.4 Equivalency units in the form of travel, foreign study, authorship, special experiences, related summer work experience and other activities, which can be considered as related to the teaching assignment, shall be considered.
 - 24.3.4.1 Maximum Thirty (30) equivalent units is the maximum allowed between Column 1 and the doctoral equivalency. The acceptability of all equivalent units referenced in this section shall require the approval of the Director of Human Resources before the activity is undertaken.
 - Travel Equivalency credit may be granted for planned and specific travel that is related to the unit member's assignment. Such travel would be preceded by study, would feature an educational itinerary, and would be followed by direct and obvious classroom applications. The plan of travel must be approved in advance by the Director of Human Resources. Upon return or before the third Friday in November, the Post Educational Travel form must be completed before credit will be granted. Approved travel shall be evaluated at the rate of one (1) unit per week for foreign travel and one-half (1/2) unit per week for domestic travel. Trips of less than three (3) weeks' duration shall not be considered. The maximum credit for travel in one summer is six units and the total maximum is twelve (12) units.
 - 24.3.4.2.1 Ordinarily no equivalency credit is allowed for prior travel. In exceptional cases, where there has been extensive travel having a direct relationship to the teaching assignment, the teacher may apply to the Director of Human Resources for a special evaluation. Credit is

not granted for travel experience acquired before the Baccalaureate Degree.

- 24.3.4.3 Related Summer Work Experience Unit members may be given credit for work experience during the summer if the work is related to their assignment.
- 24.3.4.4 Workshop, private study and work at specialized schools such as art, business, and music will be evaluated by the Director of Human Resources. Any credit value for the work will be calculated on the following basis:
 - University lecture rate
 - 15 hours, including preparation time = 1 unit
 - Laboratory work
 - 30 hours, including preparation time = 1 unit
 - Studio work
 - 45 hours, including preparation time = 1 unit
- 24.3.4.5 District in-service training units will be calculated at a rate of one (1) unit for each fifteen (15) hours of approved in-service training participating and usual prep time.
- 24.3.5 In order to be upgraded on the salary schedule for a full year, the unit member must file any materials for upgrading according to the following schedule:

<u>Assignment Type</u>	Material Filing Date		
10 months	Third Friday - Nov.		
11 months	Third Friday - Oct.		
12 months	Third Friday - Sept.		
Year-Round School	Third Friday - Sept.		

24.3.6 For a change of classification for a school year, credit must be earned and courses completed before the first school day of the second semester. Materials must be submitted by the following schedule.

Assignment Type	Material Filing Date
10 months	Third Friday - April
11 months	Third Friday - March
12 months	Third Friday - Feb.
Year-Round School	Third Friday - Feb.

24.3.7 A Unit member who has completed requirements for upgrading on the salary schedule prior to these dates but, because of circumstances beyond their control is unable to present evidence to this effect, shall be reclassified for either the Fall or Spring Semester, whichever is relevant, and retroactive salary shall be allowed when the unit member presents proper verification of successful completion of requirements.

24.4 Anniversary Increments

An additional amount will be added to the basic salary as provided in the salary schedule to an employee after twenty or twenty-five years of creditable service in

Oakland. Creditable service shall be defined as the total number of full years of outside experience granted at the time of employment plus the total number of years served in Oakland since first employment in the District. An employee who has completed twenty or twenty-five years of creditable service shall notify the Human Resources Services and Support of their eligibility by completing forms provided for this purpose by the Human Resources Services and Support.

24.5 Special Rates for Certain Unit Members

24.5.1 Stipends

- 24.5.1.1 Assignment to a position that qualifies for payment of a stipend shall be for one year only. If the position continues beyond one year, non-reassignment shall be for good cause only.
- 24.5.1.2 Positions in this category are Teachers on Special Assignment, Senior High School Department Heads, including Librarians and Head Counselors, Head Teachers, Audiometrists, ROTC instructors, and yearbook supervision.
- 24.5.1.3 Authorization for payment of such stipends shall automatically terminate at the expiration of that year, unless the unit member is notified, to the contrary by the Board of Education or its authorized representative.

24.5.2 Other Stipends

- 24.5.2.1 Placement of Elementary Pupils in Other Classrooms when Substitutes are not provided.
 - 24.5.2.1.1 Teachers not receiving their preparation periods shall receive the following compensation: The rate of pay shall be the teacher's hourly per diem from beginning of the 2000-01 school year. (See Appendix 1324.5.2.1)" \s "Appendix 13" \c 8)
 - 24.5.2.1.2 When teachers provide substitute coverage to parts of classes for which substitutes are not provided by the Human Resources Services and Support under circumstances where a substitute is normally authorized, the pay which would have been paid to a daily substitute unit member shall be paid proportionally to the unit members involved. (See Appendix 13)
- 24.5.2.2 Use of Regular Classroom Teachers for Preparation Period Substitute
 - 24.5.2.2.1 When the regular classroom teachers must cover the preparation period because no substitute has been provided, under circumstances where a substitute is normally authorized, the unit member will be paid per period as provided in the salary schedule.
 - 24.5.2.2.2 In order for the regular classroom teacher to receive payment for preparation period substitute duty, the teachers is required to work 50 minutes on one day during the same calendar month or within a month of the day of service. This work must be accomplished and approved before or after school hours at the school site by the Principal/Site Leader.
- 24.5.2.3 Use of Conference Period for Substitute Duty at Secondary Level
 - 24.5.2.3.1 When a regular substitute has not been assigned from the Employer, secondary school unit members shall be paid per period as provided

in the salary schedule for each conference period used for substitute teaching under circumstances where a regular substitute is normally so assigned, and when such unit members remain on duty in the school after regular working hours for an amount of time equivalent to the substitute service time on any day during the same calendar month of the day of services.

- 24.5.2.4 Unit members who teach an additional regular on-going teaching assignment beyond their regular assignment for grades 6-12, will receive pro-rata salary schedule pay.
- 24.5.2.5 National Board Certification Stipend
 - 24.5.2.5.1 National Board Certification \$1,000 per year, from Measure E

24.6 Salary Provisions for Certain Unit Members

24.6.1 Nurses

- 24.6.1.1 Nurses whose hospital training and other college work culminated in the awarding of a Bachelor's Degree will be placed on the unit members' salary schedule in accordance with usual procedures. The possession of a Public Health Nursing Certificate and Bachelor's Degree will qualify nurses for placement on Column 3.
- 24.6.2 Teachers With Industrial Arts Credentials Not Teaching Vocational Courses
 - 24.6.2.1 Initial placement will be on Column 3. The usual rules for establishing five, six, and seven-year training apply.
- 24.6.3 Teachers with the Standard Designated Subject 8.0 full-time Credential to Teach Industrial Education and Teaching Vocational Courses.
 - 24.6.3.1 The initial placement of a unit member with a Standard Designated Subject Credential will be on Column 4.
 - 24.6.3.2 Trade experience beyond seven (7) years will be counted as unit member experience for placement purposes.
 - 24.6.3.3 Additional academic training of at least 60 units will be considered advanced training and will entitle the instructor to placement on Column 4.
- 24.6.3.4 Academic work totaling at least 90 units, in addition to the Bachelor's Degree, will meet the requirements for the doctoral equivalency (Column 6).
- 24.6.4 Teachers with a Clear Regular Teaching Credential Plus the 8.1 Occupational Credential and Teaching Vocational Courses
 - 24.6.4.1 Initial placement will be on Column 3 for teachers who will be teaching vocational courses.
 - 24.6.4.2 Credit will be allowed for industrial experience applicable to the teaching field on a ratio of one academic year (or 30 units) for each two year's industrial experience.
 - 24.6.4.3 Teachers may start on the salary schedule no higher than Column 4 and may qualify for the doctoral equivalency.
- 24.6.5 Unit Members with Designated Subjects Credential (Ryan) to Teach the Vocational or Non-Vocational Subject Named on the Credential

- 24.6.5.1Initial placement will be on Column 3.
- 24.6.5.2 Trade experience beyond five (5) years will be counted as teaching experience for placement purposes.
- 24.6.5.3 Advancement on the salary schedule beyond Column 3 will be in accordance with items (24.7).

24.7 Work Year for Unit Members Paid on the Teachers' Salary Schedule

24.7.1 Basic Assignments

24.7.1.1 Unless otherwise noted, unit members paid on the Teachers' Salary Schedule shall be on duty all days of the school calendar scheduled as teacher duty days.

24.7.2 Nurses

24.7.2.1 The work year for nurses shall be the same as for teachers.

24.7.3 Psychologists

School psychologists are required to be on duty and perform services for the five days immediately preceding the day teachers are required to report in the Fall, two days at the close of school in June, three days at the employee's option during the fiscal year from July 1 to June 30, on days when schools are not in session and the school buildings are open.

24.7.4 Counselors

- 24.7.4.1 The work year for full-time counselors shall be the same as for teachers, except that they shall be on duty five (5) additional days, prior to the first duty day for teachers and five (5) additional days after the last duty day for teachers.
- Variations to the aforementioned schedule of days, but not the total number of days may be made with the approval of the Principal/Site Leader of the school. Counselors shall be paid a proportionate per diem rate for these extra duty days. The additional duty days and per diem day for part-time counselors shall be proportionate to that of a full-time counselor.
- 24.7.4.3 For the purpose of calculating retirement, all the pre-and post-days shall be considered as part of the contract year.

24.7.5 Eleven and Twelve-Month Assignments

24.7.5.1 Unit members paid on the Teachers' Salary Schedule who work on eleven or twelve month assignments shall be on duty for those days described in Section 10.1 above, plus eighteen (18) additional days for an eleven-month assignment and thirty-six (36) additional days for a twelve-month assignment. Eleven-month unit members paid on the unit members' Salary Schedule shall receive their basic salary, including stipends, plus an additional 10%. Twelve-month unit members paid on the Teachers' Salary Schedule shall receive their basic salary, including stipends, plus an additional 20%.

24.7.6 Method of Payment

- 24.7.6.1 Ten (10) installments unless the unit members request that his or her salary be paid in twelve installments.
- 24.7.6.2 Commencing with the 2018-19 school year, the first salary installment for a school year for employees on a ten installment plan shall be payable the last working day of August.
- 24.7.6.3 Requests for payment on the twelve installment plan shall be made to the Human Resources Services and Support prior to August 15 or within two weeks of the first duty day of unit members hired after the opening of school.
- 24.7.6.4 The first salary installment for unit members on the twelve installment plan shall be payable on the last working day of August. The eleventh and twelfth installments shall be paid at the end of the months of June and July respectively.
- 24.7.6.5 Requests to change from the twelve installment plan to the ten installment plan shall be made to the Human Resources Services and Support prior to August.

24.8 Health and Welfare Insurance (See Agreement in Appendix 15)

- 24.8.1 Decisions regarding available medical, dental, vision, life insurance, accidental death, plan designs (including but not limited to coverages, deductibles, co-payments, pharmaceutical coverage, etc.), and employer and employee premium contributions shall be delegated to the Health Benefits Governance Board as stated in the Health Benefits Governance Board Agreement.
- 24.8.2 Employer Employee Contribution to Health Benefits
 - 24.8.2.1 In the 2005-2006 school year, there shall be no cost to the unit members for medical coverage for members or dependents.
 - 24.8.2.2 The Employer shall continue to pay, on behalf of all eligible unit members, the full cost of dental, vision, life, AD&D and long-term disability insurance.
 - 24.8.2.3 Effective July 1, 2006 and thereafter, the following changes shall be made in the Plan designs of the Least Expensive Health Maintenance Organization ("LEHMO") and Second Health Care Provider ("SHCP"):

	Co-Insurance							
LEHMO	Professional Services		Prescriptions			Dental & Vision, Life,	Dual Enroll-	
	OV	ΙΡ	ER	Gen	Brand Form AD&D, LTD		ment	
2005-2006	\$10	\$0	\$35	\$10	\$25	N/A	fully covered	Υ
2006-2007	\$10	\$250	\$100	\$10	\$30	N/A	fully covered	Υ
2007-2008	\$15	\$250	\$100	\$10	\$30	N/A	fully covered	Υ

	Co-Insurance							
SHCP	Professional Services		Prescriptions			Dental & Vision, Life,	Dual Enroll-	
	OV	ΙΡ	ER	Gen.	Bran d	Form	AD&D, LTD	ment
2005-2006	\$10	\$250	\$50	\$10	\$25	\$35	fully covered	Υ
2006-2007	\$15	\$250	\$100	\$10	\$30	\$50	fully covered	Υ
2007-2008	\$20	\$250	\$100	\$15	\$35	\$55	fully covered	Υ

- 24.8.2.3.1 OV Office Visit IP Inpatient Care ER Emergency Room Visit
- 24.8.2.4 Coverage for a newly employed, eligible unit member begins on the first day of the month following the month in which employed. When a new unit member or one who has moved from an ineligible to an eligible position does not exercise the option to select a health plan within the prescribed time limit, the employee is required to wait until the annual open enrollment period, which shall be May 15 through June 16 for FY 2005/2006, and each May thereafter. Those desiring to change from one health plan to another will be processed only during the annual open enrollment period.
- 24.8.2.5 Effective July 1, 2006 and until June 30, 2008, the implementation of a payment for health benefits equivalent to 0.5% (.005 times salary, only) of salary only, not to exceed in any year an annual amount of \$500 for any unit member in any medical plan year. This contribution will be part of an IRS 125 plan.
- 24.8.2.6 IRS 125 Plan
 - 24.8.2.6.1 Effective July 1, 2006 and each school year thereafter, the District shall set up and maintain an IRS 125 plan. Unit members may elect to participate in this plan to make pre-tax contributions for payments of medical co-pays, deductibles and any other legally allowable purpose.
- 24.8.2.7 Measure E Contributions To Medical Benefits
 - 24.8.2.7.1 Effective July 1, 2006, the District shall commit from Measure E, \$1.8 million or 9.7%, whichever is greater, collected in the 2006-2007 school year for medical benefit insurance premiums. Deposits are to be made monthly after July 1, 2006.
 - 24.8.2.7.2 Effective July 1, 2007, the District shall commit from Measure E, \$1.8 million or 10%, whichever is greater, collected in the 2007-2008 school year for medical benefit insurance premiums. Deposits are to be made monthly after July 1, 2007.
- 24.8.2.8 Effective upon the ratification of this agreement the Association and the District shall establish a Medical Benefits Reserve Fund (Fund). The sole purpose of the Fund shall be to pay OEA unit members' and dependents'/domestic partners' medical benefit premium costs, as determined and outlined below. Deposits into the Fund shall be held in a County of Alameda pooled interest investment account. All investment earnings calculated as of June 30th shall be transferred to the Fund.

- 24.8.2.9 Effective July 1, 2005 and until June 30, 2008, District increased costs during the term of this Agreement (if any), for medical insurance of the OEA unit shall have the Measure E contribution applied, then the contributions from OEA unit members for 2006/2007 and 2007/2008. If the total contributions from these two sources exceeds the cumulative increased costs of medical benefit premiums, the excess shall be placed in the Medical Benefits Reserve Fund on a monthly basis. These excess funds shall be used to pay for future medical benefit premiums.
- 24.8.2.10 No later than May 30 of each school year and thereafter, the District and the Association shall meet to determine the increased cost of medical benefits for the ensuing year and the estimated allocation of Measure E and member contributions, in accordance with these provisions. Final accounting of Measure E and member contributions will be made after the finalization of the District's annual audit for the prior year.
- Effective June 30, 2008 and thereafter, the District will pay 86% of the costs of medical benefit premiums, Measure E shall contribute 10% of medical benefit premiums, and OEA unit members shall pay 4% of medical benefit premiums in any school year. The individual unit member share shall be determined by identifying the total dollar amount generated by the 4%. This amount shall be used to determine the percent of salary for the unit members necessary to generate that total dollar amount. Each unit member's maximum dollar contribution shall not exceed \$700 per year. This contribution will be part of a premium-only IRS 125 plan and to be considered as part of the premium for payment of health insurance.
- 24.8.3 Employer's contribution to the cost of a health plan for a part-time unit member is prorated as follows:
 - Full payment for unit members working three-fourths time or more.
 - Three-fourths payment for unit members working half-time to three-fourths time
 - One-half payment for unit members working one-fourth to one-half time.
 - One-fourth payment for unit members working less than one-fourth time.
 - 24.8.3.1 Part-time unit members pay the difference between the Employer contribution and the full cost of the plan by payroll deductions made for the part-time unit member's share, according to the authorization signed upon enrollment in a plan.
- 24.8.4 Coverage for a newly employed, eligible unit member begins on the first day of the month following the month in which employed. When a new unit member or one who has moved from an ineligible to an eligible position does not exercise the option to select a health plan within the prescribed time limit, the employee is required to wait until the annual Open Enrollment period. Those desiring to change from one health plan to another will be processed only during the annual enrollment.

24.9 Health and Insurance Revisions 2003-2004

The following changes shall be effective September 1, 2003:

- 24.9.1 The Kaiser Health Plan shall be provided under the Kaiser III Plan.
- 24.9.2 The Health Net Health Plan shall be provided under the Health Net V Plan.

24.10 Dental Plan

All eligible full-time unit members and their eligible dependents shall be eligible for coverage under an Employer-paid group dental insurance plan. Coverage under the Dental Plan begins six months from the first day of the month in which employed.

- 24.10.1 Where a unit member and spouse are each employed by the District in an eligible capacity, each may subscribe individually to the same plan, but each party may not be enrolled as dependents of both.
 - 24.10.1.1 In such cases, each dependent child may be enrolled in one plan or the other, but not in both. A new unit member who does not exercise the option of entering the dental plan within the prescribed time limit will be required to wait until the annual open enrollment during September of each year.
- 24.10.2 Dental plan maximum per enrollee \$1,500 annually.

24.11 Long-Term Disability Insurance

- 24.11.1 Each unit member in an eligible capacity is automatically covered in a disability income plan six months from the first day of the month in which work begins in the eligible capacity, at no cost to the unit member. The benefit will be paid starting at the end of the qualifying disability period of sixty calendar days or beyond expiration of all paid leaves, including vacation, should the accrued sick leave and vacation extend beyond sixty days. Monthly benefit is two-thirds of the first \$1,200 of basic pay, plus one-third of additional basic pay to a maximum of \$1,200 per month total benefit.
- 24.11.2 An amendment rider has been added to the Employer's long-term disability contract with the insurance carrier which provides that certificated unit member with five or more years of credited STRS (State Teachers' Retirement System) employment shall be covered by the Employer's policy for a maximum of one year, thereafter to be covered by STRS; and that certificated unit members with less than five (5) years of service credited by the STRS shall be covered by the District policy to age 65.

24.12 Life Insurance

- 24.12.1 Each unit member in an eligible capacity is automatically covered at no cost to the unit member at the time of employment to the limit of \$10,000 for death from natural causes, plus an additional \$10,000 if cause is accidental.
- 24.12.2 Enrollment is not required for this benefit, but each unit member should complete a beneficiary designation card at the time of employment or when a change is desired. Conversion option on group life insurance is available without evidence of insurability.

24.13 Liability Insurance

24.13.1 A unit member, during the normal performance of duty, is protected against claims. Since the determination of individual responsibility is often difficult to establish and often dependent upon court findings, it is recommended that unit members assure themselves of coverage, especially when transporting students to school-sponsored events, by carrying the minimum insurance required to meet state financial responsibility laws. Unit members are protected in case of injury during the

performance of normal duty to the extent defined in the Workers' Compensation provisions of the Labor Code.

24.14 Vision Care

Upon implementation of the Health Net HMO medical insurance plan, the current vision insurance program available to unit members shall be replaced for all unit members by the Vision Services Plan. In addition, the Employer shall provide supplemental insurance against lost or damaged glasses; upon reasonable proof or appropriate employee certification, said lost/damaged glasses shall be replaced annually.

Kaiser Optical Services shall be available for unit members enrolled in the Kaiser Health Plan.

24.15 Annuity Program

- 24.15.1 During the 1992/1993 school year, 1991/1992 annuity dollar contributions were paid only for those unit members who were eligible for the annuity as of June 30, 1992. The Annuity Plan was continued as then constituted through the 1991/1992 school year and for eligible plan participants accounts have been maintained by the annuity plan administrator.
- 24.15.2 Effective July 1, 1993, no further annuity contributions were made, and the 7.5% annuity contribution was added to the three OEA salary schedules, at all steps and columns for temporary, probationary and permanent unit members.
- 24.15.3 The Association and the Employer agreed to restructure the Board of the Supplemental Annuity Plan for certificated employees, as follows:
 - 1 Representative appointed by the Employer
 - 1 Representative appointed by the UAOS
 - 5 Representatives shall be appointed by the Association
- 24.15.4 The parties agree that the above changes were fully implemented during the prior contract and further agree that no changes or modifications are required in the current contract.

24.16 Health Benefits Governance Board (See Side Letter Agreement in Appendix 14)

The Association and Employer understand and agree to the necessity of health care cost containment. The parties will work with all employees to ensure health care costs are contained or reduced without sacrificing the quality of healthcare. The Association agrees to continue to participate in the Health Benefits Governance Board (HBGB). The parties also recognize the need to reauthorize Measure E to help offset the costs of health care premiums for all employees.

24.17 General Information on Employer-Paid Insurance Plans

- 24.17.1 Unit members on paid leave will continue to have Employer contributions according to the work assignment that existed at the beginning of the leave.
- 24.17.2 Unit members on non-paid leave for a full calendar month or longer have no Employer contribution to insurance plans. Except for those unit members on maternity leave, for whom the Employer will pay the fringe benefit premiums for health, dental, life insurance and vision care for that period of time during which job performance is not feasible for physical/medical reasons, as certified by a physician.
- 24.17.3 Unit members who receive long-term disability benefits, or who have been granted leave beyond paid leave for illness, maternity, child adoption, disability, or advanced

study or any other leave approved by the Employer, may continue their insurance and health benefits for one year by paying the full cost. Except that those who receive long-term disability benefits may continue coverage at their own expense for as long as the disability continues, or to retirement. Unit members granted leave on some other basis should make their own arrangement for health or dental insurance.

- 24.17.4 Employer contributions toward costs of health, dental, disability income and group life insurance for ten-and eleven- month unit members will be made for twelve months, from July 1 through June 30.
- 24.17.5 The Employer has made voluntary tax-sheltered annuities available for unit members who wish them. A tax-sheltered annuity is a plan for diverting a portion of earnings for cash payment to the purchase of an annuity, the Employer making the purchase for the unit member under an agreement. Provided that all legal provisions are met, such earnings are not taxable until withdrawn later in the employee's life under conditions of the specific plans. Theoretically, this will result in a lesser tax liability than if tax had been paid at the time of earnings.
 - 24.17.5.1 The Board is in no position to recommend that a unit member either participate or not participate in such a plan. Each unit member must make the choice, as a unit member does not have similar tax positions and financial goals. Unit members who choose to participate in the plan may apply on forms available at the Insurance Programs Office.
- 24.17.6 If a National Health Plan is passed by the Congress (or statewide California equivalent) during the term of this Agreement, which provides for Employer contributions, the Association and Employer may meet to discuss the plan upon request.
- 24.17.7 The Employer will provide health plan benefits to unit members when they retire until age 65 at the retiree's option and expense.
- 24.17.8 Benefits as provided in this article shall be continued for the duration of the contract.

24.18 Transportation Reimbursement

- 24.18.1 Unit Members who are required by the Employer to use their privately owned automobiles as part of their assignment shall be reimbursed for such use.
 - 24.18.1.1 Eligibility will be determined in accordance with the appropriate Administrative Bulletin currently in effect. All procedures and instructions contained in that Bulletin are pertinent; however, the following rates shall be effective for unit members during the term of this contract.

24.18.2 Flat Rate Reimbursement

24.18.2.1 Regular flat rate reimbursement is based on the number of months of the duty assignment. The annual rates are:

Assignment	Flat Rates		
10-month basis	\$ 681.00		
11-month basis	\$ 704.00		
12-month basis	\$ 728.00		

24.18.2.2 Variable cost reimbursement for those whose annual mileage rate is one thousand (1,000) miles or more will be paid at the maximum allowable per mile rate provided for in the Internal Revenue Code.

24.18.3 Special Flat Rate Reimbursement for High Mileage and Weekend Assignments

24.18.3.1 For Unit members who are required to provide a car continuously for weekend business use and for high mileage positions (over 5,000 miles annually) the fixed cost reimbursement will be \$911.00.

24.18.4 Excess Mileage Reimbursement

24.18.4.1 Unit members receiving regular or special flat rate reimbursement who may be eligible for excess mileage reimbursement over 1,000 miles should record all mileage performed. When their mileage exceeds the established rate, they should submit excess mileage claims monthly if possible, but in no case less than once each semester.

24.18.5 Mileage Reimbursement

24.18.5.1 Unit members who are occasionally required to use their privately owned automobiles in performing their duties but do not qualify for flat rate reimbursement will be reimbursed at the maximum per mile rate allowed for in the Internal Revenue Code.

24.19 Miscellaneous

24.19.1 Student Teaching

24.19.1.1 Cooperating unit members shall receive recognition for the teacher education responsibility undertaken in working with student teachers. Such recognition shall include the contract amount paid by the teacher training institution.

24.19.2 Teachers of Adults

24.19.2.1 Teachers of adults not paid as regular teachers shall be paid at the extra-duty rate.

24.20 Programs for Personal Growth

- 24.20.1 Any unit member receiving their first clear multi or single subject teaching credential must take 150 clock hours of professional growth each five (5) years to keep the credential valid.
- 24.20.2 Any unit member receiving credit for equivalency units as stated in Article 24, can apply those units to the 150-hour requirement of professional growth and vice versa.
- 24.20.3 Each unit of credit granted shall equal 15 hours toward the 150 of professional growth requirement. Thirty (30) equivalency units are the maximum allowed between Column I, and the Doctoral equivalent.
- 24.20.4 An individual program of professional growth shall consist of a minimum of 150 clock hours of participation in activities that contribute to competence, performance, or effectiveness in the profession of education. Acceptable activities shall be defined to include, among other acceptable activities, the completion of courses offered by regionally accredited colleges and universities; participation in professional conferences, workshops, teacher center programs, or staff development programs; service as a mentor teacher pursuant to Section 44496; participation in school curriculum development projects; participation in systematic

- programs of observation and analysis of teaching; service in a leadership role in a professional organization; and participation in educational research or innovation efforts.
- 24.20.5 In order for the units of credit to count towards the 150 hours of professional growth, the teacher must submit, for approval, their initial plan to the Director of Human Resources or their designee by October 1 for the Fall Semester and by February 1 for the Spring Semester. (Forms are to be provided by the Employer.)
- 24.20.6 Upon completion of the units of work, the unit member shall be responsible for having the appropriate person in charge sign a "Professional Growth form". Once the form is completed, the teacher must submit it to the Human Resource Services and Support.
- 24.20.7 The unit member must submit verification of 150 clock hours of professional growth at five (5) year intervals to the State Commission on Teacher Credentialing in order for their credential to be deemed to remain valid.

24.21 Domestic Partner Benefit

- 24.21.1 The Employer shall provide health insurance for unmarried domestic partners of eligible unit members as provided in this Article. "Domestic partner" is defined as a single member with an unmarried, unrelated individual who has lived with the unit member for not less than 12 months. To enroll, the unit member must complete an application form during the open enrollment period, accompanied by a notarized affidavit (see Appendix 15 (F2)24.20.1)" \s "F2" \c 8).
- 24.21.2 The unit member will be taxed on the cash value of the Domestic Partner premiums, in compliance with IRS regulations.
- 24.21.3 COBRA provisions do not apply for Domestic Partner coverage.

ARTICLE 25 - Peer Assistance and Review

25.1 General

- 25.1.1 The Association and the Employer are continuously striving to provide the highest possible quality of education. In order for all students to succeed in learning, all unit members must succeed in teaching. The Association and the Employer believe that all unit members should focus on continuous improvement in professional practice and that unit members having difficulties can benefit from the assistance and review of colleagues. Therefore the Association and the Employer hereby establish a unit member Peer Assistance and Review Program (PAR) for K-12 classroom unit members. PAR shall use the standards in the evaluation article of the Collective Bargaining Agreement. (Article 13 Performance Evaluation)
 - 25.1.1.1 Peer assistance may be provided to a beginning unit member or a volunteer permanent teacher and must be provided to a permanent unit member who has received an overall unsatisfactory evaluation in the areas of teaching strategies and instruction as provided for in the evaluation article in this agreement. The Peer Assistance shall be provided through Consulting Teachers as described in detail in Section 25.5. This assistance shall not involve the participation in or the conducting of the annual evaluation of unit members, except for making available to the evaluator the results of the referred unit members' participation in the program.

25.2 Definitions

25.2.1 Joint Committee

25.2.1.1 The PAR Program shall be governed by the Joint Committee composed of five (5) Association members and four (4) Employer members. One of the Association members, the Association President, and one Employer member, designated by the Superintendent, shall be ex-officio, non-voting, non-stipend members.

25.2.2 Consulting Teacher (CT)

- 25.2.2.1 A consulting teacher is a permanent teacher selected by the Joint Committee to provide assistance and to review the progress of the teachers participating in the program.
- 25.2.3 Participating Teachers: The three types of participating teachers are Volunteer, Referred, and Beginning.
 - 25.2.3.1 Volunteer Teacher (VT): A permanent teacher who wishes to participate in the program.
 - 25.2.3.2 Referred Teacher (RT): A permanent teacher who is referred to the program because of an unsatisfactory evaluation.
 - 25.2.3.3 Beginning Teacher (BT): A non-permanent, California preliminary or clear-credentialed teacher.
 - 25.2.3.4 Teachers on emergency credentials who are serving as classroom teachers of record or have a caseload assigned to them when a consulting teacher is working with a group of educators at a site or sites.

25.2.4 Peer Coach: A teacher who may be released on a part-time basis to work in the Program.

25.3 Joint Committee

25.3.1 Composition, Selection, Terms

- 25.3.1.1 The Joint Committee (JC) shall consist of nine (9) members composed of five (5) Association members and four (4) Employer members. Four (4) members shall be certificated classroom teachers who shall be selected by the Association. Three (3) members shall be administrators selected by the Employer. One of the Association members, the Association President, and one Employer member, designated by the Superintendent, shall be ex-officio, non-voting, non-stipend members.
- The Association and the Employer shall individually determine the method for selection, the qualifications, and the term of its appointees, being mindful of the purpose of PAR and the value of continuity in administering PAR. To promote continuity, the terms shall be staggered so that no party's appointees are all new in any one school year. The initial terms shall be for two and three years.
- 25.3.1.3 The Joint Committee may call on additional resources as appropriate. The Joint Committee shall establish its operational procedures, including the method for selection of a chairperson.

25.3.2 Meeting and Composition

- 25.3.2.1 The Joint Committee shall establish its meeting schedule. Adequate notice will be provided for any additional meetings. To hold meetings a majority of the members, with a minimum of two teachers and one administrator, must be present. Meetings may be held during the school day, with a grant of release time to teachers, or during non-school time.
- 25.3.2.2 Association appointees to the Joint Committee shall receive a stipend of \$4,000 paid on a pro-rata basis for each year of service on the JC.

25.3.3 Decision Making

The Joint Committee shall make decisions by consensus of the entire body. If no consensus can be reached, the decision shall be made by a majority vote of seven voting members. For any decision to be made at a JC meeting, a majority of the JC members present shall be teacher members of the committee. A JC member shall recuse themselves from discussion or vote on any matter regarding:

- 25.3.3.1 An RT for whom they are the current evaluator or the evaluator who initiated the current PAR referral.
- 25.3.3.2 A participating teacher from their site or department.

In the case of such recusal, the Employer's ex-officio Committee member may vote on matters regarding the RT. A recused JC member shall have the same opportunity to provide input to the PAR process as any other administrator of an RT, for example, written statements or interviews with the JC.

25.3.4 Responsibilities

Joint Committee shall be responsible for the following:

- 25.3.4.1 Joint Committee Training
- 25.3.4.2 Development of Rules of Procedure
- 25.3.4.3 Consulting Teachers (CT)
 - Selection of CTs.
 - Training of CTs.
 - Reviewing/directing the CTs provision of services.
 - Reviewing peer assistance and review reports prepared by the CTs.
 - Evaluating the CTs.

25.3.4.4 Permanent Teachers

The Joint Committee shall provide participating permanent teachers, Volunteer or Referred, with a list of available CTs. The permanent teacher may indicate the CT that they prefer, but the Joint Committee shall not be bound by the preference. The Joint Committee shall notify the permanent teacher, their Principal/Site Leader, and the CT in writing that the permanent teacher is participating in PAR. The appointed CT shall be identified.

25.3.4.5 Cooperation between Consulting Teachers and Principal/Site Leader

25.3.4.5.1 The Joint Committee shall expect and strongly encourage a cooperative relationship between the consulting teacher and the Principal/Site Leader in the peer assistance and review process. The JC may recommend to the evaluator and/or the evaluator's immediate superior that an alternate evaluator be assigned for an RT's evaluation process.

25.3.4.6 Recommendations to Board of Education

25.3.4.6.1 The Joint Committee shall review PAR reports prepared by CTs and make recommendations to the Board regarding a referred teacher's progress in PAR, including, but not limited to, identifying a referred teacher who is unable to demonstrate satisfactory improvement after sustained assistance.

25.3.4.7 Annual Evaluation of Program

25.3.4.7.1 The Joint Committee shall evaluate annually the impact of PAR in order to improve PAR. The review and evaluation may include interviews or surveys of PAR participants, CTs, Principals/Site Leaders, and others as deemed appropriate. The Joint Committee shall submit this annual evaluation in writing, including any recommendations for improvement, to the Superintendent and the Association at the same time that it submits the proposed budget. This submission shall be no later than May 30.

25.3.4.8 Annual Budget

- 25.3.4.8.1 The parties agree that the annual budget for the PAR Program shall not exceed the revenue provided by the State in support of the PAR Program each year. In addition the parties agree that there shall be no encroachment into the Employer's unrestricted general fund to support the PAR Program.
- 25.3.4.8.2 The Joint Committee shall annually submit a proposed budget to the Superintendent. The proposed budget shall be designed to carry out the provisions of this Article and shall take into consideration:
 - The number of Consulting Teachers which will be required in the coming year in light of the projected level of participation in the program
 - The cost, if any, to augment the BTSA program adequately to support beginning teachers
 - The recommendations for improvement of PAR which it made in the annual report to the Board
 - Other relevant factors
 - Payment to Employer's members for participation on the Joint Committee
- 25.3.4.8.3 The proposed budget, including related administrative costs of not more than 5%, shall not exceed the state funding allocation for the coming year as estimated by the Employer's chief financial officer.
- 25.3.4.8.4 This proposal shall be submitted at the time and in the form requested by the Employer. The Joint Committee shall recommend to the Superintendent that the Board authorize the necessary number of consulting teacher positions or any increase or decrease thereof, providing to the Superintendent the rationale for the request. The Board shall consider the requests promptly, normally in accord with a previously published calendar on budget review and position authorization.
- 25.3.4.8.5 At the conclusion of the fiscal year, if the revenue for the PAR Program exceeds expenditures for the PAR Program, the Joint Committee shall determine the allocation of the surplus in a manner that facilitates the professional development activities of the District within the intent of this Article.

25.3.4.9 Special Support Schools

- 25.3.4.9.1 The Joint Committee may, with Employer approval, provide assistance to teachers at special Support Schools. These schools are those with high teacher turnover and/or large numbers of inexperienced teachers or teachers with less than three (3) years experience.
- 25.3.4.9.2 The Joint Committee shall:
 - Identify the schools using indicators from the Human Resources Services and Support.
 - Identify the circumstances contributing to the teacher turnover.

 Coordinate support with the intent to build and keep capacity at the school.

25.3.4.10 Procedures

25.3.4.10.1 The Joint Committee shall adopt procedures for implementing the provisions of this Article. These procedures shall be consistent with the statutes, the provisions of this Article, and this Collective Bargaining Agreement. If there is any inconsistency, the statutes, the provisions of this Article, and this Collective Bargaining Agreement shall prevail.

25.3.5 Confidentiality

- 25.3.5.1 All materials related to evaluations, reports and other personnel matters regarding individuals, which are created or reviewed by the Joint Committee, shall be strictly confidential. Therefore, Joint Committee members may not disclose such information obtained by way of the program or in the peer review process with the following exceptions:
- 25.3.5.2 The Employer may make use of the following documents regarding referred unit members:
 - Final peer review reports prepared by Consulting Teachers.
 - Intermittent peer review reports prepared by Consulting Teachers subsequent to the acceptance of the final peer review report.
 - Recommendations of the Joint Committee or CT regarding participants in the Program.
 - Evaluations of a teacher's participation in the Program by the Joint Committee or CT.
 - 25.3.5.3 Materials shall be disclosed if required by law.

25.3.6 Indemnity

25.3.6.1 The Employer shall have the same duty to defend and indemnify Joint Committee members participating in the PAR program who are acting within the course and scope of their designated functions as it has to other district employees pursuant to Division 3.6, section 810 et seq., of the Government Code.

25.3.7 Non-Management/Supervisory Status

25.3.7.1 Functions performed by teacher Joint Committee members pursuant to the Program shall not constitute either management or supervisory functions as defined by subdivision and of section 3540.1 of the Government Code.

25.4 Participating Teachers

25.4.1 Beginning Teachers (BT)

25.4.1.1 A Beginning Teacher is defined as a non-permanent, California preliminary or clear-credentialed teacher, or a teacher on an emergency credential who is serving as a classroom teacher of record, or has a

- caseload assigned to them when a consulting teacher is working with a group of educators at a site or sites.
- 24.4.1.2 The PAR program for Beginning Teachers may include the BTSA program. BTSA may be supplemented with PAR funds as determined by the Joint Committee. All support provided the Beginning Teacher shall be confidential in accord with BTSA program guidelines.
- 24.4.1.3 The Employer shall provide the Joint Committee and BTSA with a list of all Beginning Teachers at the beginning of every school year, indicating the support program for which the Beginning Teacher is eligible, and shall supplement the list during the year as required.

25.4.2 Referred Teachers (RT)

25.4.2.1 Permanent unit members who have received unsatisfactory evaluation shall be referred to the Joint Committee and required to participate in the PAR program as an intervention.

25.4.2.2 Unsatisfactory Evaluation

- 25.4.2.2.1 PAR Participation shall not be required unless the procedures provided for in Article 13 (Evaluation) have been fully followed.
- 25.4.2.2.2 An unsatisfactory evaluation shall require participation in PAR when a unit member receives a summative evaluation on Form A of two (2) or more "unsatisfactory" ("beginning") ratings out of these four (4) evaluation domains:
 - i. Domain 1: Planning and Preparation
 - ii. Domain 2: Classroom Environment
 - iii. Domain 3: Teaching and Learning for All Students
 - iv. Domain 4: Professional Reflection and Revision
- 25.4.2.2.3 The summative evaluation shall be based on evaluation criteria appropriate to the unit member's instructional program, and consistent with the goals and objectives (SMARTe Goals) agreed to by the unit member and their evaluator at the outset of the evaluation process.

25.4.2.3 Referral

- 25.4.2.3.1 The administrator who authored the unsatisfactory evaluation shall refer the RT to the Joint Committee. The administrator shall provide the Joint Committee a copy of the unsatisfactory evaluation, together with appropriate documentation. A copy of the referral packet shall be sent to the RT.
- 25.4.2.3.2 The Joint Committee shall review the referral packet including the unsatisfactory evaluation and supporting documentation. It may also interview the referring administrator and the teacher being referred. Based on the needs of the teacher, the Joint Committee shall determine the nature of assistance, which the CT shall provide.

25.4.2.4 Teacher Objection or Non-Cooperation

25.4.2.4.1 A teacher who has been referred to PAR and objects to such participation on the grounds that the unsatisfactory evaluation leading

to the referral was procedurally or otherwise flawed, may at their request, appear before the Joint Committee with a representative of their choice to explain their point of view regarding the defects in the evaluation. Notwithstanding this provision, the Joint Committee shall proceed to assign a CT to the RT in accord with its normal provisions unless it is determined that the unsatisfactory evaluation was invalid. At the time an unsatisfactory evaluation is determined invalid, any reports and documentation generated through the PAR process shall be removed from the Referred Teacher's personnel file.

- 25.4.2.4.2 A Referred Teacher may be exited unsatisfactorily from the PAR process prior to the final report upon recommendation of the JC that they are not cooperating with the PAR process provided:
 - i. The assigned consulting teacher provides notice in writing to the Referred Teacher of the recommendation and the possible consequences of failure to cooperate no less than five (5) days prior to the meeting of the JC where the early unsatisfactory exit shall be considered.
 - ii. The assigned consulting teacher provides clear evidence of non-cooperation to the Joint Committee and the Superintendent or designee in writing.
 - iii. For referred teachers, the assigned consulting teacher provides clear evidence to support non-cooperation to the Joint Committee and the Superintendent or designee in writing.
 - iv. The Joint Committee considers the evidence, makes a recommendation, and notifies the Referred Teacher of the recommendation.
 - v. The provisions of Article 25.7 including but not limited to notice, right to respond, and representation have been met.
 - vi. For a referred teacher, the Superintendent or designee approves the recommendation.

25.4.3 Volunteer Teachers (VT)

- Any permanent classroom teacher may volunteer to participate in the Program for the purpose of obtaining peer assistance to improve performance. To participate in the program, a unit member may volunteer at any time. A volunteer may withdraw from the program at any time. When the volunteer applies to the program, they shall provide the program with an initial draft of the goals, which they would like to accomplish by participation in PAR.
- 25.4.3.2 The CT shall maintain a log of assistance, which shall not include any evaluative comments. A copy of this log shall be provided to the Joint Committee and the volunteer. It shall not be placed in the volunteer's personnel file unless the volunteer so requests.
- 25.4.3.3 All communications between the VT and the CT shall be confidential and shall not be shared with others, including the Principal/Site Leader,

evaluator, or Joint Committee, without the written consent of the VT and CT. The CT shall not without the written consent of the VT, be called by the Association or Employer to testify, produce documents, or to participate in any way in any proceeding involving the teaching performance of the VT if the subject of the inquiry is the teaching performance of VT during the period the CT assisted the VT as a Volunteer participant in PAR.

25.5 Consulting Teachers

25.5.1 Duties

A consulting teacher (CT) shall participate in training and provide assistance to Beginning Teachers, Referred Teachers, and Volunteer Teachers pursuant to the Program.

25.5.2 Qualifications

- 25.5.2.1 A CT should be considered by colleagues to be a highly skilled practitioner. A consulting teacher shall possess the following qualifications:
 - Permanent status with clear California credentials.
 - At least four recent years of teaching experience, and with at least four years in the District preferred.
 - Teaching ability as demonstrated by the evaluation article of the Collective Bargaining Agreement (Article 13 - Performance Evaluation).
 - Knowledge of subject matter.
 - Demonstrated mastery of a range of teaching strategies, instructional techniques, and classroom management skills necessary to meet the diverse needs of students.
 - Demonstrated ability to communicate effectively orally and in writing.
 - Demonstrated ability to work cooperatively and effectively with others.
 - A consulting teacher cannot be a member of the Joint Committee.

25.5.3 Posting and Application

- 25.5.3.1 There shall be continuous recruitment for the position of consulting teacher at all sites and in the District office.
- 25.5.3.2 Such recruitment shall be consistent with the Employer's affirmative action goals. There shall be a permanent job description and application in the District office and on the District Web Page. Applications may be submitted at any time and will be kept on file until the end of the current school year.
- 25.5.3.3 In addition to the application, an applicant shall submit letters of recommendation from:
 - Their Principal/Site Leader or immediate supervisor
 - An Association representative
 - Two teachers at the applicant's site/program

- 25.5.3.4 The Joint Committee may include additional procedures as it sees fit regarding the application process.
- 25.5.3.5 All applications and references shall be treated with confidentiality. They shall not be placed in the CT's personnel file.

25.5.4 Selection

- The Joint Committee shall review the applications and identify teachers for further consideration. The Joint Committee may contact any person submitting a letter of recommendation for a prospective CT. The Joint Committee shall observe a prospective CT in a classroom instruction setting. A prospective CT cannot be named to said position without first having been observed teaching by the Joint Committee. "Observations" in this section may be in person or via video recording. The Joint Committee shall also conduct interviews with the finalists.
- 25.5.4.2 Selection of CTs shall be consistent with the Employer's affirmative action goals. Written notification of selection or non-selection as a CT shall be given to each applicant. The Joint Committee may include additional procedures as it sees fit regarding the selection process. The Joint Committee shall determine the number of CTS needed.

25.5.5 Terms of Assignment

A Consulting Teacher shall be appointed for and agree to accept a three-year term, subject to annual renewal. A consulting teacher shall be placed in a classroom assignment for a minimum of one (1) year before reapplying to be a consulting teacher if the consulting teacher's out-of-class CT assignment was half-time or greater. If no qualified consulting teachers apply, the JC may elect to extend a CT's term by one year. Every effort shall be made to stagger the terms of consulting teachers to ensure continuity.

25.5.6 Release Time

A Consulting Teacher shall be released full-time to work in this program. A Consulting Teacher shall be provided sufficient release time for travel to and from sites, completing documentation, meeting with the Joint Council, regular meetings with the other consulting teachers, preparation time for class visits, making arrangements for support for the participating teacher and any other situation deemed necessary. A Consulting Teacher shall be provided sufficient release time for their own training and staff development.

25.5.7 Compensation

- A full-time Consulting Teacher shall receive their regular salary, benefits, and a stipend. A CT shall continue to accrue seniority as a regular certificated employee for the time served as a consulting teacher in the same manner and for the same purposes as if they had remained in a regular assignment. If a CT works beyond the regular teacher work year at Joint Committee direction, the CT shall be compensated in accord with the Collective Bargaining Agreement.
- 25.5.7.2 The CT shall be paid a stipend. The stipend shall be \$4,000 for each full year of service as a CT, or a pro rata amount for less than a full year of service. A change in this stipend shall be subject to bargaining.

25.5.8 Return to Regular Assignment

While on assignment as a full-time consulting teacher, the CT will be considered on leave to a categorical program from their regular assignment. Thus, their regular assignment may be filled on a temporary basis. Upon completion of their service as a full-time released Consulting Teacher, the CT has the right to return to their most recent teaching assignment regardless of the length of the CT assignment. The teacher filling the assignment left vacant by the CT must be informed in writing that the assignment is limited to the length of the CT's participation in the PAR program.

25.5.9 Caseload

25.5.9.1 The Joint Committee shall determine the caseload for a consulting teacher. The number is dependent on the amount of intervention time that the Joint Committee determines, in consultation with the CT, is necessary for each participating teacher. Consulting teachers shall assist the teachers on their caseload by demonstrating, observing, coaching, conferencing, referring, or by other activities which in their professional judgment, will assist the teacher in accord with the PAR process set forth below in Section 2725.6.

25.5.10 Appraisal

- 25.5.10.1 The Joint Committee shall oversee the work of the Consulting Teacher. The Joint Committee shall make a written evaluation of each CT's work by June 1 of each year, using a form developed by the Joint Committee. Reasons for removal may include the specific needs of the PAR program or inadequate performance of the CT. Removal shall be at the sole discretion of the Joint Committee.
- 25.5.10.2 Prior to the effective date of such removal, the Joint Committee shall provide the CT with a written statement of the reasons for the removal, and upon request, meet with the CT to discuss the reasons.

25.5.11 Peer Coaches

- Joint Council may also select a pool of Coaches who may be released on a part-time basis to work in the Program depending upon the needs of the Program, funding availability and teaching areas of Program participants. The cost of releasing coaches for service in the program shall be necessary release time or the Peer Coach's pro rata rate of pay for work beyond the regular work day or work year.
- 25.5.11.2 If a Peer Coach is released and required to work beyond contracted work hours, the extra time will be compensated at the pro rata rate of pay.

25.5.12 Indemnity

Employer shall have the same duty to defend and indemnify Consulting Teachers participating in the PAR Program who are acting within the course and scope of their designated functions as it has to other district employees pursuant to Division 3.6, section 810, et seq., of the Government Code.

25.5.13 Non-Management/Supervisory Status

- 25.5.13.1 Functions performed by Consulting Teachers pursuant to the Program shall not constitute either management or supervisory functions as defined by subdivisions (g) and (m) of section 3540.1 of the Government Code.
- 25.5.13.2 The Employer shall provide to the released Consulting Teachers appropriate, common work/office space, which shall include secure file storage.

25.6 Peer Assistance and Review Process for Permanent Teachers

25.6.1 Preparation of Assistance Plan

- As soon as possible after assignment, the Consulting Teacher shall review the referring packet for the RT or VT. Thereafter, the Consulting Teacher shall meet with the RT or VT and the Principal/Site Leader together or separately, as appropriate, to review the employee's performance and recommendations for improvement.
- 25.6.1.2 The Referred Teacher shall give input into the development of the plan. The Consulting Teacher will then prepare a written Assistance Plan aimed at remedying the deficiencies which were cited in the RT's unsatisfactory evaluation or meeting the goals set forth in the VT's application. The RT, the Assistance Plan will include:
- 25.6.1.3 Performance goals which are aligned with pupil learning and which are consistent with the Stull Bill and/or the OETF and the evaluation article of the Collective Bargaining Agreement. (Article 13 Performance Evaluation)
- 25.6.1.4 A projected date for completion, which will ordinarily be at the end of the current school year. The Plan will be submitted to the Joint Committee for final development and approval.

25.6.2 Classroom Observations

25.6.2.1 The Assistance Plan shall include multiple observations of the RT or VT by the Consulting Teacher. These observations will be in addition to any that are performed as part of the evaluation Article in this agreement.

25.6.3 Progress Reports

- 25.6.3.1 The Consulting Teacher shall prepare and discuss with the Joint Committee periodic (at least every three months) reports of the RT's or VT's participation and progress in the program. The Consulting Teacher's report shall include an assessment as to whether the Assistance Plan can or should be discontinued, whether the Plan needs revision, or whether the Plan needs to be extended beyond its original projected term.
- 25.6.3.2 For purposes of the VT, these reports shall be limited to the assistance log referred to in Section 25.4.3.2 above.

25.6.4 Final Report

25.6.4.1 By May 1, or at a later date if specified in the Assistance Plan, the Consulting Teacher shall make a final report to the RT or VT, and, for RTs, to the Joint Committee and the evaluator. A copy of the final report shall be included in the RTs personnel file after he or she has had an

opportunity to review and attach a written response to it within twenty (20) workdays after receipt of the report.

25.6.4.2 The Final Report shall not constitute the Employer's evaluation of the employee's performance but in the case of an RT shall (and, in the case of a VT, may at his or her request) be considered by the site administrator in preparing any evaluation document or proposing any personnel action.

25.7 Participating Teacher Due Process Rights

- 25.7.1 The participating teacher shall be entitled to review all reports generated by the Consulting Teacher and Principal/Site Leader prior to their submission to the Joint Committee, and to have their comments attached. To effectuate this right, the Consulting Teacher shall provide the permanent teacher being reviewed with copies of such reports at least five (5) working days prior to the meeting of the Joint Committee at which the reports will be considered.
- 25.7.2 The permanent teacher shall have the right to be represented by the Association in any meetings of the JC to which they are called and shall be given a reasonable opportunity to present their point of view concerning any report being made.
- 25.7.3 Elementary participating teachers shall not normally be assigned to a different grade level from the one in which they received an unsatisfactory evaluation except by their request. In no event shall an elementary participating teacher be assigned more than two (2) grade levels from the one in which they received an unsatisfactory evaluation except by their request.
- 25.7.4 Secondary participating teachers shall not normally be assigned more than one (1) different preparation from the one(s) in which they received an unsatisfactory evaluation except by their request.
- 25.7.5 The participating teacher shall have the right to timely reports of progress made.
- 25.7.6 The participating teacher may request a change in their Consulting Teacher no more than once during their participation in the PAR program providing an alternate Consulting Teacher is available.
- 25.7.7 Upon written request by the participating teacher, the record of PAR intervention shall be removed from the personnel file after four (4) years, if there have been no subsequent incidents of unsatisfactory service during said period.
- 25.7.8 Participation in the PAR Program shall not diminish the legal rights of bargaining unit members.
- 25.7.9 The JC, in consultation with the RT, may provide a recommendation to the District regarding the movement or transfer of the RT who is participating in PAR.

25.8 Miscellaneous

- 25.8.1 Direct control and monitoring of the PAR Program shall be the responsibility of the PAR Joint Committee as described herein. With respect to the Employer's management structure, the PAR Program shall be placed under the general purview of the Employer's Human Resources Services and Support.
- 25.8.2 The provisions of the PAR article are not subject to individual member grievances. This does not diminish a unit member's right to grieve under the evaluation article of this contract. 25.8.3 Any claims that the article has not been properly

- implemented shall be presented in writing to the Joint Committee, with a copy to the Employer and the Association.
- 25.8.4 Expenditures made for this program shall not exceed the revenues received under AB1X and where applicable BTSA.
- 25.8.5 The stipend to be paid to the Joint Committee members and Consulting Teachers shall be added to the unit member's annual salary and shall be STRS applicable.
- 25.8.6 The Joint Committee may request data necessary to fulfill its duties.
- 25.8.7 Nothing herein shall modify or in any manner affect the right of the Employer under provisions of the Education Code relating to the employment, classification, retention, or non-reelection of certificated employees.

ARTICLE 26 - CONTRACT WAIVERS

26.1 General Provisions

- 26.1.1 School site improvement, restructuring or reform plans shall not contain any provisions contrary to, or in conflict with any article or section of this Agreement, unless specific waivers to such Agreement articles and sections are agreed to by the Association and the Employer.
- 26.1.2 The following Articles in this Agreement shall not be eligible for contract waivers: Recognition, Non-Discrimination, Affirmative Action, Academic Freedom, Association Rights, and Grievances. Compensation (i.e., Article 24 and appropriate appendices) is also non-waviable, subject to section 26.2.1.4.
- 26.1.3 The District shall develop its own internal process by which District policies and administrative procedures may be waived.
- 26.1.4 The Association shall develop its own internal processes by which provisions of this Agreement may be waived (See Appendix 13).
- 26.1.5 Requests for waivers shall be approved or denied within thirty (30) days of submission to the OEA Waiver Panel.
- 26.1.6 No waiver shall be granted that is in contravention of federal or state law.

26.2 Waiver for Specialized Programs developed at site-level

- 26.2.1 An application for creation/modification of a Specialized program pursuant to Article 27 may receive waivers from certain provisions of this Agreement for a period not to exceed three (3) years pursuant to the following process and requirements:
 - 26.2.1.1 The OEA Waiver Panel has up to 20 work days to review the application, identify any impacted provisions of the Agreement and consult with unit members at the site about the same. Upon request of waiver panel members, the District may, but is not required, to approve release time to facilitate this process.
 - A vote at the site on the application for designation as a Specialized program pursuant to Article 27 occurs after the actions in section 26.2.1.1 above. Approval of the application requires an affirmative vote pursuant to Appendix 13 of this Agreement. Such vote will be conducted by secret ballot by the OEA site representative.
 - 26.2.1.3 The Association may develop its own internal processes, distinct from section 26.1.4, by which provisions of this Agreement may be waived in conjunction with an application for a specialized program pursuant to Article 27 .
 - Notwithstanding the non-waivability of Article 24 as stated in section 26.1.2, a specialized program application may call for the performance of additional hours of student instruction. If so, any such hours required to be worked beyond the contractual work day and/or such days required to be worked beyond the contractual work year shall be compensated at each teacher's per diem hourly or daily rate respectively.

ARTICLE 27 - SPECIALIZED PROGRAMS

27.1 Specialized Program Development

- 27.1.1 The Employer and the Association recognize that development of school environments in which site-level employees have flexibility to implement targeted strategies to advance student academic achievement may improve effective professional practice and the educational process. Accordingly, to facilitate the development of such specialized programs and provide more flexibility for site-level decision-making, the parties agree as follows:
- 27.1.2 Schools may develop a plan to implement an innovative, specialized program as defined in Article 27.2.1 that may require flexibility and, therefore, deviation from certain provisions of this Agreement and/or District policies.

27.2 Specialized Programs

- 27.2.1 "Specialized Programs" are schools or programs within a school designated to implement a specialized academic and/or operational program targeted to improve student academic achievement. Examples of existing specialized programs include, but are not limited to:
 - 27.2.1.1 Dual Language Immersion
 - 27.2.1.2 Linked Learning
 - 27.2.1.3 Newcomers Program
 - 27.2.1.4 Expeditionary Learning
 - 27.2.1.5 Big Picture
 - 27.2.1.6 Blended Learning
 - 27.2.1.7 Lab Schools associated with Teacher Training Programs
- 27.2.2 Schools may submit a specialized program application for approval by the Board of Education for school-based flexibilities.

27.3 Application Process

- 27.3.1 The Principal/Site Leader and faculty shall jointly decide to submit the application upon approval by a minimum of two-thirds of the OEA-represented employees assigned half-time or more to the site/program submitting the proposal.
- 27.3.2 Each submitted application must include:
 - 27.3.2.1 A detailed description of the specialized program design, including the duration thereof.
 - 27.3.2.2 Any hiring criteria for unit members determined to be necessary to the success of the specialized program.
 - 27.3.2.3 An Affirmation Agreement specifying the requirements of unit members at the site, if applicable.
 - 27.3.2.4 Written confirmation of waiver approval pursuant to Article 26 , if applicable.

- 27.3.2.5 The conditions under which a specialized program may either be terminated prior to, or renewed as of its specified expiration date.
- 27.3.3 Applications shall be reviewed by the Superintendent/designee and will include assessment of the research on which the plan is based and the capacity of the school community to implement the plan.
- 27.3.4 The Superintendent shall forward their recommendation on approval of the application to the Board of Education, whose decision shall be final.
- 27.3.5 Once designated, a specialized program shall maintain its designation for the duration of the approved application, provided, however, that continued waivers of this Agreement beyond three (3) years in duration, if applicable, must be obtained pursuant to Article 26. Upon expiration, the specialized program will transition to regular school status unless the program is terminated early or a successful renewal application and redesignation is approved pursuant to the conditions set forth in the approved application.
- 27.3.6 At the end of each semester, or at other benchmarks established by the site, unit members may elect to evaluate and propose modifications to the Affirmation Agreement to more effectively address the needs of students. Any modifications which require waiver of provisions of this Agreement beyond waivers already granted, if any, must be approved through the process in Article 26.
- 27.4 Teachers will be notified of Affirmation Agreements and hiring criteria of specialized programs in vacancy lists and alerts announcing new vacancies pursuant to Article 12 of this Agreement.
- **Election to Remain at Site.** Teachers currently assigned to a school site with a specialized program(s) may elect to remain at the site provided that they sign the Affirmation Agreement, if applicable. If an Affirmation Agreement is signed, the provisions of Article 12.2.2.7 of this Agreement shall apply.
- 27.6 Election to Exit Site. Teachers currently assigned to a site with a specialized program(s) may elect not to remain at the site prior to implementation of the program(s). Written notice of this election shall be provided to Human Resources not later than the commencement of the second semester immediately prior to the year of implementation. Teachers electing to exit the site shall be placed in the Talent Pool and participate in the transfer procedure, set forth in Article 12 of this Agreement, at Phase 1 for the subsequent school year.
- **Specialized Program Evaluation.** Upon the request of the District or the Association, representatives of the parties shall meet to evaluate fidelity in implementation of the specialized program while it is in progress and/or fulfillment of the goals and expectations set forth in the specialized program during the first semester of the final year of the special program.

ARTICLE 28 - RESTRUCTURED SCHOOLS

- Any schools designated by the District for possible intervention and/or restructuring shall comply with all provisions of this Agreement, unless waivers are agreed to and approved by the Association and the District according to Article 26 of this Agreement. The District shall notify the Association and impacted school site(s) of schools the District has designated for possible intervention and/or restructuring by May 1st. The District may provide support during the next school year. If intervention and/or restructuring is mandated, the year following identification shall be spent planning for such intervention and/or restructuring. For example, if the District notifies the association that a site has been designated for possible intervention and/or restructuring on or before May 1, 2016, the District may then provide support during the 2016-2017 school year and that year would be spent in planning such intervention and/or restructuring, if any was contemplated.
- As part of the intervention and/or restructuring process, school communities shall be offered the opportunity to develop and submit improvement plans that may include a specialized program application pursuant to Article 27. The District will consider creative means to make funds available to support the site staff's efforts to develop and implement an improvement plan that may include a specialized program.
- **Election to Remain at Site:** Teachers currently assigned to a school site designated by the District for intervention may elect to remain at the site provided that they sign the Affirmation Agreement, if applicable. If an Affirmation Agreement is signed, the provisions of Article 12.2.2.7 of this Agreement shall apply.
- **28.4 Election to Exit Site:** Teachers currently assigned to a site designated by the District for intervention and/or restructuring may elect not to remain at the site prior to implementation of the specialized plan or restructuring. Written notice of this election shall be provided to Human Resources not later than the commencement of the second semester immediately prior to the year of implementation. Teachers electing to exit the site shall be placed in the Talent Pool and participate in the transfer procedure, set forth in Article 12 of this Agreement, at Phase 1 for the subsequent school year.

This page is intentionally blank.